### UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

### **FORM 10-Q**

	QUARTERLY REPORT PURSUANT T ACT OF 1934	TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE
	For the quarterly period ended OCTOBER 27, 20	<u>12</u>
		or
	TRANSITION REPORT PURSUANT T ACT OF 1934	O SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE
	For the transition period from to	
	Commission	n File Number: 1-4365
		DUSTRIES, INC. trant as specified in its charter)
	Georgia	58-0831862
(State o	or other jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
	999 Peachtree Street, N.E (Address of principal execut	., Suite 688, Atlanta, Georgia 30309 (ve offices) (Zip Code)
		4) 659-2424 ne number, including area code)
	(Registrant's telephol	ic number, including area code)
during the prece		required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 ant was required to file such reports), and (2) has been subject to such filing
required to be su		ically and posted on its corporate Web site, if any, every Interactive Data File 3-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter ☑ No □
	check mark whether the registrant is a large accelerated fif "large accelerated filer," "accelerated filer" and "smalle	ler, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See reporting company" in Rule 12b-2 of the Exchange Act.
Large accelerate	ed filer £ Accelerated filer ☑	Non-accelerated filer $\pounds$ Smaller reporting company $\pounds$ (Do not check if a smaller reporting company)
Indicate by	check mark whether the registrant is a shell company (as	defined in Rule 12b-2 of the Exchange Act). Yes □ No ☑
Indicate the	number of shares outstanding of each of the issuer's class	ses of common stock, as of the latest practicable date.
	Title of each class	Number of shares outstanding as of November 30, 2012
	Common Stock, \$1 par value	16,572,583

#### OXFORD INDUSTRIES, INC. INDEX TO FORM 10-Q For the third quarter of fiscal 2012

	Page
PART I. FINANCIAL INFORMATION	
tem 1. Financial Statements	
Condensed Consolidated Balance Sheets (Unaudited)	4
Condensed Consolidated Statements of Earnings (Unaudited)	5
Condensed Consolidated Statements of Comprehensive Income (Unaudited)	6
Condensed Consolidated Statements of Cash Flows (Unaudited)	7
Notes to Condensed Consolidated Financial Statements (Unaudited)	8
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	11
Item 3. Quantitative and Qualitative Disclosures About Market Risk	32
Item 4. Controls and Procedures	33
PART II. OTHER INFORMATION	
ftem 1. Legal Proceedings	33
ftem 1A. Risk Factors	34
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	34
Item 3. Defaults Upon Senior Securities	34
Item 4. Mine Safety Disclosures	34
Item 5. Other Information	34
Item 6. Exhibits	34
Signatures	35
Signature .	33
2	
_	

#### CAUTIONARY STATEMENTS REGARDING FORWARD-LOOKING STATEMENTS

Our SEC filings and public announcements may include forward-looking statements about future events. Generally, the words "believe," "expect," "intend," "estimate," "anticipate," "project," "will" and similar expressions identify forward-looking statements, which generally are not historical in nature. We intend for all forward-looking statements contained herein, in our press releases or on our website, and all subsequent written and oral forward-looking statements attributable to us or persons acting on our behalf, to be covered by the safe harbor provisions for forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 and the provisions of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934 (which Sections were adopted as part of the Private Securities Litigation Reform Act of 1995). Important assumptions relating to these forward-looking statements include, among others, assumptions regarding the impact of economic conditions on consumer demand and spending, particularly in light of general economic uncertainty that continues to prevail, demand for our products, timing of shipments requested by our wholesale customers, expected pricing levels, competitive conditions, disciplined execution by key management, the timing and cost of store openings and of planned capital expenditures, costs of products and raw materials we purchase, costs of labor, acquisition and disposition activities, expected outcomes of pending or potential litigation and regulatory actions and access to capital and/or credit markets. Forward-looking statements reflect our current expectations, based on currently available information, and are not guarantees of performance. Although we believe that the expectations reflected in such forward-looking statements are reasonable, these expectations could prove inaccurate as such statements involve risks and uncertainties, many of which are beyond our ability to control or predict. Should one or more of these risks or uncertainties, or other risks or uncertainties not currently known to us or that we currently deem to be immaterial, materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those anticipated, estimated or projected. Important factors relating to these risks and uncertainties include, but are not limited to, those described in Part I, Item IA. Risk Factors contained in our Annual Report on Form 10-K for fiscal 2011, as updated by Part II, Item 1A. Risk Factors in this report and those described from time to time in our future reports filed with the SEC. We caution that one should not place undue reliance on forward-looking statements, which speak only as of the date on which they are made. We disclaim any intention, obligation or duty to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, except as required by law.

#### **DEFINITIONS**

Unless the context requires otherwise, the following terms, or words of similar import, have the following meanings:

Our, us or we: Oxford Industries, Inc. and its consolidated subsidiaries

SG&A: Selling, general and administrative expenses

Discontinued operations: The assets and operations of our former Oxford Apparel operating group which we sold in the fourth quarter of fiscal 2010, as discussed in our Annual Report on Form 10-K for fiscal 2011

SEC: U.S. Securities and Exchange Commission

FASB: Financial Accounting Standards Board

U.S. GAAP: Generally accepted accounting principles in the United States

Fiscal 2013
Fiscal 2012
Fiscal 2011
First nine months fiscal 2012
First nine months fiscal 2011
Fourth quarter fiscal 2012
Third quarter fiscal 2012
Second quarter fiscal 2012
First quarter fiscal 2012
Fourth quarter fiscal 2011
Third quarter fiscal 2011
Third quarter fiscal 2011
First quarter fiscal 2011
First quarter fiscal 2011

52 weeks ending February 1, 2014
53 weeks ending February 2, 2013
52 weeks ended January 28, 2012
39 weeks ended October 27, 2012
39 weeks ended October 29, 2011
14 weeks ending February 2, 2013
13 weeks ended October 27, 2012
13 weeks ended July 28, 2012
13 weeks ended April 28, 2012
13 weeks ended January 28, 2012
13 weeks ended October 29, 2011
13 weeks ended July 30, 2011
13 weeks ended April 30, 2011

#### PART I. FINANCIAL INFORMATION

#### ITEM 1. FINANCIAL STATEMENTS

## OXFORD INDUSTRIES, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(in thousands, except par amounts)

	October 27, 2012	January 28, 2012	October 29, 2011
ASSETS			
Current Assets:			
Cash and cash equivalents	\$ 5,621	\$ 13,373	\$ 4,962
Receivables, net	68,920	59,706	66,372
Inventories, net	102,172	103,420	91,003
Prepaid expenses, net	21,251	19,041	17,425
Deferred tax assets	19,327	19,733	17,596
Total current assets	217,291	215,273	197,358
Property and equipment, net	123,841	93,206	91,121
Intangible assets, net	165,013	165,193	166,082
Goodwill	17,273	16,495	16,555
Other non-current assets, net	21,404	19,040	18,385
Total Assets	\$544,822	\$509,207	\$489,501
LIABILITIES AND SHAREHOLDERS' EQUITY			
Current Liabilities:			
Trade accounts payable and other accrued expenses	\$ 78,550	\$ 89,149	\$ 78,209
Accrued compensation	21,705	23,334	21,748
Contingent consideration payable	2,500	2,500	_
Short-term debt and current maturities of long-term debt	6,955	2,571	3,279
Total current liabilities	109,710	117,554	103,236
Long-term debt, less current maturities	123,301	103,405	103,290
Non-current contingent consideration	9,945	10,645	12,545
Other non-current liabilities	43,107	38,652	41,328
Non-current deferred income taxes	31,459	34,882	30,738
Commitments and contingencies			
Shareholders' Equity:			
Common stock, \$1.00 par value per common share	16,572	16,522	16,499
Additional paid-in capital	103,603	99,670	98,434
Retained earnings	130,153	111,551	106,645
Accumulated other comprehensive loss	(23,028)	(23,674)	(23,214)
Total shareholders' equity	227,300	204,069	198,364
Total Liabilities and Shareholders' Equity	\$544,822	\$509,207	\$489,501

See accompanying notes.

# OXFORD INDUSTRIES, INC. CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS (UNAUDITED) (in thousands, except per share amounts)

	Third Quarter Fiscal 2012	Third Quarter Fiscal 2011	First Nine Months Fiscal 2012	First Nine Months Fiscal 2011
Net sales	\$181,414	\$170,280	\$619,296	\$559,234
Cost of goods sold	84,592	81,540	274,980	249,897
Gross profit	96,822	88,740	344,316	309,337
SG&A	94,146	85,161	295,656	264,947
Change in fair value of contingent consideration	600	600	1,800	1,800
Royalties and other operating income	3,844	3,837	12,166	12,650
Operating income	5,920	6,816	59,026	55,240
Interest expense, net	959	3,705	7,876	12,777
Loss on repurchase of senior secured notes	_	769	9,143	9,017
Earnings from continuing operations before income taxes	4,961	2,342	42,007	33,446
Income taxes	1,951	731	15,967	11,255
Earnings from continuing operations	3,010	1,611	26,040	22,191
Earnings from discontinued operations, net of taxes		13		137
Net earnings	\$ 3,010	\$ 1,624	\$ 26,040	\$ 22,328
Earnings from continuing operations per common share:				
Basic	\$ 0.18	\$ 0.10	\$ 1.57	\$ 1.34
Diluted	\$ 0.18	\$ 0.10	\$ 1.57	\$ 1.34
Earnings from discontinued operations, net of taxes per common share:				
Basic	\$ —	\$ —	\$ —	\$ 0.01
Diluted	\$ —	\$ —	\$ —	\$ 0.01
Net earnings per common share:				
Basic	\$ 0.18	\$ 0.10	\$ 1.57	\$ 1.35
Diluted	\$ 0.18	\$ 0.10	\$ 1.57	\$ 1.35
Weighted average common shares outstanding:				
Basic	16,580	16,502	16,555	16,510
Diluted	16,591	16,517	16,572	16,527
Dividends declared per common share	\$ 0.15	\$ 0.13	\$ 0.45	\$ 0.39

See accompanying notes.

# OXFORD INDUSTRIES, INC. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (UNAUDITED) (in thousands)

	Third Quarter Fiscal 2012	Third Quarter Fiscal 2011	First Nine Months Fiscal 2012	First Nine Months Fiscal 2011
Net earnings	\$ 3,010	\$ 1,624	\$ 26,040	\$ 22,328
Other comprehensive income (loss), net of taxes				
Foreign currency translation gain (loss)	1,031	(642)	1,129	483
Net unrealized gain (loss) on forward foreign currency				
exchange contracts	(569)	151	(458)	122
Net unrealized gain (loss) on interest rate swap contract	(25)	_	(25)	_
Total other comprehensive income (loss), net of taxes	437	(491)	646	605
Comprehensive income	\$ 3,447	\$ 1,133	\$ 26,686	\$ 22,933

See accompanying notes.

## OXFORD INDUSTRIES, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

	First Nine Months Fiscal 2012	First Nine Months Fiscal 2011
Cash Flows From Operating Activities:		
Earnings from continuing operations	\$ 26,040	\$ 22,191
Adjustments to reconcile earnings from continuing operations to net cash provided by operating		
activities:		
Depreciation	17,430	15,288
Amortization of intangible assets	769	897
Change in fair value of contingent consideration	1,800	1,800
Amortization of deferred financing costs and bond discount	846	1,286
Loss on repurchase of senior secured notes	9,143	9,017
Stock compensation expense	2,215	1,635
Deferred income taxes	(3,151)	3,223
Changes in working capital, net of acquisitions and dispositions:		
Receivables	(8,902)	(16,080)
Inventories	2,266	(5,511)
Prepaid expenses	(2,541)	(4,717)
Current liabilities	(12,501)	(8,690)
Other non-current assets	(3,182)	2,536
Other non-current liabilities	4,444	(3,441)
Net cash provided by operating activities	34,676	19,434
Cash Flows From Investing Activities:		
Acquisitions, net of cash acquired	(4,313)	(398)
Purchases of property and equipment	(47,653)	(22,448)
Net cash used in investing activities	(51,966)	(22,846)
Cash Flows From Financing Activities:		
Repayment of revolving credit arrangements	(149,266)	(60,579)
Proceeds from revolving credit arrangements	276,826	63,865
Repurchase of senior secured notes	(111,000)	(52,175)
Deferred financing costs paid	(1,524)	_
Proceeds from issuance of common stock	1,768	2,017
Dividends on common stock	(7,438)	(6,425)
Net cash provided by (used in) financing activities	9,366	(53,297)
Cash Flows from Discontinued Operations:		
Net cash provided by discontinued operations		17,479
Net change in cash and cash equivalents	(7,924)	(39,230)
Effect of foreign currency translation on cash and cash equivalents	172	98
Cash and cash equivalents at the beginning of year	13,373	44,094
Cash and cash equivalents at the end of the period	\$ 5,621	\$ 4,962
Supplemental disclosure of cash flow information:		
Cash paid for interest, net	\$ 7,279	\$ 8,890
Cash paid for income taxes, including income taxes paid for discontinued operations	\$ 20,904	\$ 40,065
See accompanying notes.		

### OXFORD INDUSTRIES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) THIRD QUARTER OF FISCAL 2012

1. Basis of Presentation: The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with U.S. GAAP for interim financial reporting and the instructions of Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by U.S. GAAP. We believe the accompanying unaudited condensed consolidated financial statements reflect all normal, recurring adjustments that are necessary for a fair presentation of our financial position and results of operations as of the dates and for the periods presented. Results of operations for the interim periods presented are not necessarily indicative of results to be expected for our full fiscal year. The accounting policies applied during the interim periods presented are consistent with the significant accounting policies described in our Annual Report on Form 10-K for fiscal 2011.

In May 2011, the FASB amended ASC 820 "Fair Value Measurements and Disclosures" in order to clarify existing guidance in U.S. GAAP, better align ASC 820 with International Accounting Standards and require additional fair value disclosures. The amendments to ASC 820 were adopted by us in the first quarter of fiscal 2012, with all amendments applied prospectively with changes in measurements, if any, recognized in earnings in the first quarter of fiscal 2012. The adoption of the amendments to ASC 820 in the first quarter of fiscal 2012 did not have a material impact on our consolidated financial statements.

In July 2012, the FASB amended ASC 350 "Intangibles - Goodwill and Other." This amendment is intended to reduce the cost and complexity of performing an impairment test for indefinite-lived intangible assets by simplifying how we test those assets for impairment and to improve consistency in impairment testing among long-lived asset categories. The amendment provides us with the option first to assess qualitative factors to determine whether it is more likely than not that an indefinite-lived intangible asset is impaired as a basis for determining whether it is necessary to perform the quantitative impairment test in accordance with Subtopic 350-30, Intangibles - Goodwill and Other — General Intangibles Other than Goodwill. The more-likely-than-not threshold is defined as having a likelihood of more than 50 percent. We also have the option to bypass the qualitative assessment for any indefinite-lived intangible asset in any period and proceed directly to performing the quantitative impairment test. We will be able to resume performing the qualitative assessment in any subsequent period. The amendment is effective for annual and interim impairment tests performed for fiscal years beginning after September 15, 2012. Early adoption is permitted. We do not expect the adoption of the amendment to ASC 350 to have a material impact on our financial position, results of operations or cash flows.

2. Inventories: The components of inventories as of the dates specified are summarized as follows (in thousands):

	October 27, 2012	January 28, 2012	October 29, 2011
Finished goods	\$146,044	\$143,482	\$126,570
Work in process	3,561	6,244	4,096
Fabric, trim and supplies	4,943	6,070	6,291
LIFO reserve	(52,376)	(52,376)	(45,954)
Total	\$102,172	\$103,420	\$ 91,003

LIFO accounting adjustments, which we consider to include changes in the LIFO reserve as well as the impact of changes in inventory reserves related to lower of cost or market adjustments that do not exceed the LIFO reserve, were a credit of \$0.4 million in the third quarter of fiscal 2012, a charge of \$0.2 million in the third quarter of fiscal 2011, a credit of \$0.5 million in the first nine months of fiscal 2012 and \$0.0 million impact in the first nine months of fiscal 2011.

3. Operating Group Information: Our business is primarily operated through our four operating groups: Tommy Bahama, Lilly Pulitzer, Ben Sherman and Lanier Clothes, as disclosed in our Annual Report on Form 10-K for fiscal 2011. We identify our operating groups based on the way our management organizes the components of our business for purposes of allocating resources and assessing performance. The table below presents certain information (in thousands) about our operating groups, as well as Corporate and Other, which is a reconciling category for reporting purposes.

	Third Quarter Fiscal 2012	Third Quarter Fiscal 2011	First Nine Months Fiscal 2012	First Nine Months Fiscal 2011
Net Sales				
Tommy Bahama	\$103,193	\$ 92,500	\$371,790	\$324,546
Lilly Pulitzer	26,939	16,668	93,475	71,364
Ben Sherman	19,781	25,191	57,234	65,505
Lanier Clothes	27,180	33,080	84,995	88,995
Corporate and Other	4,321	2,841	11,802	8,824
Total Net Sales	\$181,414	\$170,280	\$619,296	\$559,234
Depreciation and Amortization				
Tommy Bahama	\$ 4,511	\$ 4,012	\$ 12,793	\$ 11,277
Lilly Pulitzer	604	553	1,674	1,482
Ben Sherman	729	583	2,016	1,736
Lanier Clothes	107	106	298	322
Corporate and Other	526	473	1,418	1,368
Total Depreciation and Amortization	\$ 6,477	\$ 5,727	\$ 18,199	\$ 16,185
Operating Income (Loss)				
Tommy Bahama	\$ 3,366	\$ 4,624	\$ 45,511	\$ 45,381
Lilly Pulitzer	3,528	(363)	21,949	12,264
Ben Sherman	(2,149)	301	(6,352)	(2,281)
Lanier Clothes	2,402	4,331	8,845	11,319
Corporate and Other	(1,227)	(2,077)	(10,927)	(11,443)
Total Operating Income	5,920	6,816	59,026	55,240
Interest expense	959	3,705	7,876	12,777
Loss on repurchase of senior secured notes	_	769	9,143	9,017
<b>Earnings from Continuing Operations Before Income</b>				
Taxes	\$ 4,961	\$ 2,342	\$ 42,007	\$ 33,446

4. Income Taxes: The income tax expense for each period presented reflects effective tax rates of 39.3%, 31.2%, 38.0% and 33.7% for the third quarter of fiscal 2012, the third quarter of fiscal 2011, the first nine months of fiscal 2012 and the first nine months of fiscal 2011, respectively. The effective tax rate for the third quarter of fiscal 2012 and the first nine months of fiscal 2012 were impacted by losses in foreign jurisdictions for which we were not able to recognize an income tax benefit and a greater proportion of our earnings being in higher tax jurisdictions, which offset favorable discrete items, including a reduction in income tax contingency reserves upon the expiration of the corresponding statute of limitations totaling approximately \$2.1 million. The effective tax rate for the third quarter of fiscal 2011 and the first nine months of fiscal 2011 were impacted by certain favorable discrete items, including the reduction of income tax contingency reserves upon the expiration of the corresponding statute of limitations totaling \$0.2 million for the third quarter and \$0.6 million for the first nine months, as well as the recognition of an income tax benefit for losses in foreign jurisdictions. Additionally, the first nine months of fiscal 2012 and the first nine months of fiscal 2011 reflected the favorable benefit of reductions of enacted tax rates in certain jurisdictions, including the United Kingdom.

**5. Debt:** The following table details our debt (in thousands) as of the dates specified:

	October 27, 2012	January 28, 2012	October 29, 2011
\$235 million U.S. Secured Revolving Credit Facility ("U.S. Revolving Credit			
Agreement")(1)	\$123,301	N/A	N/A
\$175 million U.S. Secured Revolving Credit Facility ("Prior Revolving Credit			
Agreement")(1)	N/A	\$ —	\$ 1,621
£7 million Senior Secured Revolving Credit Facility ("U.K. Revolving Credit			
Agreement")(2)	6,955	2,571	1,658
11.375% Senior Secured Notes ("11 <sup>3</sup> / <sub>8</sub> % Senior Secured Notes")(3)(4)	N/A	105,000	105,000
Unamortized discount	_	(1,595)	(1,710)
Total debt	130,256	105,976	106,569
Short-term debt and current maturities of long-term debt	(6,955)	(2,571)	(3,279)
Long-term debt, less current maturities	\$123,301	\$103,405	\$103,290

- (1) The U.S. Revolving Credit Agreement, entered into in June 2012, amended and restated the Prior Revolving Credit Agreement, which was scheduled to mature in August 2013. The U.S. Revolving Credit Agreement generally (i) is limited to a borrowing base consisting of specified percentages of eligible categories of assets; (ii) accrues interest, unused line fees and letter of credit fees based upon a pricing grid which is tied to average unused availability and/or utilization; (iii) requires periodic interest payments with principal due at maturity (June 2017); and (iv) is generally secured by a first priority security interest in the accounts receivable, inventory, general intangibles and eligible trademarks, investment property (including the equity interests of certain subsidiaries), deposit accounts, intercompany obligations, equipment, goods, documents, contracts, books and records and other personal property of Oxford Industries, Inc. and substantially all of its domestic subsidiaries.
- (2) The U.K. Revolving Credit Agreement generally (i) accrues interest at the bank's base rate plus an applicable margin; (ii) requires interest payments monthly with principal payable on demand; and (iii) is collateralized by substantially all of the assets of our United Kingdom Ben Sherman subsidiaries.
- (3) In the second quarter of fiscal 2012, we redeemed all of the outstanding \$105 million in aggregate principal amount of the 113/8% Senior Secured Notes, which were scheduled to mature in July 2015. The redemption of the 113/8% Senior Secured Notes for \$111.0 million, plus accrued interest, and the related write-off of approximately \$1.7 million of unamortized deferred financing costs and \$1.4 million of unamortized bond discount resulted in a loss on repurchase of senior notes of approximately \$9.1 million. The redemption of the 113/8% Senior Secured Notes satisfied and discharged all of our obligations with respect to the 113/8% Senior Secured Notes and the related indenture and was funded primarily through borrowings under our U.S. Revolving Credit Agreement.
- (4) In the second and third quarters of fiscal 2011, we repurchased, in privately negotiated transactions, \$45.0 million in aggregate principal amount of the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes for \$52.2 million, plus accrued interest. The repurchase of the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes and related write-off of approximately \$1.8 million of unamortized deferred financing costs and bond discount resulted in a loss on repurchase of senior notes of approximately \$9.0 million during the first nine months of fiscal 2011.

To the extent cash flow needs exceed cash flow provided by our operations we will have access, subject to their terms, to our lines of credit to provide funding for operating activities, capital expenditures and acquisitions, if any. Our credit facilities are also used to finance trade letters of credit for product purchases, which are drawn against our lines of credit at the time of shipment of the products and reduce the amounts available under our lines of credit and borrowing capacity under our credit facilities when issued. As of October 27, 2012, approximately \$7.8 million of trade letters of credit and other limitations on availability in the aggregate were outstanding against our credit facilities. After considering these limitations and the amount of eligible assets in our borrowing base, as applicable, as of October 27, 2012, we had approximately \$90.0 million and \$1.4 million in unused availability under the U.S. Revolving Credit Agreement and the U.K. Revolving Credit Agreement, respectively, subject to the respective limitations on borrowings set forth in the U.S. Revolving Credit Agreement and the U.K. Revolving Credit Agreement.

#### ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis should be read in conjunction with our unaudited condensed consolidated financial statements and the notes to the unaudited condensed consolidated financial statements contained in this report and the consolidated financial statements, notes to consolidated financial statements and Management's Discussion and Analysis of Financial Condition and Results of Operations contained in our Annual Report on Form 10-K for fiscal 2011.

#### **OVERVIEW**

We generate revenues and cash flow primarily through our design, sourcing, marketing and distribution of branded apparel products bearing the trademarks of our owned lifestyle brands as well as certain licensed and private label apparel products. We distribute our products through our direct to consumer channels, including our retail stores, e-commerce websites and restaurants, and our wholesale distribution channel, which includes better department stores, specialty stores, national chains, specialty catalogs and Internet retailers. In fiscal 2011, more than 90% of our consolidated net sales were to customers located in the United States, with the remainder primarily sales of our Ben Sherman products in the United Kingdom and Europe. We source substantially all of our products through third party manufacturers located outside of the United States and United Kingdom.

We operate in highly competitive domestic and international markets in which numerous U.S.-based and foreign apparel firms compete. No single apparel firm or small group of apparel firms, dominate the apparel industry and our direct competitors vary by operating group. We believe that the principal competitive factors in the apparel industry are the reputation, value and image of brand names; design; consumer preference; price; quality; marketing; and customer service. We believe our ability to compete successfully in styling and marketing is directly related to our proficiency in foreseeing changes and trends in fashion and consumer preference, and presenting appealing products for consumers. In some instances, a retailer that is our customer may compete directly with us by offering certain of their own competing products, some of which may be sourced directly by our customer, in their own retail stores. Additionally, the apparel industry is cyclical and dependent upon the overall level of discretionary consumer spending, which changes as regional, domestic and international economic conditions change. Often, negative economic conditions have a longer and more severe impact on the apparel and retail industry than the conditions have on other industries.

The global economic environment and resulting economic uncertainty that began in fiscal 2008 continue to impact each of our operating groups, and the apparel industry as a whole. Although some signs of economic improvements exist and declines in consumer spending in the United States have moderated, unemployment levels remain high, the retail environment remains highly promotional and a significant amount of economic uncertainty remains. The economies in the United Kingdom and Europe, which are important to our Ben Sherman operating group, continue to struggle.

While we anticipate sales of our products may continue to be negatively impacted as long as there is an elevated level of economic uncertainty, we believe that our Tommy Bahama and Lilly Pulitzer lifestyle brands have significant opportunities for long-term growth. We believe that these lifestyle brands have opportunities for growth in their direct to consumer businesses through expansion of our retail store operations as well as increases in same store and ecommerce sales. We also believe that these lifestyle brands provide an opportunity for moderate sales increases in their wholesale businesses primarily from current customers adding to their existing door count and the selective addition of new wholesale customers. The Ben Sherman lifestyle brand currently faces challenges due to its elevation of the distribution of the brand and the economic conditions in the United Kingdom and Europe as well as missteps in the merchandise mix during the third quarter of fiscal 2012. We believe that in the long-term Ben Sherman will have future growth opportunities if the elevation of the brand is successful and the economic conditions improve.

Although the challenging economic conditions continue to have an impact on our business and the apparel industry as a whole, and we continue to focus on minimizing inventory markdown risk and promotional pressure, we were more aggressive in our inventory purchases for fiscal 2011 and fiscal 2012 and anticipate continuing this approach for fiscal 2013 as long as the economic conditions do not deteriorate. The second half of fiscal 2011 was impacted by pricing pressures on raw materials, fuel, transportation, labor and other costs necessary for the production and sourcing of apparel products and these pricing pressures continued to negatively impact cost of goods sold into the first half of fiscal 2012. The impact of these increased product costs began to moderate in the third quarter of fiscal 2012.

We continue to believe it is important to focus on maintaining a strong balance sheet and ample liquidity, and we believe that our strong balance sheet and liquidity coupled with positive cash flow from operations will provide us sufficient resources to fund future investments in our owned lifestyle brands. In the future, we may add additional lifestyle brands to our portfolio, if we identify appropriate lifestyle brands which meet our investment criteria; however, we believe that we have significant opportunities to appropriately deploy our capital and resources in our existing owned lifestyle brands.

The following table sets forth our consolidated operating results (in thousands, except per share amounts) for the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011:

	First Nine M	Months
	Fiscal 2012	Fiscal 2011
Net sales	\$ 619,296	\$ 559,234
Operating income	\$ 59,026	\$ 55,240
Earnings from continuing operations	\$ 26,040	\$ 22,191
Earnings from continuing operations per diluted common share	\$ 1.57	\$ 1.34

The primary reasons for the improvement in earnings from continuing operations were:

- An increase in net sales in both the Tommy Bahama and Lilly Pulitzer operating groups;
- Lower interest expense in the first nine months of fiscal 2012 due to (1) our borrowing at lower interest rates in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 due to our redemption of the remaining \$105.0 million in aggregate principal amount of our 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes in July 2012 and (2) the reduction in our average debt levels during the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 primarily as a result of our repurchase of \$45.0 million in aggregate principal amount of our 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes during the second and third quarters of fiscal 2011; and
- The first nine months of fiscal 2012 not having any purchase accounting adjustments while the first nine months of fiscal 2011 included purchase accounting charges of \$1.0 million.

These items were partially offset by:

- A decrease in sales and operating results at both Ben Sherman and Lanier Clothes;
- An increase in SG&A, which was primarily due to (1) the incremental SG&A associated with the operation of retail stores opened in fiscal 2011 and fiscal 2012, (2) certain infrastructure, pre-opening retail store rent and other costs related to the Tommy Bahama international expansion, (3) higher SG&A to support the growing Tommy Bahama and Lilly Pulitzer businesses and (4) the rent charge associated with our Tommy Bahama New York location which opened in the fourth quarter of fiscal 2012; and

• A higher effective tax rate in the first nine months of fiscal 2012 due to us not being able to recognize the income tax benefit of certain losses in foreign jurisdictions and having a greater proportion of our earnings in higher tax jurisdictions, which offset the impact of certain favorable discrete items in that period, while in the first nine months of fiscal 2011, we were able to recognize the income tax benefit of foreign jurisdiction losses as well as having the impact of certain favorable discrete items.

#### Amendment and Restatement of the U.S. Revolving Credit Agreement

On June 14, 2012, we entered into the U.S. Revolving Credit Agreement, which provides for a revolving credit facility of up to \$235 million which may be used to refinance existing debt, to redeem our previously outstanding  $11^3/8\%$  Senior Secured Notes, to fund working capital, to fund future acquisitions and for general corporate purposes.

The U.S. Revolving Credit Agreement amended and restated the Prior Revolving Credit Agreement, which was entered into on August 15, 2008 and scheduled to mature in August 2013. We believe that the covenants in the U.S. Revolving Credit Agreement are generally less restrictive and provide greater flexibility than those contained in the Prior Revolving Credit Agreement. In addition, the U.S. Revolving Credit Agreement allows us to include in our borrowing base certain amounts attributable to "eligible trademarks," which amounts would not have been available for inclusion in the borrowing base under the Prior Revolving Credit Agreement.

The material terms of the U.S. Revolving Credit Agreement are described in note 4 in the unaudited condensed consolidated financial statements and the Financial Condition, Liquidity and Capital Resources section of this Management's Discussion and Analysis of Financial Condition and Results of Operations, both contained in this report.

#### Redemption of 113/8% Senior Secured Notes

On July 16, 2012, we redeemed all of the outstanding \$105 million in principal amount of the  $11^3/8\%$  Senior Secured Notes, which were scheduled to mature in July 2015. The redemption of the  $11^3/8\%$  Senior Secured Notes at a premium of \$6.0 million and the write-off of approximately \$3.1 million of unamortized deferred financing costs and unamortized bond discount resulted in a loss on repurchase of senior notes of approximately \$9.1 million. The redemption of the  $11^3/8\%$  Senior Secured Notes satisfied and discharged all of our obligations with respect to the  $11^3/8\%$  Senior Secured Notes and the related indenture and was funded through borrowings under our U.S. Revolving Credit Agreement and cash on hand.

#### **OPERATING GROUPS**

Our business is primarily operated through our four operating groups: Tommy Bahama, Lilly Pulitzer, Ben Sherman and Lanier Clothes. We identify our operating groups based on the way our management organizes the components of our business for purposes of allocating resources and assessing performance.

Tommy Bahama designs, sources and markets collections of men's and women's sportswear and related products. The target consumers of Tommy Bahama are primarily affluent men and women age 35 and older who embrace a relaxed and casual approach to daily living. Tommy Bahama® products can be found in our owned Tommy Bahama retail stores within and outside the United States and on our Tommy Bahama e-commerce website, www.tommybahama.com, as well as in better department stores and independent specialty stores throughout the United States and licensed Tommy Bahama retail stores outside the United States. We also operate Tommy Bahama restaurants and license the Tommy Bahama name for various product categories. As of October 27, 2012, we operated 110 owned Tommy Bahama retail stores, including 73 full-price stores, 13 restaurant-retail locations and 24 outlet stores. This store count includes both our 103 owned domestic and seven owned international retail store locations. The seven international stores include five retail stores in Australia which we acquired from our former licensee during the second quarter of fiscal 2012 as well as our Macau and Singapore stores opened in the first quarter and second quarter of fiscal 2012, respectively.

Lilly Pulitzer designs, sources and distributes upscale collections of women's and girl's dresses, sportswear and other products. Lilly Pulitzer was originally created in the late 1950's and is an affluent brand with a heritage and aesthetic based on the Palm Beach resort lifestyle. The brand is somewhat unique among women's brands in that it has demonstrated multi-generational appeal, including young women in college or recently graduated from college; young

mothers with their daughters; and women who are not tied to the academic calendar. Lilly Pulitzer® products can be found in our owned Lilly Pulitzer stores, in Lilly Pulitzer Signature Stores and on our Lilly Pulitzer website, www.lillypulitzer.com, as well as in better department and independent specialty stores. We also license the Lilly Pulitzer name for various product categories. As of October 27, 2012, we operated 18 owned Lilly Pulitzer retail stores.

Ben Sherman is a London-based designer, marketer and distributor of men's branded sportswear and related products. Ben Sherman was established in 1963 as an edgy shirt brand that was adopted by the "Mods" and has throughout its history been inspired by what is new and current in British art, music, culture and style. The brand has evolved into a British modernist lifestyle brand of apparel targeted at style conscious men ages 25 to 40 in multiple markets throughout the world. Ben Sherman® products can be found in better department stores, a variety of independent specialty stores and our owned and licensed Ben Sherman retail stores, as well as on Ben Sherman e-commerce websites. We also license the Ben Sherman name for various product categories. As of October 27, 2012, we operated 19 owned Ben Sherman international and domestic retail stores, including seven outlet stores.

Lanier Clothes designs, sources and markets branded and private label men's tailored clothing, including suits, sportcoats, suit separates and dress slacks across a wide range of price points, with the majority of the business at moderate price points. Our Lanier Clothes branded products are sold under certain trademarks licensed to us by third parties including Kenneth Cole®, Dockers®, Geoffrey Beene® and Ike Behar®. Additionally, we design and market products for our owned Billy London®, Arnold Brant® and Oxford Republic® brands. In addition to the branded businesses, Lanier Clothes designs and sources certain private label tailored clothing products. Significant private label brands for which we produce tailored clothing include Lands' End, Stafford, Alfani, Structure, and Kenneth Roberts. Our Lanier Clothes products are sold to national chains, department stores, specialty stores, specialty catalog retailers and discount retailers throughout the United States.

Corporate and Other is a reconciling category for reporting purposes and includes our corporate offices, substantially all financing activities, elimination of inter-segment sales, LIFO inventory accounting adjustments, other costs that are not allocated to the operating groups and operations of our other businesses which are not included in our four operating groups. The operations that are included in Corporate and Other include our Oxford Golf business and our Lyons, Georgia distribution center.

For further information regarding our operating groups, see Note 3 to our unaudited condensed consolidated financial statements included in this report and Part I, Item 1. Business in our Annual Report on Form 10-K for fiscal 2011.

#### RESULTS OF OPERATIONS

#### THIRD QUARTER OF FISCAL 2012 COMPARED TO THIRD QUARTER OF FISCAL 2011

The following table sets forth the specified line items in our unaudited condensed consolidated statements of earnings both in dollars (in thousands) and as a percentage of net sales. The table also sets forth the dollar change and the percentage change of the data as compared to the same period of the prior year. We have calculated all percentages based on actual data, but percentage columns may not add due to rounding. For purposes of the tables below, "NM" means not meaningful. Individual line items of our consolidated statements of earnings may not be directly comparable to those of our competitors, as classification of certain expenses may vary by company.

Third Quarter						
	Fiscal 201	2	Fiscal 201	1	\$ Change	% Change
Net sales	\$ 181,414	100.0%	\$ 170,280	100.0%	\$ 11,134	6.5%
Cost of goods sold	84,592	46.6%	81,540	47.9%	3,052	3.7%
Gross profit	96,822	53.4%	88,740	52.1%	8,082	9.1%
SG&A	94,146	51.9%	85,161	50.0%	8,985	10.6%
Change in fair value of contingent						
consideration	600	0.3%	600	0.4%	_	_
Royalties and other operating income	3,844	2.1%	3,837	2.3%	7	0.2%
Operating income	5,920	3.3%	6,816	4.0%	(896)	(13.1)%
Interest expense, net	959	0.5%	3,705	2.2%	(2,746)	(74.1)%
Loss on repurchase of senior secured						
notes	_	%	769	0.5%	(769)	(100.0)%
Earnings from continuing operations						
before income taxes	4,961	2.7%	2,342	1.4%	2,619	111.8%
Income taxes	1,951	1.1%	731	0.4%	1,220	166.9%
Earnings from continuing operations	\$ 3,010	1.7%	\$ 1,611	0.9%	\$ 1,399	86.9%

The discussion and tables below compare certain line items included in our statements of earnings for the third quarter of fiscal 2012 to the third quarter of fiscal 2011. Each dollar and percentage change provided reflects the change between these periods unless indicated otherwise. Each dollar and share amount included in the tables is in thousands except for per share amounts.

#### Net Sales

	Third Quarter			
	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Tommy Bahama	\$ 103,193	\$ 92,500	\$ 10,693	11.6%
Lilly Pulitzer	26,939	16,668	10,271	61.6%
Ben Sherman	19,781	25,191	(5,410)	(21.5)%
Lanier Clothes	27,180	33,080	(5,900)	(17.8)%
Corporate and Other	4,321	2,841	1,480	52.1%
Total net sales	\$ 181,414	\$ 170,280	\$ 11,134	6.5%

Consolidated net sales increased \$11.1 million, or 6.5%, in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 primarily due to the increase in net sales at Tommy Bahama and Lilly Pulitzer, which were partially offset by decreased net sales at Lanier Clothes and Ben Sherman, each as discussed below. The third quarter is typically our lowest sales quarter of the fiscal year.

#### Tommy Bahama:

The 11.6% increase in net sales for Tommy Bahama was primarily driven by increased sales in both the direct to consumer and wholesale distribution channels with the largest increase in the direct to consumer business. The increased sales in the direct to consumer business resulted from net sales at additional retail stores operated during fiscal 2012, a significant increase in e-commerce sales and a low single digit percentage increase in comparable full-price retail store sales. Tommy Bahama apparel unit sales increased by 10.3% due to the higher volume in each distribution channel, and the average selling price per unit increased by 1.9%, primarily as a result of the higher proportion of net sales from the direct to consumer channel of distribution. As of October 27, 2012, Tommy Bahama operated 110 retail stores compared to 94 retail stores as of October 29, 2011.

#### Lilly Pulitzer:

The 61.6% increase in net sales for Lilly Pulitzer was driven by increased sales in both the direct to consumer distribution channel and the wholesale distribution channel. The increase in the direct to consumer channel included a significant increase in e-commerce sales, net sales at the additional retail stores operated during fiscal 2012 and a low double digit percentage increase in comparable retail store sales. The higher e-commerce sales benefitted from a very strong end of season flash e-commerce sale which generated approximately \$6.1 million of sales in the third quarter of fiscal 2012 compared to approximately \$2.0 million of end of season flash sales in the third quarter of fiscal 2011 and also benefitted from a healthy full-price ecommerce sales increased during the period. Unit sales increased by 63.8% due to

the higher volume in each distribution channel, while the average selling price per unit decreased by 1.4% despite a greater percentage of Lilly Pulitzer sales being direct to consumer sales. The decreased selling price per unit primarily resulted from a change in product mix as sportswear and knit dresses, both of which generally sell at lower price points than woven dresses, represented a greater proportion of the Lilly Pulitzer business and the impact of the e-commerce sale which sold product at prices more in line with typical wholesale prices rather than full price direct to consumer prices. As of October 27, 2012, Lilly Pulitzer operated 18 retail stores compared to 16 retail stores as of October 29, 2011.

#### Ben Sherman:

Net sales for Ben Sherman decreased by 21.5% in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011. The decrease in net sales for Ben Sherman was primarily driven by a reduction in unit volume of 25.6% primarily attributable to (1) missteps in Ben Sherman's merchandise assortment planning, which resulted in too much of the product offering in styles at the higher end of the price range, particularly in the current economic environment, (2) the difficult economic conditions in the United Kingdom and Europe and (3) our exit from certain moderate wholesale accounts in the United Kingdom. The reduction in units sold was partially offset by an increase in the average selling price per unit of 5.6%. The increase in average selling price per unit was primarily due to (1) a greater proportion of Ben Sherman's sales being retail sales, which generally have higher selling prices, and (2) our strategy to improve the distribution of the brand in the United Kingdom by exiting certain moderate wholesale accounts and increasing product prices, which were partially offset by an unfavorable foreign currency translation impact resulting from a 0.5% change in average exchange rates between the two periods.

#### Lanier Clothes:

Net sales for Lanier Clothes decreased by 17.8% in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011. This decrease resulted from a 16.1% decrease in unit sales, with decreases in both private label and branded businesses, and a 2.1% decrease in average selling price per unit primarily due to competitive factors and a change in sales mix. The decrease in unit sales was due in part to the prior year benefitting from initial shipments related to a new product launch and a slow-down in inventory intake on a replenishment program by a key customer in fiscal 2012.

#### Corporate and Other:

Corporate and Other net sales primarily consisted of the net sales of our Oxford Golf business and our Lyons, Georgia distribution center. The increase in the net sales for Corporate and Other was primarily driven by the higher net sales in our Oxford Golf business during the third quarter of fiscal 2012.

#### Gross Profit

	Third Qua			
	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Gross profit	\$ 96,822	\$ 88,740	\$ 8,082	9.1%
Gross margin (gross profit as a % of net sales)	53.4%	52.1%		
LIFO (credits) charges included in gross profit	\$ (426)	\$ 220		

The 9.1% increase in gross profit in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 was primarily due to the 6.5% increase in net sales, as discussed above, and higher consolidated gross margins. The third quarter of fiscal 2012 experienced an increase in consolidated gross margins resulting, in part, from a change in our consolidated sales mix, the net favorable impact of LIFO accounting adjustments between the two periods and lower product cost pressures on our gross margins. During the third quarter of fiscal 2012, the changes in sales mix included an increased proportion of sales of the higher gross margin Tommy Bahama and Lilly Pulitzer brands and an increased proportion of sales being from the higher gross margin direct to consumer businesses rather than wholesale sales.

Certain of the year-over-year product cost pressures that impacted our operating groups during the second half of fiscal 2011 and the first half of fiscal 2012 began to ease in the third quarter of fiscal 2012 as the impact of certain of the higher product costs moderated. We expect that certain competitive factors impacting Lanier Clothes and Ben Sherman will persist, which may continue to negatively impact the gross margins in these operating groups. Our gross profit may not be directly comparable to those of our competitors, as statement of earnings classification of certain expenses may vary by company.

#### SG&A

	Third Quarter							
	Fiscal	2012	Fiscal	2011	\$ Cha	nge	% Chang	ge
SG&A	\$	94,146	\$	85,161	\$	8,985		10.6%
SG&A (as % of net sales)		51.9%		50.0%				

The increase in SG&A was primarily due to (1) certain infrastructure, pre-opening retail store rent and other costs related to the Tommy Bahama international expansion, (2) the incremental SG&A in the third quarter of fiscal 2012 associated with operating additional Tommy Bahama and Lilly Pulitzer retail stores, (3) the pre-opening rent charge associated with our Tommy Bahama New York location, which in accordance with U.S. GAAP was required to be recognized as an ongoing expense of the store from the time we took possession of the space and (4) higher SG&A to support the growing Tommy Bahama and Lilly Pulitzer businesses. The third quarter of fiscal 2012 included SG&A charges of approximately \$5.1 million related to the Tommy Bahama international expansion and the Tommy Bahama New York store. The third quarter of fiscal 2011 included SG&A charges of \$1.2 million related to international operations and the Tommy Bahama New York store. The increases in SG&A for Tommy Bahama and Lilly Pulitzer were partially offset by SG&A reductions in Ben Sherman and Lanier Clothes. SG&A for both the third quarter of fiscal 2012 and the third quarter of fiscal 2011 included charges of \$0.3 million related to the amortization of intangible assets.

#### Change in fair value of contingent consideration

Third Quarter						
	Fiscal 2012	Fiscal 2011	\$ Change	% Change		
Change in fair value of contingent consideration	\$ 600	\$ 600	\$ —	%		

Change in fair value of contingent consideration reflects the current period impact of the change in fair value of the contingent consideration obligation associated with the Lilly Pulitzer acquisition, as discussed in our Annual Report on Form 10-K for fiscal 2011. We anticipate that the change in contingent consideration for the full year of fiscal 2012 will be approximately \$2.4 million; however, that amount could change significantly depending upon whether there are any changes in future periods to our assumptions about the probability of payment of the contingent consideration, appropriate discount rate or other factors.

#### Royalties and other operating income

	Third Qu	ıarter		
	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Royalties and other operating income	\$ 3,844	\$ 3,837	\$ 7	0.2%

Royalties and other operating income in the third quarter of fiscal 2012 primarily reflects income received from third parties from the licensing of our Tommy Bahama, Ben Sherman and Lilly Pulitzer brands, which were comparable on a consolidated basis to the royalty income recognized in the third quarter of fiscal 2011.

#### Operating income (loss)

	Third Quarter						
	Fiscal 2	2012	Fiscal	2011	\$ Chan	ge	% Change
Tommy Bahama	\$	3,366	\$	4,624	\$	(1,258)	(27.2)%
Lilly Pulitzer		3,528		(363)		3,891	NM
Ben Sherman		(2,149)		301		(2,450)	NM
Lanier Clothes		2,402		4,331		(1,929)	(44.5)%
Corporate and Other		(1,227)		(2,077)		850	40.9%
Total operating income	\$	5,920	\$	6,816	\$	(896)	(13.1)%
LIFO (credits) charges included in operating income	\$	(426)	\$	220			
Charge for increase in fair value of contingent consideration included in operating income	\$	600	\$	600			
		17					

Operating income, on a consolidated basis, decreased to \$5.9 million in the third quarter of fiscal 2012 from \$6.8 million in the third quarter of fiscal 2011. The 13.1% decrease in operating income was primarily due to the lower operating results in Ben Sherman and Lanier Clothes and SG&A increases in Tommy Bahama and Lilly Pulitzer related to expansion of these brands. These items were partially offset by higher net sales in Tommy Bahama and Lilly Pulitzer. Changes in operating income by operating group are discussed below.

Tommy Bahama:

	Third Quarter						
	Fiscal	2012	Fiscal	2011	\$ Cha	inge	% Change
Net sales	\$	103,193	\$	92,500	\$	10,693	11.6%
Operating income	\$	3,366	\$	4,624	\$	(1,258)	(27.2)%
Operating income as % of net sales		3.3%		5.0%			

The decrease in operating income for Tommy Bahama was primarily due to increased SG&A associated with (1) certain infrastructure, pre-opening retail store rent and other costs associated with Tommy Bahama's international expansion, (2) operating additional retail stores during the third quarter of fiscal 2012, (3) the pre-opening rent charge associated with our Tommy Bahama New York location, which in accordance with U.S. GAAP is required to be recognized as an ongoing expense of the store from the time we took possession of the space and (4) higher SG&A to support the growing Tommy Bahama business. The additional SG&A costs were partially offset by the impact of the increased net sales of Tommy Bahama products in the third quarter of fiscal 2012.

The third quarter of fiscal 2012 included operating losses of approximately \$4.0 million related to the Tommy Bahama international expansion and the Tommy Bahama New York store, compared to operating losses of \$1.2 million related to the Tommy Bahama international expansion and the Tommy Bahama New York store in the third quarter of fiscal 2011. The \$4.0 million operating loss related to the Tommy Bahama international expansion and the Tommy Bahama New York store reflect approximately \$5.1 million of SG&A costs partially offset by approximately \$1.1 million of gross margin related to sales in our international stores and royalty income during the third quarter of fiscal 2012.

Lilly Pulitzer:

		Third Q	uarter				
	Fiscal	2012	Fiscal	2011	\$ Ch	ange	% Change
Net sales	\$	26,939	\$	16,668	\$	10,271	61.6%
Operating income (loss)	\$	3,528	\$	(363)	\$	3,891	NM
Operating income (loss) as % of net sales		13.1%		(2.2)%			
Charge for increase in fair value of contingent consideration							
included in operating income	\$	600	\$	600			

The improved operating results for Lilly Pulitzer were primarily due to increased net sales in each distribution channel. The increased net sales were partially offset by increased SG&A associated with the cost of operating additional retail stores during the third quarter of fiscal 2012 and higher SG&A to support the growing Lilly Pulitzer business. Both periods were impacted by a charge of \$0.6 million for the change in the fair value of contingent consideration.

Ben Sherman:

Third Quarter					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change	
Net sales	\$ 19,781	\$ 25,191	\$ (5,410)	(21.5)%	
Operating income (loss)	\$ (2,149)	\$ 301	\$ (2,450)	NM	
Operating income (loss) as % of net sales	(10.9)%	1.2%			

The decline in operating results for Ben Sherman in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 was primarily due to the decreased sales, as discussed above, partially offset by decreased SG&A.

#### Lanier Clothes:

	Third Qu			
	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Net sales	\$ 27,180	\$ 33,080	\$ (5,900)	(17.8)%
Operating income	\$ 2,402	\$ 4,331	\$ (1,929)	(44.5)%
Operating income as % of net sales	8.8%	13.1%		

The decrease in operating income for Lanier Clothes was primarily the result of the lower sales and gross margins, partially offset by decreased SG&A. Gross margin pressures resulted from both competitive factors and product cost pressures.

#### Corporate and Other:

	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Net sales	\$ 4,321	\$ 2,841	\$ 1,480	52.1%
Operating loss	\$ (1,227)	\$ (2,077)	\$ 850	40.9%
LIFO (credits) charges included in operating loss	\$ (426)	\$ 220		

The Corporate and Other operating results improved by \$0.9 million from a loss of \$2.1 million in the third quarter of fiscal 2011 to a loss of \$1.2 million in the third quarter of fiscal 2012. The improved operating results for fiscal 2012 reflect the net impact of LIFO accounting, with credits of \$0.4 million in the third quarter of fiscal 2012 and charges of \$0.2 million in the third quarter of fiscal 2011 as well as better operating results in our Oxford Golf business.

#### Interest expense, net

	Third Quarter					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change		
Interest expense, net	\$ 959	\$ 3,705	\$ (2,746)	(74.1)%		

Interest expense for the third quarter of fiscal 2012 decreased due to our borrowing at lower interest rates in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011. During the third quarter of fiscal 2012, substantially all of our borrowings were under our U.S. Revolving Credit Agreement, whereas substantially all of our borrowings in the third quarter of fiscal 2011 were from our  $11^3/8\%$  Senior Secured Notes. The change in the source of our borrowings resulted from our redemption of our  $11^3/8\%$  Senior Secured Notes in July 2012, which was funded with borrowings under our U.S. Revolving Credit Agreement and cash on hand. We anticipate that interest expense for the fourth quarter of fiscal 2012 will be approximately \$1.0 million.

#### Loss on repurchase of senior secured notes

Third Quarter						
	Fiscal 2012	Fiscal 2011	\$ Change	% Change		
Loss on repurchase of senior secured notes	\$ —	\$ 769	\$ (769)	NM		

In the third quarter of fiscal 2011, we repurchased in a privately negotiated transaction, \$5.0 million in aggregate principal amount of our  $11^3/8\%$  Senior Secured Notes. The repurchase of the  $11^3/8\%$  Senior Secured Notes and related write-off of unamortized deferred financing costs and discount resulted in a loss of approximately \$0.8 million in the third quarter of fiscal 2011. No loss on repurchase of senior secured notes was incurred in the third quarter of fiscal 2012 as we redeemed the remaining \$105.0 million in aggregate principal amount of our  $11^3/8\%$  Senior Secured Notes in the second quarter of fiscal 2012.

#### Income taxes

	Third Qu	Third Quarter					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change			
Income taxes	\$ 1,951	\$ 731	\$ 1,220	166.9%			
Effective tax rate	39.3%	31.2%					
	19						

Income tax expense for the third quarter of fiscal 2012 increased compared to the third quarter of fiscal 2011, primarily due to higher earnings in the third quarter of fiscal 2012 as well as an increase in the effective tax rate. Income taxes for the third quarter of fiscal 2012 were impacted by losses in foreign jurisdictions for which we were not able to recognize an income tax benefit and a greater proportion of our earnings being in jurisdictions with higher tax rates, which offset favorable discrete items during the period, including the reduction in income tax contingency reserves upon the expiration of the corresponding statute of limitations. Income taxes for the third quarter of fiscal 2011 were impacted by certain favorable discrete items, including the reduction of income tax contingency reserves upon the expiration of the corresponding statute of limitations, as well as the recognition of an income tax benefit for losses in foreign jurisdictions.

#### Net earnings

	Third Qua	rter	
	Fiscal 2012	Fiscal 2011	
Earnings from continuing operations	\$ 3,010	\$ 1,611	
Earnings from continuing operations per diluted common share	\$ 0.18	\$ 0.10	
Weighted average common shares outstanding-diluted	16,591	16,517	

The increase in earnings from continuing operations for the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 was primarily due to (1) the higher sales in Tommy Bahama and Lilly Pulitzer, (2) lower interest expense due to our third quarter fiscal 2012 borrowings being primarily under our U.S. Revolving Credit Agreement compared to our third quarter fiscal 2011 borrowings being primarily our  $11^3/8\%$  Senior Secured Notes, and (3) the absence of any loss on repurchase of senior secured notes in the third quarter of fiscal 2012. These items were partially offset by (1) lower sales at Lanier Clothes and Ben Sherman, (2) higher SG&A in Tommy Bahama and Lilly Pulitzer to support the continued growth and expansion of these brands, and (3) a higher effective tax rate during the third quarter of fiscal 2012. The third quarter of each fiscal year is generally our lowest net sales and earnings quarter compared to the other quarters during our fiscal year.

#### FIRST NINE MONTHS OF FISCAL 2012 COMPARED TO FIRST NINE MONTHS OF FISCAL 2011

The following table sets forth the specified line items in our unaudited condensed consolidated statements of earnings both in dollars (in thousands) and as a percentage of net sales. The table also sets forth the dollar change and the percentage change of the data as compared to the same period of the prior year. We have calculated all percentages based on actual data, but percentage columns may not add due to rounding. For purposes of the tables below, "NM" means not meaningful. Individual line items of our consolidated statements of earnings may not be directly comparable to those of our competitors, as classification of certain expenses may vary by company.

First Nine Months							
Fiscal 201	2	Fiscal 201	1	\$ Change			
\$ 619,296	100.0%	\$ 559,234	100.0%	\$ 60,062	10.7%		
274,980	44.4%	249,897	44.7%	25,083	10.0%		
344,316	55.6%	309,337	55.3%	34,979	11.3%		
295,656	47.7%	264,947	47.4%	30,709	11.6%		
1,800	0.3%	1,800	0.3%				
12,166	2.0%	12,650	2.3%	(484)	(3.8)%		
59,026	9.5%	55,240	9.9%	3,786	6.9%		
7,876	1.3%	12,777	2.3%	(4,901)	(38.4)%		
9,143	1.5%	9,017	1.6%	126	1.4%		
42,007	6.8%	33,446	6.0%	8,561	25.6%		
15,967	2.6%	11,255	2.0%	4,712	41.9%		
26,040	4.2%	22,191	4.0%	3,849	17.3%		
	\$ 619,296 274,980 344,316 295,656 1,800 12,166 59,026 7,876 9,143 42,007 15,967	Fiscal 2012 \$ 619,296	Fiscal 2012         Fiscal 201           \$ 619,296         100.0%         \$ 559,234           274,980         44.4%         249,897           344,316         55.6%         309,337           295,656         47.7%         264,947           1,800         0.3%         1,800           12,166         2.0%         12,650           59,026         9.5%         55,240           7,876         1.3%         12,777           9,143         1.5%         9,017           42,007         6.8%         33,446           15,967         2.6%         11,255	Fiscal 2012         Fiscal 2011           \$ 619,296         100.0%         \$ 559,234         100.0%           274,980         44.4%         249,897         44.7%           344,316         55.6%         309,337         55.3%           295,656         47.7%         264,947         47.4%           1,800         0.3%         1,800         0.3%           12,166         2.0%         12,650         2.3%           59,026         9.5%         55,240         9.9%           7,876         1.3%         12,777         2.3%           9,143         1.5%         9,017         1.6%           42,007         6.8%         33,446         6.0%           15,967         2.6%         11,255         2.0%	Fiscal 2012         Fiscal 2011         \$ Change           \$ 619,296         100.0%         \$ 559,234         100.0%         \$ 60,062           274,980         44.4%         249,897         44.7%         25,083           344,316         55.6%         309,337         55.3%         34,979           295,656         47.7%         264,947         47.4%         30,709           1,800         0.3%         1,800         0.3%         —           12,166         2.0%         12,650         2.3%         (484)           59,026         9.5%         55,240         9.9%         3,786           7,876         1.3%         12,777         2.3%         (4,901)           9,143         1.5%         9,017         1.6%         126           42,007         6.8%         33,446         6.0%         8,561           15,967         2.6%         11,255         2.0%         4,712		

The discussion and tables below compare certain line items included in our statements of earnings for the first nine months of fiscal 2012 to the first nine months of fiscal 2011. Each dollar and percentage change provided reflects the change between these periods unless indicated otherwise. Each dollar and share amount included in the tables is in thousands except for per share amounts.

#### Net Sales

	First Nine Months				
	Fiscal 2012	Fiscal 2011	\$ Change	% Change	
Tommy Bahama	\$ 371,790	\$ 324,546	\$ 47,244	14.6%	
Lilly Pulitzer	93,475	71,364	22,111	31.0%	
Ben Sherman	57,234	65,505	(8,271)	(12.6)%	
Lanier Clothes	84,995	88,995	(4,000)	(4.5)%	
Corporate and Other	11,802	8,824	2,978	33.7%	
Total net sales	\$ 619,296	\$ 559,234	\$ 60,062	10.7%	

Consolidated net sales increased \$60.1 million, or 10.7%, in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 primarily due to the increase in net sales at Tommy Bahama and Lilly Pulitzer, partially offset by the decrease in net sales at Ben Sherman and Lanier Clothes, each as discussed below.

#### Tommy Bahama:

The 14.6% increase in net sales for Tommy Bahama was primarily driven by increased sales in both the direct to consumer and wholesale distribution channels with the largest increase in the direct to consumer business. The increased sales in the direct to consumer business resulted from a significant increase in e-commerce sales, net sales at the additional retail stores operated during fiscal 2012 and a high single digit percentage increase in comparable full-price retail store sales for the first nine months of fiscal 2012. Tommy Bahama apparel unit sales increased by 12.8% due to the higher volume in each distribution channel, and the average selling price per unit increased by 3.0%, primarily as a result of the higher proportion of net sales from the direct to consumer channel of distribution. As of October 27, 2012, Tommy Bahama operated 110 retail stores compared to 94 retail stores as of October 29, 2011.

#### Lilly Pulitzer:

The 31.0% increase in net sales for Lilly Pulitzer was driven by increased sales in both the direct to consumer and wholesale distribution channels. The increases in the direct to consumer channel included a significant increase in e-commerce sales, sales at additional stores operated in fiscal 2012 and a low double digit percentage increase in comparable retail store sales. The higher e-commerce sales benefitted from a very strong end of season flash e-commerce sale which generated \$6.1 million of sales in the first nine months of fiscal 2012 compared to \$2.0 million of end of season flash sales in the first nine months of fiscal 2011 and also benefitted from increased full-price ecommerce sales during the nine month period. Unit sales increased by 37.9% due to the higher volume in each distribution channel, while the average selling price per unit decreased by 5.0% despite a greater percentage of Lilly Pulitzer sales being direct to consumer sales. The decreased selling price per unit resulted from a change in product mix as sportswear and knit dresses, both of which generally sell at lower price points than woven dresses, represented a greater proportion of the Lilly Pulitzer business and the impact of the e-commerce sale which sold product at prices more in line with typical wholesale prices rather than full price direct to consumer prices. As of October 27, 2012, Lilly Pulitzer operated 18 retail stores compared to 16 retail stores as of October 29, 2011.

#### Ben Sherman:

Net sales for Ben Sherman decreased by 12.6% in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011. The decrease in net sales for Ben Sherman was driven by a reduction in unit volume of 19.7% primarily attributable to (1) missteps in Ben Sherman's merchandise assortment planning for the third quarter, which resulted in too much of the product offering in styles at the higher end of the price range, particularly in the current economic environment, (2) the difficult economic conditions in the United Kingdom and Europe and (3) our exit from certain moderate wholesale accounts in the United Kingdom. The reduction in units sold was partially offset by an increase in the average selling price per unit of 8.7%. The increase in average selling price per unit was primarily due to (1) a greater proportion of Ben Sherman's sales being retail sales, which generally have higher selling prices, and (2) our strategy to improve the distribution of the brand in the United Kingdom by exiting certain moderate wholesale accounts and increasing product prices, which were partially offset by an unfavorable foreign currency translation impact resulting from a 1.9% change in average exchange rates between the two periods.

#### Lanier Clothes:

Net sales for Lanier Clothes decreased by 4.5% in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011. A decrease in unit sales of 8.4% was primarily driven by the decreased unit sales in the private label businesses, which was partially offset by an increase in unit sales of branded tailored clothing products. The decrease in unit sales in private label was primarily due to the first nine months of fiscal 2011 benefitting from initial shipments related to a new product launch, while the first nine months of fiscal 2012 were impacted by a slow-down in the inventory intake on a replenishment program by a key customer. The average selling price per unit increased 4.2% as a result of the change in product sales mix towards branded apparel, which generally has a higher average selling price per unit.

#### Corporate and Other:

Corporate and Other net sales primarily consisted of the net sales of our Oxford Golf business and our Lyons, Georgia distribution center. The increase in the net sales for Corporate and Other was primarily driven by the higher net sales in our Oxford Golf business during the first nine months of fiscal 2012.

#### Gross Profit

	First Nine Months					
	Fiscal 201	12	Fiscal 2	011	\$ Change	% Change
Gross profit	\$ 344,	,316	\$ 30	9,337	\$ 34,979	11.3%
Gross margin (gross profit as a % of net sales)	:	55.6%		55.3%		
LIFO (credits) charges included in gross profit	\$ (	(461)	\$	6		
Purchase accounting charge related to write-up of acquired inventory						
included in gross profit	\$	_	\$	996		

The 11.3% increase in gross profit in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 was primarily due to the 10.7% increase in net sales, as discussed above. The first nine months of fiscal 2012 experienced a modest increase in gross margins primarily resulting from a change in sales mix with an increase in the proportion of total sales at our higher gross margin Tommy Bahama and Lilly Pulitzer brands and a higher proportion of sales from our higher gross margin direct to consumer businesses. The first nine months of fiscal 2012 included LIFO accounting credits of \$0.5 million while the first nine months of fiscal 2011 included purchase accounting charges of \$1.0 million. The impact of these favorable items on gross margins in the first nine months of fiscal 2012 were partially offset by the higher product cost pressures during the first six months of fiscal 2012 compared to the first six months of fiscal 2011.

Certain of the year-over-year product cost pressures that impacted our operating groups during the second half of fiscal 2011 and the first half of fiscal 2012 began to ease in the third quarter of fiscal 2012 as the impact of certain of the higher product costs moderated. We expect that certain competitive factors impacting Lanier Clothes and Ben Sherman will persist, which may continue to negatively impact the gross margins in those operating groups. Our gross profit may not be directly comparable to those of our competitors, as statement of earnings classification of certain expenses may vary by company.

#### SG&A

	First Nine Months				
	Fiscal 2012	Fiscal 2011	\$ Change	% Change	
SG&A	\$ 295,656	\$ 264,947	\$ 30,709	11.6%	
SG&A (as % of net sales)	47.7%	47.4%			

The increase in SG&A was primarily due to (1) the incremental SG&A in the first nine months of fiscal 2012 associated with operating additional Tommy Bahama and Lilly Pulitzer retail stores, (2) certain infrastructure, pre-opening retail store rent and other costs related to the Tommy Bahama international expansion, (3) the pre-opening rent charge associated with our Tommy Bahama New York location, which in accordance with U.S. GAAP was required to be recognized as an ongoing expense of the store at the time we took possession of the space and (4) higher SG&A to support the growing Tommy Bahama and Lilly Pulitzer businesses. The first nine months of fiscal 2012 included SG&A charges of approximately \$11.6 million, compared to \$2.0 million related to the Tommy Bahama international expansion and the Tommy Bahama New York store in the first nine months of fiscal 2011. The increases in SG&A in Tommy Bahama and Lilly Pulitzer were partially offset by reductions in SG&A in Lanier Clothes during

the first nine months of fiscal 2012. SG&A for the first nine months of fiscal 2012 and the first nine months of fiscal 2011 included charges of \$0.8 million and \$0.9 million, respectively, related to the amortization of intangible assets.

#### Change in fair value of contingent consideration

First Nine Months					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change	
Change in fair value of contingent consideration	\$ 1,800	\$ 1,800	\$ —	%	

Change in fair value of contingent consideration reflects the current period impact of the change in fair value of the contingent consideration obligation associated with the Lilly Pulitzer acquisition, as discussed in our Annual Report on Form 10-K for fiscal 2011. We anticipate that the change in contingent consideration for the full year of fiscal 2012 will be approximately \$2.4 million; however, that amount could change significantly depending upon whether there are any changes in future periods to our assumptions about the probability of payment of the contingent consideration, appropriate discount rate or other factors.

#### Royalties and other operating income

First Nine Months					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change	
Royalties and other operating income	\$ 12,166	\$ 12,650	\$ (484)	(3.8)%	

The decrease in royalties and other operating income was primarily due to lower royalty income in the Ben Sherman and Tommy Bahama businesses, partially offset by increased royalty income at Lilly Pulitzer during the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011.

#### Operating income (loss)

	First Nine Months						
	Fiscal 2012	Fiscal 2	011	\$ Change	% Change		
Tommy Bahama	\$ 45,51	1 \$	45,381	\$ 130	0.3%		
Lilly Pulitzer	21,94	9	12,264	9,685	79.0%		
Ben Sherman	(6,35	2)	(2,281)	(4,071)	(178.5)%		
Lanier Clothes	8,84	5	11,319	(2,474)	(21.9)%		
Corporate and Other	(10,92	7)	(11,443)	516	4.5%		
Total operating income	\$ 59,02	6 \$	55,240	\$ 3,786	6.9%		
LIFO (credits) charges included in operating income	\$ (46	1) \$	6				
Purchase accounting charge related to write-up of acquired inventory included in operating income	\$ -	- \$	996				
Charge for increase in fair value of contingent consideration included in operating income	\$ 1,80	\$	1,800				

Operating income, on a consolidated basis, increased to \$59.0 million in the first nine months of fiscal 2012 from \$55.2 million in the first nine months of fiscal 2011. The 6.9% increase in operating income was primarily due to (1) higher net sales in Lilly Pulitzer and Tommy Bahama and (2) the first nine months of fiscal 2011 including an approximate \$1.0 million charge for the write-up of acquired inventory at Lilly Pulitzer. These positive items were partially offset by lower operating results in Ben Sherman and Lanier Clothes and the SG&A increases in Tommy Bahama and Lilly Pulitzer related to the expansion of these brands. Changes in operating income by operating group are discussed below.

#### Tommy Bahama:

	First Nine			
	Fiscal 2012	Fiscal 2011	\$ Change	% Change
Net sales	\$ 371,790	\$ 324,546	\$ 47,244	14.6%
Operating income	\$ 45,511	\$ 45,381	\$ 130	0.3%
Operating income as % of net sales	12.2%	14.0%		

The increase in operating income for Tommy Bahama was primarily due to higher sales in Tommy Bahama for the first nine months of fiscal 2012. The impact of the higher sales was partially offset by increased SG&A associated with (1) operating additional retail stores, (2) certain infrastructure, preopening retail store rent and other costs associated with Tommy Bahama's international expansion, (3) the pre-opening rent charge associated with our Tommy Bahama New York location, which in accordance with U.S. GAAP is required to be recognized as an ongoing expense of the store at the time we took possession of the space and (4) higher SG&A to support the growing Tommy Bahama business.

The first nine months of fiscal 2012 and the first nine months of fiscal 2011 included operating losses of approximately \$9.8 million and \$1.9 million, respectively, related to the Tommy Bahama international expansion and the Tommy Bahama New York store, which opened in the fourth quarter of fiscal 2012. The \$9.8 million operating loss attributable to the Tommy Bahama international expansion and the Tommy Bahama New York store reflect approximately \$11.6 million of SG&A costs partially offset by approximately \$1.8 million of gross margin related to sales in our international stores and royalty income during the first nine months of fiscal 2012.

#### Lilly Pulitzer:

First Nine Months							
	Fiscal	2012	Fiscal	2011	\$ Ch	ange	% Change
Net sales	\$	93,475	\$	71,364	\$	22,111	31.0%
Operating income	\$	21,949	\$	12,264	\$	9,685	79.0%
Operating income as % of net sales		23.5%		17.2%			
Purchase accounting charge related to write-up of acquired							_
inventory included in operating income	\$	_	\$	996			
Charge for increase in fair value of contingent consideration							
included in operating income	\$	1,800	\$	1,800			

The increase in operating income for Lilly Pulitzer was primarily due to increased net sales and the absence of the \$1.0 million purchase accounting charge which was incurred in the first nine months of fiscal 2011. The increased net sales were partially offset by increased SG&A associated with the cost of operating additional retail stores during the first nine months of fiscal 2012 and higher SG&A to support the growing Lilly Pulitzer business. The \$1.0 million of purchase accounting charges to cost of goods sold in the first nine months of fiscal 2011 resulted from the write-up of acquired inventory to fair value pursuant to the purchase method of accounting. Both periods were impacted by a charge of \$1.8 million for the change in the fair value of contingent consideration.

#### Ben Sherman:

	First Nine	First Nine Months					
	Fiscal 2012	Fiscal 2011	\$ Change	% Change			
Net sales	\$ 57,234	\$ 65,505	\$ (8,271)	(12.6)%			
Operating loss	\$ (6,352)	\$ (2,281)	\$ (4,071)	(178.5)%			
Operating loss as % of net sales	(11.1)%	(3.5)%					

The decline in operating results for Ben Sherman in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 was primarily due to (1) lower net sales, as discussed above, and (2) lower royalty income. These items were partially offset by improved gross margins in the Ben Sherman business in the first nine months of fiscal 2012.

#### Lanier Clothes:

First Nine Months						
	Fiscal 2012	Fiscal 2011	\$ Change	% Change		
Net sales	\$ 84,995	\$ 88,995	\$ (4,000)	(4.5)%		
Operating income	\$ 8,845	\$ 11,319	\$ (2,474)	(21.9)%		
Operating income as % of net sales	10.4%	12.7%				

The decrease in operating income for Lanier Clothes was primarily the result of decreased net sales and gross margin erosion, partially offset by decreased SG&A. The gross margin erosion resulted from both competitive factors and cost pressures, which continued to impact the operating results of Lanier Clothes.

#### Corporate and Other:

First Nine Months							
	Fiscal 2012	Fiscal 2011	\$ Change	% Change			
Net sales	\$ 11,802	\$ 8,824	\$ 2,978	33.7%			
Operating loss	\$ (10,927)	\$ (11,443)	\$ 516	4.5%			
LIFO (credits) charges included in operating loss	\$ (461)	\$ 6					

The Corporate and Other operating results improved by \$0.5 million from a loss of \$11.4 million in the first nine months of fiscal 2011 to a loss of \$10.9 million in the first nine months of fiscal 2012. The improvement in operating results for the first nine months of fiscal 2012 reflects the net impact of LIFO accounting, with credits of \$0.5 million in the first nine months of fiscal 2012 and charges of \$0.0 million in the first nine months of fiscal 2011.

#### Interest expense, net

First Nine Months							
	Fiscal 2012	Fiscal 2011	\$ Change	% Change			
Interest expense, net	\$ 7,876	\$ 12,777	\$ (4,901)	(38.4)%			

Interest expense for the first nine months of fiscal 2012 decreased due to (1) our borrowing at lower interest rates in the third quarter of fiscal 2012 compared to the third quarter of fiscal 2011 and (2) the reduction in our average debt levels in the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 as a result of our repurchase of \$45.0 million in aggregate principal amount of our  $11^3/8\%$  Senior Secured Notes during the second and third quarters of fiscal 2011. During the third quarter of fiscal 2012, substantially all of our borrowings were under our U.S. Revolving Credit Agreement, whereas substantially all of our borrowings in the third quarter of fiscal 2011 were from our  $11^3/8\%$  Senior Secured Notes. The change in the source of our borrowings resulted from our redemption of our  $11^3/8\%$  Senior Secured Notes in July 2012, which was funded with borrowings under our U.S. Revolving Credit Agreement and cash on hand. We anticipate that interest expense for the fourth quarter of fiscal 2012 will be approximately \$1.0 million.

#### Loss on repurchase of senior secured notes

First Nine Months							
	Fiscal 2012	Fiscal 2011	\$ Change	% Change			
Loss on repurchase of senior secured notes	\$ 9,143	\$ 9,017	\$ 126	1.4%			

On July 16, 2012, we redeemed the remaining outstanding \$105.0 million in aggregate principal amount of our  $11^3/8\%$  Senior Secured Notes for approximately \$111.0 million, plus accrued interest, using borrowings under our U.S. Revolving Credit Agreement and cash on hand. The redemption of the  $11^3/8\%$  Senior Secured Notes and related write-off of approximately \$3.1 million of unamortized deferred financing costs and discount resulted in a loss of approximately \$9.1 million.

In the second and third quarters of fiscal 2011, we repurchased, in privately negotiated transactions, \$45.0 million in aggregate principal amount of our 113/8% Senior Secured Notes for \$52.2 million, plus accrued interest. The repurchase of the 113/8% Senior Secured Notes and related write-off of approximately \$1.8 million of unamortized deferred financing costs and discount resulted in a loss of approximately \$9.0 million.

#### Income taxes

First Nine Months						
	Fiscal 2012	Fiscal 2011	\$ Change	% Change		
Income taxes	\$ 15,967	\$ 11,255	\$ 4,712	41.9%		
Effective tax rate	38.0%	33.7%				

Income tax expense for the first nine months of fiscal 2012 increased compared to the first nine months of fiscal 2011, primarily due to higher earnings in the first nine months of fiscal 2012 as well as an increase in the effective tax rate. Income taxes for the first nine months of fiscal 2012 were impacted by losses in foreign jurisdictions for which we were not able to recognize an income tax benefit and a greater proportion of our earnings being in jurisdictions with high tax rates, which were offset by favorable discrete items, including the reduction of income tax

contingency reserves upon the expiration of the corresponding statute of limitations and changes in the enacted tax rates in certain jurisdictions. Income taxes for the first nine months of fiscal 2011 were impacted by certain favorable discrete items, including the reduction of income tax contingency reserves upon the expiration of the corresponding statute of limitations and changes in the enacted tax rates in certain jurisdictions, as well as recognition of an income tax benefit for losses in foreign jurisdictions.

#### Net earnings

	First Nine	First Nine Months			
	Fiscal 2012	Fiscal 2011			
Earnings from continuing operations	\$ 26,040	\$ 22,191			
Earnings from continuing operations per diluted common share	\$ 1.57	\$ 1.34			
Weighted average common shares outstanding-diluted	16,572	16,527			

The increase in earnings from continuing operations for the first nine months of fiscal 2012 compared to the first nine months of fiscal 2011 was primarily due to (1) the higher sales in Tommy Bahama and Lilly Pulitzer and (2) lower interest expense resulting from lower interest rates on our borrowings and lower average debt levels. These items were partially offset by (1) lower sales at Lanier Clothes and Ben Sherman, (2) higher SG&A in Tommy Bahama and Lilly Pulitzer to support the continued growth and expansion of these brands, and (3) a higher effective tax rate during the first nine months of fiscal 2012.

#### FINANCIAL CONDITION, LIQUIDITY AND CAPITAL RESOURCES

Our primary source of revenue and cash flow is our operating activities in the United States and, to a lesser extent, the United Kingdom. Our primary uses of cash include the acquisition of apparel products in the operation of our business, as well as employee compensation and benefits, occupancy costs, marketing and advertising costs, other general and administrative operating expenses, funding of capital expenditures for retail stores, payment of quarterly dividends, periodic interest payments related to our financing arrangements and repayment of our indebtedness. If cash inflows are less than cash outflows, we have access to amounts under our U.S. Revolving Credit Agreement and U.K. Revolving Credit Agreement, subject to their terms, each of which is described below. We may seek to finance future capital investment programs through various methods, including, but not limited to, cash on hand, cash flow from operations, borrowings under our current or additional credit facilities and sales of debt or equity securities.

#### **Key Liquidity Measures**

(\$ in thousands)	Octobe	er 27, 2012	January	28, 2012	October	29, 2011	January	y 29, 2011
Current assets	\$	217,291	\$	215,273	\$	197,358	\$	268,913
Current liabilities		109,710		117,554		103,236		147,091
Working capital	\$	107,581	\$	97,719	\$	94,122	\$	121,822
Working capital ratio		1.98		1.83		1.91		1.83
Debt to total capital ratio		36%		34%		35%		45%

Our working capital ratio is calculated by dividing total current assets by total current liabilities, including the assets and liabilities related to discontinued operations. Current assets increased from October 29, 2011 to October 27, 2012, primarily as a result of higher inventory levels at October 27, 2012 as well as increases in receivables, prepaid expenses and deferred tax assets.

For the ratio of debt to total capital, debt is defined as short-term and long-term debt, and total capital is defined as debt plus shareholders' equity. Debt was \$130.3 million at October 27, 2012 and \$106.6 million at October 29, 2011, while shareholders' equity was \$227.3 million at October 27, 2012 and \$198.4 million at October 29, 2011. The change in the debt to total capital ratio from October 29, 2011 to October 27, 2012 reflects an increase in debt as well as an increase in shareholders' equity. The increase in debt was primarily a result of the significant capital expenditures incurred subsequent to October 29, 2011 and the \$6.0 million premium required to redeem our senior secured notes in the second quarter of fiscal 2012, which exceeded cash flows from operations during the period subsequent to October 29, 2011. The increase in shareholders' equity was primarily a result of earnings from continuing operations subsequent to October 29, 2011. Our debt levels and ratio of debt to total capital in future periods may not be comparable to historical amounts as we continue to assess, and possibly make changes to, our

capital structure. Changes in our capital structure in the future, if any, will depend on prevailing market conditions, our liquidity requirements, contractual restrictions and other factors. The amounts involved may be material.

#### **Balance Sheet**

The following tables set forth certain information included in our consolidated balance sheets (in thousands). Below each table are explanations for any significant changes in the balances from October 29, 2011 to October 27, 2012.

#### Current Assets:

	October 27, 2012	January 28, 2012	October 29, 2011	January 29, 2011
Cash and cash equivalents	\$ 5,621	\$ 13,373	\$ 4,962	\$ 44,094
Receivables, net	68,920	59,706	66,372	50,177
Inventories, net	102,172	103,420	91,003	85,338
Prepaid expenses, net	21,251	19,041	17,425	12,554
Deferred tax assets	19,327	19,733	17,596	19,005
Total current assets related to continuing operations	217,291	215,273	197,358	211,168
Assets related to discontinued operations	_	_	_	57,745
Total current assets	\$ 217,291	\$ 215,273	\$ 197,358	\$ 268,913

Cash and cash equivalents as of October 27, 2012 and October 29, 2011 reflect typical cash amounts used on an ongoing basis in our operations, with any excess cash generally being used to repay amounts outstanding under our revolving credit agreements, if any. Receivables, net as of October 27, 2012 increased compared to October 29, 2011 primarily due to the increased wholesale sales in our operating groups in the last two months of the third quarter of fiscal 2012 compared to the last two months of the third quarter of fiscal 2011 which was a result of timing of shipments within the quarter as well as an increase in wholesale sales. Inventories, net as of October 27, 2012 increased from October 29, 2011 primarily to support anticipated sales growth and additional retail stores for Tommy Bahama and Lilly Pulitzer. As of October 27, 2012, Ben Sherman inventory levels increased from the prior year, while Lanier Clothes inventory levels decreased from October 29, 2011. The increase in prepaid expenses, net from October 29, 2011 to October 27, 2012 was primarily due to the timing of payments of certain prepaid expenses. Deferred tax assets increased from October 29, 2011 primarily as a result of the change in timing differences associated with inventory.

#### Non-current Assets:

	October	27, 2012	January	28, 2012	October	29, 2011	January	29, 2011
Property and equipment, net	\$	123,841	\$	93,206	\$	91,121	\$	83,895
Intangible assets, net		165,013		165,193		166,082		166,680
Goodwill		17,273		16,495		16,555		16,866
Other non-current assets, net		21,404		19,040		18,385		22,117
Total non-current assets, net	\$	327,531	\$	293,934	\$	292,143	\$	289,558

The increase in property and equipment, net at October 27, 2012 was primarily due to capital expenditures subsequent to October 29, 2011, which were partially offset by depreciation expense subsequent to October 29, 2011. The decrease in intangible assets, net was primarily due to amortization of intangible assets associated with Tommy Bahama, Lilly Pulitzer and Ben Sherman subsequent to October 29, 2011. The increase in goodwill from October 29, 2011 was primarily related to the goodwill associated with our acquisition of the Tommy Bahama business in Australia from our former licensee that operated that business. The increase in other non-current assets was primarily due to security deposit payments for certain retail store lease agreements and higher asset balances set aside for potential deferred compensation obligations, partially offset by decreases in deferred financing costs.

#### Liabilities:

	October	r 27, 2012	January	28, 2012	October	29, 2011	January	29, 2011
Current liabilities related to continuing operations	\$	109,710	\$	117,554	\$	103,236	\$	106,306
Long-term debt, less current maturities		123,301		103,405		103,290		147,065
Non-current contingent consideration		9,945		10,645		12,545		10,745
Other non-current liabilities		43,107		38,652		41,328		44,696
Non-current deferred income taxes		31,459		34,882		30,738		28,846
Total liabilities related to continuing operations	\$	317,522	\$	305,138	\$	291,137	\$	337,658
Liabilities related to discontinued operations	\$	_	\$	_	\$	_	\$	40,785
Total liabilities	\$	317,522	\$	305,138	\$	291,137	\$	378,443

The change in current liabilities related to continuing operations at October 27, 2012 compared to October 29, 2011 was primarily due to the higher debt levels under our U.K. Revolving Credit Agreement and the classification of \$2.5 million of contingent consideration as a current liability as of October 27, 2012. The fiscal 2012 contingent consideration amount was classified as a current liability as of October 27, 2012 after consideration of the Lilly Pulitzer operating results for fiscal 2012 compared to the target earnings for fiscal 2012 and the likelihood of the payment being earned. The increase in debt at October 27, 2012 compared to October 29, 2011 was primarily a result of the significant capital expenditures incurred subsequent to October 29, 2011 and the \$6.0 million premium required to redeem our senior secured notes in the second quarter of fiscal 2012, which exceeded cash flows from operations during the period subsequent to October 29, 2011. The decrease in non-current contingent consideration from October 29, 2011 was primarily due to the payment of October 27, 2012, which were offset by a \$2.4 million charge related to the increase in the fair value of the contingent consideration subsequent to October 29, 2011.

Other non-current liabilities increased primarily due to increases in deferred rent and deferred compensation liabilities partially offset by reductions in reserves for uncertain tax positions upon the expiration of statutes of limitations. Non-current deferred income taxes increased from October 29, 2011 primarily as a result of the change in timing differences associated with depreciable and amortizable assets and deferred rent liabilities which were partially offset by the change in timing differences associated with compensation and the impact of changes in the effective tax rate at which certain timing differences are expected to reverse in the future.

#### Statement of Cash Flows

The following table sets forth the net cash flows for the first nine months of fiscal 2012 and the first nine months of fiscal 2011 (in thousands):

	First Nine Mont	ths
	Fiscal 2012	Fiscal 2011
Net cash provided by operating activities	\$ 34,676	\$ 19,434
Net cash used in investing activities	(51,966)	(22,846)
Net cash provided by (used in) financing activities	9,366	(53,297)
Net cash provided by discontinued operations	_	17,479
Net change in cash and cash equivalents	\$ (7,924)	\$ (39,230)

Cash and cash equivalents on hand was \$5.6 million and \$5.0 million at October 27, 2012 and October 29, 2011, respectively. Changes in cash flows in the first nine months of fiscal 2012 and the first nine months of fiscal 2011 related to operating activities, investing activities, financing activities and discontinued operations are discussed below.

#### Operating Activities:

In the first nine months of fiscal 2012, operating activities generated \$34.7 million of cash, while in the first nine months of fiscal 2011 operating activities generated \$19.4 million of cash, with the increase in cash flow from operating activities for the first nine months of fiscal 2012 primarily being due to an increase in net earnings and more favorable changes in working capital accounts as compared to the prior year. The cash flow from operating activities was primarily the result of net earnings for the relevant period, adjusted for non-cash activities such as depreciation, amortization, stock compensation expense and change in fair value of contingent consideration as well as the loss on repurchase of senior secured notes and the net impact of changes in our working capital accounts. In the first nine months of fiscal 2012, the more significant changes in working capital were a decrease in current liabilities and an increase in receivables, both of which decreased cash. In the first nine months of fiscal 2011, the more significant changes in working capital were an increase in receivables, inventories and prepaid expenses and a decrease in current liabilities, each of which decreased cash.

#### Investing Activities:

During the first nine months of fiscal 2012 and the first nine months of fiscal 2011, investing activities used \$52.0 million and \$22.8 million, respectively, of cash. During the first nine months of fiscal 2012 and the first nine months of fiscal 2011, \$47.7 million and \$22.4 million, respectively, of cash was used for capital expenditures primarily related to costs associated with new retail stores, information technology initiatives, retail store and restaurant remodeling and distribution center enhancements. During the first nine months of fiscal 2012, we paid approximately \$4.3 million for acquisitions consisting of \$2.5 million for the payment of the fiscal 2011 contingent consideration payment related to the Lilly Pulitzer acquisition and \$1.8 million related to our acquisition of the assets and operations of the Tommy Bahama business in Australia from our former licensee that operated that business.

#### Financing Activities:

During the first nine months of fiscal 2012 financing activities provided \$9.4 million of cash, while in the first nine months of fiscal 2011, financing activities used \$53.3 million of cash with changes in debt being the most significant changes in financing activities during each period. In the first nine months of fiscal 2012, we increased debt by \$16.6 million, while replacing our borrowings under our senior secured notes with borrowings under our U.S. Revolving Credit Agreement. During the first nine months of fiscal 2011, we reduced debt by \$48.9 million by using cash on hand to repurchase a portion of our senior secured notes. We used \$7.4 million and \$6.4 million of cash to pay dividends during the first nine months of fiscal 2012 and the first nine months of fiscal 2011, respectively.

#### Discontinued Operations:

The cash flows provided by discontinued operations during fiscal 2011 reflect cash flow provided by or used in the activities of our discontinued operations, which include the operations related to substantially all of our former Oxford Apparel operating group. There were no cash flows from discontinued operations in the first nine months of fiscal 2012, while the cash flow from discontinued operations in the first nine months of fiscal 2011 primarily reflects the conversion of assets related to the discontinued operations into cash, net of the use of cash to pay liabilities, including income taxes, associated with the sold business during fiscal 2011.

#### Liquidity and Capital Resources

The table below provides a description of our significant financing arrangements and the amounts outstanding under these financing arrangements (in thousands) as of October 27, 2012:

\$235 million U.S. Secured Revolving Credit Facility ("U.S. Revolving Credit Agreement")	\$ 123,301
£7 million Senior Secured Revolving Credit Facility ("U.K. Revolving Credit Agreement")	6,955
Total debt	130,256
Short-term debt and current maturities of long-term debt	(6,955)
Long-term debt, less current maturities	\$ 123,301

The U.S. Revolving Credit Agreement, entered into in June 2012, amended and restated our Prior Revolving Credit Agreement, which was scheduled to mature in August 2013. The U.S. Revolving Credit Agreement generally (i) is limited to a borrowing base consisting of specified percentages of eligible categories of assets; (ii) accrues interest, unused line fees and letter of credit fees based upon a pricing grid which is tied to average unused availability and/or utilization; (iii) requires periodic interest payments with principal due at maturity (June 2017); and (iv) is generally secured by a first priority security interest in the accounts receivable, inventory, general intangibles and eligible trademarks, investment property (including the equity interests of certain subsidiaries), deposit accounts, intercompany obligations, equipment, goods, documents, contracts, books and records and other personal property of Oxford Industries, Inc. and substantially all of its domestic subsidiaries.

The U.K. Revolving Credit Agreement generally (i) accrues interest at the bank's base rate plus an applicable margin; (ii) requires interest payments monthly with principal payable on demand; and (iii) is collateralized by substantially all of the assets of our United Kingdom Ben Sherman subsidiaries.

To the extent cash flow needs exceed cash flow provided by our operations we will have access, subject to their terms, to our lines of credit to provide funding for operating activities, capital expenditures and acquisitions, if any. Our credit facilities are also used to finance trade letters of credit for product purchases, which are drawn against our lines of credit at the time of shipment of the products and reduce the amounts available under our lines of credit and borrowing capacity under our credit facilities when issued. As of October 27, 2012, approximately \$7.8 million of trade letters of credit and other limitations on availability in the aggregate were outstanding against our credit facilities. After considering these limitations and the amount of eligible assets in our borrowing base, as applicable, as of October 27, 2012, we had approximately \$90.0 million and \$1.4 million in unused availability under the U.S. Revolving Credit Agreement and the U.K. Revolving Credit Agreement, respectively, subject to the respective limitations on borrowings set forth in the U.S. Revolving Credit Agreement and the U.K. Revolving Credit Agreement.

#### Covenants and Other Restrictions:

Our credit facilities, consisting of our U.S. Revolving Credit Agreement and our U.K. Revolving Credit Agreement, are subject to a number of affirmative covenants regarding the delivery of financial information, compliance with law, maintenance of property, insurance and conduct of business. Also, our credit facilities are subject to certain negative covenants or other restrictions including, among other things, limitations on our ability to (i) incur debt, (ii) guaranty certain obligations, (iii) incur liens, (iv) pay dividends to shareholders, (v) repurchase shares of our common stock, (vi) make investments, (vii) sell assets or stock of subsidiaries, (viii) acquire assets or businesses, (ix) merge or consolidate with other companies, or (x) prepay, retire, repurchase or redeem debt.

Our U.S. Revolving Credit Agreement contains a financial covenant that applies if unused availability under the U.S. Revolving Credit Agreement for three consecutive days is less than the greater of (i) \$23.5 million or (ii) 10% of the total revolving commitments. In such case, our fixed charge coverage ratio as defined in the U.S. Revolving Credit Agreement must not be less than 1.0 to 1.0 for the immediately preceding 12 fiscal months for which financial statements have been delivered. This financial covenant continues to apply until we have maintained unused availability under the U.S. Revolving Credit Agreement of more than the greater of (i) \$23.5 million or (ii) 10% of the total revolving commitments for 30 consecutive days.

We believe that the affirmative covenants, negative covenants, financial covenants and other restrictions under our credit facilities are customary for those included in similar facilities entered into at the time we entered into our agreements. During the first nine months of fiscal 2012 and as of October 27, 2012, no financial covenant testing was required pursuant to our U.S. Revolving Credit Agreement as the minimum availability threshold was met at all times. As of October 27, 2012, we were compliant with all covenants related to our credit facilities.

#### Redemption of Senior Notes:

In the second quarter of fiscal 2012, we redeemed all of the remaining outstanding \$105 million in aggregate principal amount of the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes, which were scheduled to mature in July 2015. The redemption of the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes at a premium of \$6.0 million and the write-off of approximately \$3.1 million of unamortized deferred financing costs and unamortized bond discount resulted in a loss on repurchase of senior notes of approximately \$9.1 million. The redemption of the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes satisfied and discharged all of our obligations with respect to the 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes and the related indenture and was funded through borrowings under our U.S. Revolving Credit Agreement and also from cash on hand.

#### Other Liquidity Items:

We anticipate that we will be able to satisfy our ongoing cash requirements, which generally consist of working capital and other operating activity needs, capital expenditures, interest payments on our debt and dividends, if any, primarily from positive cash flow from operations supplemented by cash on hand and borrowings under our lines of credit, if necessary. Our need for working capital is typically seasonal with the greatest requirements generally existing in the fall and spring of each year. Our capital needs will depend on many factors including our growth rate, the need to finance inventory levels and the success of our various products. We anticipate that at the maturity of any of our financing arrangements or as otherwise deemed appropriate, we will be able to refinance the facilities and debt with terms available in the market at that time, which may or may not be as favorable as the terms of the current agreements or current market terms.

Our contractual obligations as of October 27, 2012 have not changed significantly from the contractual obligations outstanding at July 28, 2012, as disclosed in our Quarterly Report on Form 10-Q for the second quarter of fiscal 2012 filed with the SEC, other than changes in the amounts outstanding under our revolving credit agreements, as discussed above.

Our anticipated capital expenditures for fiscal 2012, including the \$47.7 million incurred in the first nine months of fiscal 2012, are expected to approach \$60 million. These expenditures consist primarily of costs associated with opening new retail stores, information technology initiatives, retail store remodeling, and distribution center enhancements.

#### Off Balance Sheet Arrangements

We have not entered into agreements which meet the SEC's definition of an off balance sheet financing arrangement, other than operating leases, and have made no financial commitments to or guarantees with respect to any unconsolidated subsidiaries or special purpose entities.

#### CRITICAL ACCOUNTING POLICIES AND ESTIMATES

The discussion and analysis of our financial condition and results of operations are based upon our unaudited condensed consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses and related disclosures of contingent assets and liabilities. On an ongoing basis, we evaluate our estimates, including those related to bad debts, inventories, intangible assets, income taxes, stock compensation expense, contingencies and litigation and certain other accrued expenses. We base our estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. Our critical accounting policies and estimates are discussed in Part II, Item 7, Management's Discussion and Analysis of Financial Condition and Results of Operations in our Annual Report on Form 10-K for fiscal 2011.

There have not been any significant changes to the application of our critical accounting policies and estimates during the first nine months of fiscal 2012.

A detailed summary of significant accounting policies is included in Note 1 to our consolidated financial statements contained in our Annual Report on Form 10-K for fiscal 2011.

#### **SEASONALITY**

Although our various product lines are sold on a year-round basis, the demand for specific products or styles may be seasonal. For example, the demand in our direct to consumer operations for Tommy Bahama products is generally higher in the spring, summer and holiday seasons and lower in the fall season. Similarly, the demand in our direct to consumer operations for our Lilly Pulitzer products is generally higher in the spring and summer seasons and lower in the holiday and fall seasons. Also, wholesale product shipments are shipped prior to each of the retail selling seasons; as a result, the Tommy Bahama and Lilly Pulitzer wholesale businesses are typically stronger just prior to the stronger retail seasons of each brand. The sales and operating results of Ben Sherman are more consistent with a typical wholesale and retail apparel company whereby the fall and holiday seasons are typically stronger quarters than the first half of the fiscal year. As a wholesale tailored clothing business, the seasonality of Lanier Clothes reflects stronger spring and fall wholesale deliveries, which occur in our first and third fiscal quarters. We anticipate that as our retail store count increases in the future, the third quarter will continue to be our weakest net sales and operating income quarter and the percentage of the full year net sales and operating income generated in the third quarter will continue to decrease. The following table presents the percentage of net sales and operating income by quarter (unaudited) for fiscal 2011:

	First Quarter	Second Quarter	Third Quarter	Fourth Quarter
Net sales	27%	24%	23%	26%
Operating income (1)	44%	26%	10%	20%

(1) The fourth quarter of fiscal 2011 operating income included a \$5.8 million LIFO accounting charge.

As the timing of certain unusual or non-recurring items, economic conditions, wholesale product shipments or other factors affecting the retail business may vary from one year to the next, we do not believe that net sales or operating income for any particular quarter or the distribution of net sales and operating income for fiscal 2011 are necessarily indicative of anticipated results for the full fiscal year or expected distribution in future years.

#### ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to certain interest rate, foreign currency, commodity and inflation risks as discussed in Part II, Item 7A, Quantitative and Qualitative Disclosures About Market Risk in our Annual Report on Form 10-K for fiscal 2011. We do not believe there have been any significant changes to our exposure to foreign currency, commodity and inflation risks during the first nine months of fiscal 2012. However, as a result of changes in our debt structure during fiscal 2012 and entry into an interest rate swap agreement during the third quarter of fiscal 2012, which is discussed below, we believe that our exposure to interest rate changes has changed.

We are exposed to market risk from changes in interest rates on our indebtedness, which could impact our financial condition and results of operations in future periods. Our objective is to limit the impact of interest rate changes on earnings and cash flow, primarily through a mix of fixed-rate and variable-rate debt. We may also enter into interest rate swap arrangements related to certain of our variable-rate debt in order to fix the interest rate on variable rate debt if we determine that our exposure to interest rate changes is higher than optimal. Our assessment also considers our need for flexibility in our borrowing arrangements resulting from the seasonality of our business, among other factors. We continuously monitor interest rates to consider the sources and terms of our borrowing facilities in order to determine whether we have achieved our interest rate management objectives. We do not enter into debt agreements or interest rate hedging transactions on a speculative basis.

During the second quarter of fiscal 2012, we redeemed our 11<sup>3</sup>/<sub>8</sub>% Senior Secured Notes which were outstanding at that time. This redemption was funded through borrowings under our U.S. Revolving Credit Agreement and cash on hand, resulting in all of our borrowings being variable rate borrowings subsequent to this redemption. In order to mitigate our exposure to changes in interest rates in future periods, we entered into an interest rate swap agreement under which we fixed the interest rate on certain of our borrowings, ranging from \$25 million to \$45

million, during the period from August 2013 until March 2015, which essentially results in a portion of our anticipated debt levels during those periods being fixed rate borrowings at a rate equal to 0.42% plus the applicable margin, as specified in our U.S. Revolving Credit Agreement.

As of October 27, 2012, we had \$130.3 million of debt outstanding which was subject to variable interest rates. Our lines of credit, which include our U.S. Revolving Credit Agreement and our U.K. Revolving Credit Agreement, accrue interest based on variable interest rates, while providing the necessary borrowing flexibility we require due to the seasonality of our business and our need to fund certain product purchases with trade letters of credit. Based on our current borrowings under our revolving credit agreement and expected borrowings in future periods, we anticipate that interest expense of \$1.0 million will be incurred quarterly in the near term assuming no significant changes in interest rates. We estimate that a 100 basis point change in interest rates would not have a material impact on our consolidated financial statements. To the extent that the amounts outstanding under our variable-rate lines of credit change our exposure to changes in interest rates would also change to the extent we have not entered into an interest rate swap for those amounts.

#### **ITEM 4. CONTROLS AND PROCEDURES**

#### **Evaluation of Disclosure Controls and Procedures**

Our principal executive officer and our principal financial officer have evaluated the effectiveness of our disclosure controls and procedures as of the end of the period covered by this report. Based upon that evaluation, our principal executive officer and our principal financial officer concluded that, as of the end of the period covered by this report, our disclosure controls and procedures were effective in ensuring that information required to be disclosed by us in our Securities Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and then communicated to our management, including our principal executive officer and principal financial officer, as appropriate, to allow timely decisions regarding required disclosure.

#### Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting during the third quarter of fiscal 2012 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### **Additional Information**

During the third quarter of fiscal 2012, we converted the payroll and human resource systems of certain of our operations to common payroll and human resource systems. As a result of the conversions, certain controls were modified, as necessary, to supplement and complement the existing internal controls over financial reporting. In the future, we may convert other systems used in our operations to a common system across operating groups and geographic areas. The conversion of our payroll and human resource systems during the third quarter of fiscal 2012 was undertaken to provide more integrated systems across our operating groups and geographic areas, efficiencies in our operations and enhanced customer service, and not in response to any actual or perceived deficiencies in our internal control over financial reporting.

#### PART II. OTHER INFORMATION

#### **ITEM 1. LEGAL PROCEEDINGS**

In the ordinary course of business, we may become subject to litigation or claims. We are not currently a party to any litigation or regulatory action that we believe could reasonably be expected to have a material adverse effect on our financial position, results of operations or cash flows.

#### ITEM 1A. RISK FACTORS

In addition to the other information set forth in this report, investors should carefully consider the factors discussed in Part I, Item 1A. Risk Factors in our Annual Report on Form 10-K for fiscal 2011, which could materially affect our business, financial condition or operating results. The risks described in our Annual Report on Form 10-K for fiscal 2011 are not the only risks facing our company.

#### ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

- (a) During the third quarter of fiscal 2012, we did not make any unregistered sales of our equity securities.
- (c) We have certain stock incentive plans as described in Note 7 to our consolidated financial statements included in our Annual Report on Form 10-K for fiscal 2011, all of which are publicly announced plans. Under the plans, we can repurchase shares from employees to cover employee tax liabilities related to the exercise of stock options or the vesting of previously restricted shares. No shares were purchased pursuant to these plans during the third quarter of fiscal 2012.

On August 28, 2012, our Board of Directors authorized us to spend up to \$50 million to repurchase shares of our common stock. This authorization superseded and replaced all previous authorizations to repurchase shares of our common stock. As of October 27, 2012, no shares of our common stock had been repurchased pursuant to this authorization.

#### **ITEM 3. DEFAULTS UPON SENIOR SECURITIES**

None

#### ITEM 4. MINE SAFETY DISCLOSURES

None

#### **ITEM 5. OTHER INFORMATION**

None

#### **ITEM 6. EXHIBITS**

- 3.1 Restated Articles of Incorporation of Oxford Industries, Inc. Incorporated by reference to Exhibit 3.1 to the Company's Form 10-Q for the fiscal quarter ended August 29, 2003.
- Bylaws of Oxford Industries, Inc., as amended. Incorporated by reference to Exhibit 3.1 to the Company's Form 8-K filed on June 17, 2009. 3.2
- 10.1 Oxford Industries, Inc. Deferred Compensation Plan (as amended and restated effective June 13, 2012)\*
- 31.1 Section 302 Certification by Principal Executive Officer.\*
- 31.2 Section 302 Certification by Principal Financial Officer.\*
- 32 Section 906 Certification by Principal Executive Officer and Principal Financial Officer.\*
- 101.INS XBRL Instance Document\*
- 101.SCH XBRL Taxonomy Extension Schema Document\*
- 101.CAL XBRL Taxonomy Extension Calculation Linkbase Document\*
- 101.DEF XBRL Taxonomy Extension Definition Linkbase Document\*
- 101.LAB XBRL Taxonomy Extension Label Linkbase Document\*
- 101.PRE XBRL Taxonomy Extension Presentation Linkbase Document\*

<sup>\*</sup> Filed herewith.

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

December 5, 2012

#### OXFORD INDUSTRIES, INC.

(Registrant)

/s/ K. Scott Grassmyer

K. Scott Grassmyer

Senior Vice President - Finance, Chief Financial Officer and Controller
(Authorized Signatory)

35

#### OXFORD INDUSTRIES, INC.

#### DEFERRED COMPENSATION PLAN

(As amended and restated effective June 13, 2012)

# TABLE OF CONTENTS

		rage
ARTICLE I DEFINITIONS		1
Section 1.1.	Account	1
Section 1.2.	Beneficiary	1
Section 1.3.	Board	1
Section 1.4.	Code	1
Section 1.5.	Committee	1
Section 1.6.	Company	1
Section 1.7.	Company Stock	1
Section 1.8.	Compensation	1
Section 1.9.	Deferred Stock Units	2
Section 1.10.	Director	2
Section 1.11.	Discretionary Contribution	2
Section 1.12.	Election Period	2
Section 1.13.	Eligible Compensation	2
Section 1.14.	Eligible Employee	2
Section 1.15.	Eligible Participant	2
Section 1.16.	Employee	2
Section 1.17.	ERISA	2
Section 1.17.	Excess Compensation	2
Section 1.19.	401(k) Plan	3
Section 1.20.	LTIP	3
Section 1.21.	Matching Contribution	3
Section 1.22.	Maximum Deferral Percentage	3
Section 1.23.	Minimum Deferral Amount	3
Section 1.24.	Ox ford	3
Section 1.25.	Plan	3
Section 1.26.	Plan Year	3
Section 1.27.	Pre-2005 Oxford Plan	3
Section 1.28.	Restricted Share Unit	3
Section 1.29.	Restricted Share Units Award	3
Section 1.30.	Retirement Age	3
Section 1.31.	Separates from Service or Separation from Service	3
222323	(a) Leaves of Absence	4
	(b) Status Change	4
	(c) Termination of Employment	4
	(d) Separation of Director	4
	(e) Controlled Group	5
Section 1.32.	Tommy Bahama Plan	5
Section 1.33.	Years of Service	5
ARTICLE II PARTICIPATIO	N AND DEFERRAL ELECTIONS	5
Section 2.1.	Start-Up Deferral Elections	5
	(a) Eligible Employees	5
	(b) Directors	5
Section 2.2.	Annual (and Other) Deferral Elections	6
	(a) Salary	6
	(b) Bonuses	6
	·	

	(c) Restricted Share Units Awards	6
	(d) Commissions	8
	(e) Directors' Deferrals	8
Section 2.3.	Minimum Deferral Amount	9
Section 2.4.	Ongoing Election	9
Section 2.5.	Effect of Hardship Withdrawal	9
Section 2.6.	Form of Elections	9
ARTICLE III MATCHIN	G CONTRIBUTIONS	9
ARTICLE IV DISCRETION	ONARY CONTRIBUTIONS	9
ARTICLE V ACCOUNT		10
Section 5.1.	General	10
Section 5.2.	Deferrals	10
Section 5.3.	Matching and Discretionary Contributions	10
Section 5.4.	Phantom Investments	10
Section 5.5.	Phantom Investment Election	10
Section 5.6.	Phantom Investment Adjustments	10
Section 5.7.	Restricted Share Units Award Deferrals	10
	(a) Restricted Share Units	10
	(b) Deferred Stock Units	10
	(c) Company Stock Adjustments	11
	(d) Committee Discretion	11
ARTICLE VI VESTING		11
Section 6.1.	Amounts Deferred	11
Section 6.2.	Matching Contributions	11
Section 6.3.	Discretionary Contributions	11
ARTICLE VII DISTRIBU		12
Section 7.1.	Distribution Elections	12
	(a) General	12
	(b) Ongoing Election	12
a .: <b></b>	(c) Default	12
Section 7.2.	Time of Distribution	12
	(a) Separation from Service	12
	(b) Death	12
	(c) In-Service	12
	(d) Hardship Withdrawal due to Unforeseeable Emergency	12 13
Section 7.3.	(e) Delay of Payments Under Certain Circumstances Distribution Forms	13
Section 7.5.		14
	<ul><li>(a) Separation from Service After Retirement Age</li><li>(b) Separation from Service Before Retirement Age or Death</li></ul>	14
	(c) In-Service	14
	(d) Installments	14
	(e) Deferred Stock Units	14
Section 7.4.	Beneficiary	14
ARTICLE VIII NO FUNE	DING ORLIGATION	15
THE PROPERTY OF THE PROPERTY O	ALIO ODDIOLITION	13

ii

ARTICLE IX COMPLIANO	CE WITH CODE SECTION 409A	15
ARTICLE X MISCELLANI	IEOUS	15
Section 10.1.	Medium of Payment	15
Section 10.2.	Making and Revoking Elections and Designations	15
Section 10.3.	Statements	15
Section 10.4.	Claims Procedure	15
Section 10.5.	Withholding	16
Section 10.6.	No Liability	16
Section 10.7.	Nonalienation of Benefits	16
Section 10.8.	Plan Administration	16
Section 10.9.	Construction	16
Section 10.10.	No Contract of Employment	16
Section 10.11.	ERISA	16
Section 10.12.	Amendment and Termination	16
Section 10.13.	Pre-2005 Oxford Plan	17
	(a) Pre-2005 Deferrals	17
	(b) Post-2004 and Pre-2006 Deferrals	17
Section 10.14.	Tommy Bahama Plan	17
	(a) Pre-2005 Deferrals	17
	(b) Post-2004 and Pre-2006 Deferrals	17
Section 10.15.	Special Transition Bonus Election	17
EXHIBIT A		A-1
EXHIBIT B		B-1
EXHIBIT C		C-1
EXHIBIT D		D-1
	iii	

## OXFORD INDUSTRIES, INC.

#### DEFERRED COMPENSATION PLAN

(As amended and restated effective June 13, 2012)

The primary purpose of this Plan is to assist Oxford Industries, Inc. ("Oxford") and its subsidiaries in attracting and retaining employees and directors of exceptional ability by (a) allowing a select group of management or highly-compensated employees of Oxford and certain of its subsidiaries, and nonemployee members of Oxford's Board of Directors, to defer the payment of a portion of their compensation that otherwise would become payable to them, and (b) providing for discretionary contributions and matching contributions for Eligible Employees based on compensation that exceeds the compensation that may be taken into account under the Oxford Industries, Inc. Retirement Savings Plan or as a result of the dollar limitation applicable to the 401(k) Plan under Section 401(a)(17) of the Code. This Plan is an amendment and restatement of the Oxford Industries, Inc. Deferred Compensation Plan (as amended and restated effective September 1, 2010, and as subsequently amended). The terms of the Plan supersede those of the Oxford Industries, Inc. Non-Qualified Deferred Compensation Plan adopted effective January 1, 2001, and the Viewpoint International, Inc. Nonqualified Deferred Compensation Plan adopted effective July 20, 2001, except with respect to amounts deferred prior to January 1, 2006, as provided in Sections 10.13 and 10.14.

#### ARTICLE I

## **DEFINITIONS**

- Section 1.1. Account means the bookkeeping account maintained by or at the direction of the Committee to show as of any date the benefit of each Eligible Participant. Separate subaccounts may be established and maintained as part of an Eligible Participant's Account as the Committee deems necessary or appropriate to administer this Plan.
  - Section 1.2. <u>Beneficiary</u> means the person or persons designated as such in accordance with Section 7.4.
  - Section 1.3. Board means the Board of Directors of Oxford.
- Section 1.4. <u>Code</u> means the Internal Revenue Code of 1986, as amended, and rules and regulations promulgated thereunder, and any successor thereto.
  - Section 1.5. <u>Committee</u> means the committee appointed by the Board to administer the Plan.
- Section 1.6. Company means Oxford and each subsidiary of Oxford that is designated by the Board as a participating company under this Plan.
- Section 1.7. <u>Company Stock</u> means shares of Oxford common stock, \$1 par value, or any security in to which such common stock may be converted by reason of any transaction or event of the type referred to in Section 5.7(c) of this Plan (or equivalent amount of cash or other consideration as may be distributed with respect to such shares of Oxford common stock by reason of any such transaction or event).
- Section 1.8. <u>Compensation</u> means, for any Plan Year, "compensation" as defined in the 401(k) Plan for purposes of determining the amount of pre-tax contributions and matching contributions

under such plan, without regard to any limitations on compensation imposed under Section 401(a)(17) of the Code, plus any deferrals made under this Plan for such Plan Year.

- Section 1.9. <u>Deferred Stock Units</u> means credits to an Eligible Employee's Account under Section 5.7 with respect to the portion of a Restricted Share Units Award deferred under this Plan, with each such Deferred Stock Unit representing the right to receive one share of Company Stock.
  - Section 1.10. <u>Director</u> means any individual who is a member of the Board and who is not an Employee.
- Section 1.11. <u>Discretionary Contribution</u> means the amount, if any, credited to an Eligible Employee's Account in accordance with Article IV.
- Section 1.12. <u>Election Period</u> means an annual enrollment period described in Section 2.2(a), (b)(2), (c)(2), (d) and (e); a 30-day (or shorter) enrollment period described in Section 2.1(a) and (b); a Performance-based Election Period described in Section 2.2(b)(1) and (c)(1); or an enrollment period described in Section 2.2(c)(3) and (c)(4).
- Section 1.13. <u>Eligible Compensation</u> means all amounts that may be eligible for deferral under this Plan. More specifically, (i) with respect to an Eligible Employee, "Eligible Compensation" means the Eligible Employee's base salary, bonus, any Restricted Share Units Awards and, if authorized by the Committee for a Plan Year, commissions; and (ii) with respect to a Director, "Eligible Compensation" means cash compensation payable to the Director for service on the Board and committees thereof as cash retainers (excluding any such amount payable in the form of Company Stock pursuant to an election by a Director made before the deadline for making a deferral election under this Plan) and meeting fees. Additionally, with respect to an Eligible Employee, Eligible Compensation payable after the last day of the Plan Year for services performed during the final payroll period described in Section 3401(b) of the Code containing the last day of the Plan Year shall be treated for purposes of this Plan as Eligible Compensation for services performed during the final quarterly compensation period (under Oxford's compensation practices for Directors) containing the last day of the Plan Year shall be treated for purposes of this Plan as Eligible Compensation practices for Directors) containing the last day of the Plan Year shall be treated for purposes of this Plan as Eligible Compensation for services performed in the Plan Year during which such amount is paid.
- Section 1.14. <u>Eligible Employee</u> means, for each Plan Year, any Employee of a Company who was an Eligible Employee on December 31, 2009, and any other Employee of a Company whose gross annual rate of base salary is \$150,000 or more. The Committee in its discretion may adjust the foregoing salary threshold for Plan Years subsequent to Plan Year 2010; provided, however, that unless otherwise determined by the Committee, an increase in the salary threshold shall not apply to an Employee who commenced participation in the Plan prior to the Plan Year in which such increase is effective.
  - Section 1.15. Eligible Participant means each Director and Eligible Employee.
  - Section 1.16. <u>Employee</u> means an employee of Oxford or any subsidiary of Oxford.
  - Section 1.17. <u>ERISA</u> means the Employee Retirement Income Security Act of 1974, as amended.
- Section 1.18. <u>Excess Compensation</u> means the excess of an Eligible Employee's Compensation for a Plan Year over the Eligible Employee's "compensation" as defined in the 401(k) Plan for purposes of determining the amount of pre-tax contributions and matching contributions under such Plan Year.

- Section 1.19. 401(k) Plan means the Oxford Industries, Inc. Retirement Savings Plan, as amended and as in effect from time to time, or any other successor defined contribution plan maintained by Oxford or another Company that qualifies under Section 401(a) of the Code and satisfies the requirements of Section 401(k) of the Code.
- Section 1.20. <u>LTIP</u> means the Oxford Industries, Inc. Amended and Restated Long-Term Stock Incentive Plan, as amended from time to time, and any successor long-term incentive plan of Oxford. References in this Plan to the "LTIP" shall include any applicable terms and conditions contained in associated award or grant terms or agreements.
  - Section 1.21. Matching Contribution means the amount credited to an Eligible Employee's Account in accordance with Article III.
- Section 1.22. <u>Maximum Deferral Percentage</u> means, with respect to an Eligible Participant for each Plan Year, the maximum percentage of the Eligible Participant's Eligible Compensation that can be deferred under the Plan, which (i) with respect to an Eligible Employee, shall be 50% of his or her base salary, 100% of his or her bonus, 100% of his or her Restricted Share Units Award and, if authorized by the Committee for a Plan Year, 50% of his or her commissions, unless otherwise determined by the Committee prior to the beginning of such Plan Year; provided, however, that no deferral election may reduce an Eligible Employee's compensation below an amount necessary to satisfy applicable employment and income tax withholding requirements; and (ii) with respect to a Director, shall be 100% of his or her cash retainer and 100% of his or her meeting fees for each Plan Year, unless otherwise determined by the Committee prior to the beginning of such Plan Year.
- Section 1.23. <u>Minimum Deferral Amount</u> means, for each Plan Year, an amount equal to 1% of an Eligible Employee's base salary, unless otherwise determined by the Committee prior to the beginning of such Plan Year.
  - Section 1.24. Oxford means Oxford Industries, Inc. and any successor to Oxford Industries, Inc.
- Section 1.25. <u>Plan</u> means this Oxford Industries, Inc. Deferred Compensation Plan, as amended and restated effective June 13, 2012, as it may be amended in accordance with the terms hereof from time to time.
  - Section 1.26. Plan Year means the calendar year.
- Section 1.27. <u>Pre-2005 Oxford Plan</u> means the Oxford Industries, Inc. Non-Qualified Deferred Compensation Plan adopted effective January 1, 2001, as thereafter amended, as such amended plan was in effect on October 3, 2004.
  - Section 1.28. Restricted Share Unit has the meaning ascribed to such term in the LTIP.
  - Section 1.29. Restricted Share Units Award means an award of Restricted Share Units pursuant to the terms of the LTIP.
- Section 1.30. Retirement Age means, with respect to an Eligible Employee, age 55 and 5 Years of Service; each Director shall at all times be deemed to have attained Retirement Age for purposes of this Plan.
- Section 1.31. <u>Separates from Service or Separation from Service</u> means the termination of employment with Oxford and its subsidiaries or cessation of Board service, as the case may be, in such a

manner as to constitute a "separation from service" (other than death) within the meaning of Section 409A of the Code.

As a general overview of Code Section 409A's definition of "separation from service", an Eligible Participant separates from service if an Eligible Employee has a termination of employment or a Director ceases to perform services (in either case, other than for death) with Oxford and all members of the controlled group, determined in accordance with the following:

- (a) Leaves of Absence. For Eligible Employees, the employment relationship is treated as continuing intact while the Eligible Employee is on military leave, sick leave, or other bona fide leave of absence if the period of such leave does not exceed 6 months, or, if longer, so long as the Eligible Employee retains a right to reemployment with the Company or a member of the controlled group under an applicable statute or by contract. A leave of absence constitutes a bona fide leave of absence only while there is a reasonable expectation that the Eligible Employee will return to perform services for Oxford or a member of the controlled group. If the period of leave exceeds 6 months and the Eligible Employee does not retain a right to reemployment under an applicable statute or by contract, the employment relationship is deemed to terminate on the first date immediately following such 6-month period.
- (b) <u>Status Change.</u> Generally, if an Eligible Employee performs services both as an Employee and an independent contractor, the Eligible Employee must separate from service both as an Employee and as an independent contractor pursuant to standards set forth in Treasury Regulations to be treated as having a separation from service. However, if an Eligible Employee provides services as an Employee and as a member of the Board, the services provided as a Director are not taken into account in determining whether the Eligible Employee has a separation from service as an Employee for purposes of this Plan.
- (c) Termination of Employment. Whether a termination of employment has occurred is determined based on whether the facts and circumstances indicate that the Company, all members of the controlled group and the Eligible Employee reasonably anticipate that (1) no further services will be performed after a certain date, or (2) the level of bona fide services the Eligible Employee will perform after such date (whether as an Employee or as an independent contractor) will permanently decrease to no more than 20% of the average level of bona fide services performed (whether as an Employee or an independent contractor) over the immediately preceding 36-month period (or the full period of services to Oxford and all members of the controlled group if the Eligible Employee has been providing services to Oxford and all members of the controlled group for less than 36 months). Facts and circumstances to be considered in making this determination include, but are not limited to, whether the Eligible Employee continues to be treated as an Employee for other purposes (such as continuation of salary and participation in employee benefit programs), whether similarly-situated service providers have been treated consistently, and whether the Eligible Employee is permitted, and realistically available, to perform services for other service recipients in the same line of business. For periods during which an Eligible Employee is on a paid bona fide leave of absence and has not otherwise terminated employment as described in subsection (a) above, for purposes of this subsection, the Eligible Employee is treated as providing bona fide services at a level equal to the level of services that the Eligible Employee would have been required to perform to receive the compensation paid with respect to such leave of absence. Periods during which an Eligible Employee is on an unpaid bona fide leave of absence and has not otherwise terminated employment are disregarded for purposes of this subsection (including for purposes of determining the a
- (d) <u>Separation of Director</u>. For Directors, whether a separation from service has occurred is determined based on whether the facts and circumstances indicate that Oxford and the Director reasonably anticipate that the Director has ceased to perform services under circumstances constituting a good faith and complete termination of the relationship (and only if Oxford does not

anticipate a renewal of the relationship or the Director becoming an Employee). If a Director also provides additional services as an independent contractor, the Director will not be considered to have a separation from service for purposes of Section 409A until he or she has separated from service both as a Director and as an independent contractor.

- (e) <u>Controlled Group.</u> For purposes of this Section 1.31, "controlled group" means any other entity that would be required to be aggregated with Oxford under Code Sections 414(b) or (c), determined, however, by using "at least 50 percent" instead of "at least 80 percent" in applying Code Section 414(b) or (c).
- Section 1.32. <u>Tommy Bahama Plan</u> means the Viewpoint International, Inc. Nonqualified Deferred Compensation Plan adopted effective July 20, 2001 as thereafter amended, as such amended plan was in effect on October 3, 2004.
  - Section 1.33. Years of Service means "years of service" as defined in the 401(k) Plan.

#### ARTICLE II

## PARTICIPATION AND DEFERRAL ELECTIONS

- Section 2.1. <u>Start-Up Deferral Elections</u>. If permitted by the Committee, each person who first qualifies as an Eligible Participant after the beginning of a Plan Year but before the annual enrollment period for the next following Plan Year, or after the beginning of a performance period, and who is treated as first becoming eligible to participate in the Plan or any "account balance" plan aggregated with the Plan under the plan aggregation rules of Section 409A of the Code, shall be eligible to elect to participate in this Plan during the 30-day period (or a shorter period specified by the Committee) starting on the date he or she first qualifies as an Eligible Participant. All such elections shall be made in conformance with the rules under Section 409A of the Code for determining whether such an Eligible Participant may be treated as first becoming eligible.
- (a) Eligible Employees. Each such Eligible Employee may elect prior to the end of such 30-day (or shorter) period to defer from his or her Eligible Compensation up to the Maximum Deferral Percentage (in whole percentage increments) of his or her base salary and bonus, his or her Restricted Share Units Award and (if authorized by the Committee) commissions, for services performed after the date the Eligible Employee's deferral election is made and becomes effective. Any such election shall be irrevocable at the end of such 30-day (or shorter) period and through the end of the Plan Year or performance period for which it is made (except as provided in Section 2.5). Any such deferral of salary, or of commissions if applicable, shall be effective beginning with the paycheck for the first regular payroll period beginning after the deferral election is made and becomes effective. The amount of any bonus or Restricted Share Units Award deferred with respect to an election made after the beginning of a performance period will be prorated in accordance with Section 409A of the Code.
- (b) <u>Directors.</u> Each such Director may elect prior to the end of such 30-day (or shorter) period to defer from his or her Eligible Compensation up to the Maximum Deferral Percentage (in whole percentage increments) of his or her cash retainer and meeting fees for services performed thereafter. Such deferral elections shall first relate to (i) a Director's cash retainer for the 3-month period for which quarterly payments of Director compensation are made that begins after the date the Director's deferral election is made and becomes effective, and (ii) a Director's meeting fees for all meetings of the Board (or a committee thereof) that are held after the date the Director's deferral election is made and becomes effective. In any case, any such election shall be irrevocable at the end of such 30-day (or shorter) period and through the end of the Plan Year for which it is made (except as provided in Section 2.5).

#### Section 2.2. <u>Annual (and Other) Deferral Elections.</u>

(a) <u>Salary</u>. An Eligible Employee shall have the right during the enrollment period established by the Committee to defer up to the Maximum Deferral Percentage (in whole percentage increments) of his or her base salary for services performed in the following Plan Year. Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of the Plan Year for which it is made (except as provided in Section 2.5).

#### (b) <u>Bonuses</u>.

- (1) Performance-Based Compensation Bonus. An Eligible Employee may elect during an annual enrollment period established by the Committee or any other election period described in clause (iii) below to defer up to the Maximum Deferral Percentage (in whole percentage increments) of a "performance-based compensation" bonus earned for services performed during a performance period; provided that (i) such bonus constitutes "performance-based compensation" within the meaning of Section 409A of the Code, (ii) the performance period is at least 12 months, (iii) the election period ends at least 6 months before the end of the performance period (the "Performance-based Election Period"), (iv) the Eligible Employee has been an Employee continuously from the date upon which the performance criteria were established through the date of such election, and (v) at the time of the election, the performance-based compensation is not readily ascertainable. Any such election that is not revoked by the end of the applicable enrollment period described above shall be irrevocable immediately following such enrollment period and shall remain irrevocable through the end of the performance period (except as provided in Section 2.5).
- (2) Other Bonuses. If a bonus is not intended to satisfy the requirements for "performance-based compensation" within the meaning of Section 409A of the Code, then an Eligible Employee may elect during an annual or other enrollment period established by the Committee to defer up to the Maximum Deferral Percentage (in whole percentage increments) of such bonus that otherwise would be payable to such Eligible Employee for services performed during the performance period that begins in the following Plan Year. Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of such performance period (except as provided in Section 2.5).

## (c) Restricted Share Units Awards.

(1) Performance-Based Restricted Share Units Awards. Subject to Section 2.2(c)(5), an Eligible Employee may elect during an annual enrollment period established by the Committee or any other election period described in clause (iii) below to defer up to the Maximum Deferral Percentage (in whole percentage increments) of a Restricted Share Units Award that is "performance-based compensation" awarded for services performed during a performance period; provided that (i) such Restricted Share Units Award constitutes "performance-based compensation"

within the meaning of Section 409A of the Code, (ii) the performance period is at least 12 months, (iii) the election period ends at least 6 months before the end of the performance period (the "Performance-based Election Period"), (iv) the Eligible Employee has been an Employee continuously from the date upon which the performance criteria were established through the date of such election, and (v) at the time of the election, the performance-based compensation is not readily ascertainable. Any such election that is not revoked by the end of the applicable enrollment period described above shall be irrevocable immediately following such enrollment period and shall remain irrevocable through the end of the performance period (except as provided in Section 2.5).

- (2) Other Restricted Share Units Awards. Subject to Section 2.2(c)(5), if a Restricted Share Units Award is not intended to satisfy the requirements for "performance-based compensation" within the meaning of Section 409A of the Code, then an Eligible Employee may elect during an annual or other enrollment period established by the Committee to defer up to the Maximum Deferral Percentage (in whole percentage increments) of such Restricted Share Units Award awarded to such Eligible Employee for services performed during the performance period that begins in the following Plan Year. Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of such performance period (except as provided in Section 2.5).
- (3) Deferral Elections for Compensation Subject to a Risk of Forfeiture. Subject to Section 2.2(c)(5), an Eligible Employee may elect to defer up to the Maximum Deferral Percentage (in whole percentage increments) of a Restricted Share Units Award that is subject to a condition requiring the Eligible Employee to continue to provide services for a period of at least 12 months from the date of grant of the award to avoid forfeiture of payment of the award, as provided in this subsection. Any such election must be made on or before the 30th day after the date the award is granted, provided that the election is made at least 12 months before the earliest date on which the forfeiture condition could lapse. For purposes of this subsection, a condition will not be treated as failing to require the Eligible Employee to continue to provide services for a period of at least 12 months from the date of grant of the award merely because the condition immediately lapses upon the death or disability (as defined in Section 409A of the Code) of the Eligible Employee, or upon a change in control event (as defined in Section 409A of the Code). However, if the condition in fact lapses before the end of such 12-month period due to such event(s), a deferral election shall be given effect only if the deferral election satisfies Code Section 409A without regard to this special timing rule.
- (4) <u>Deferral Elections for Short-Term Deferrals.</u> Subject to Section 2.2(c)(5), an Eligible Employee who has a legally binding right to a Restricted Share Units Award that, absent a deferral election, would be treated as a short-term deferral within the meaning of Treasury Regulation § 1.409A-1(b)(4), may elect to defer up to the Maximum

Deferral Percentage (in whole percentage increments) of the Restricted Share Units Award, as provided in this subsection. Any such election shall be effective only if (i) the deferral election will not take effect until at least 12 months after the date it is made, (ii) the deferral election is made at least 12 months before the substantial risk of forfeiture lapses with respect to such Restricted Share Units Award, and (iii) payment will be deferred for at least 5 years from the date the substantial risk of forfeiture lapses with respect to such Restricted Share Units Award, and all other applicable requirements under Section 409A of the Code are satisfied. Such deferral election may, to the extent permitted by the Committee, provide for earlier payment with respect to any such Restricted Share Units Award upon the Eligible Employee's death, disability, unforeseeable emergency or change in control event, in each case only to the extent permitted under Section 409A of the Code.

- (5) <u>Committee Discretion</u>. Notwithstanding the foregoing terms of this Section 2.2(c), any election described in this Section shall be made available to Eligible Employees only if and to the extent permitted by the Committee from time to time in its discretion.
- (d) <u>Commissions</u>. If the Committee in its discretion determines to allow deferrals to be made with respect to commissions for any Plan Year, an Eligible Employee may elect during the annual enrollment period established by the Committee preceding such Plan Year to defer up to the Maximum Deferral Percentage (in whole percentage increments) of his or her commissions that are treated under Section 409A of the Code as attributable to services performed by him or her during such Plan Year. Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of such performance period (except as provided in Section 2.5).

## (e) <u>Directors' Deferrals</u>.

- (1) Cash Retainers. A Director shall have the right during the enrollment period established by the Committee to defer up to the Maximum Deferral Percentage (in whole percentage increments) of his or her cash retainers for services in the following Plan Year. Such deferral elections shall relate to a Director's cash retainers that would, absent such deferral, have been paid during such Plan Year (e.g., quarterly payments of cash retainers that are regularly paid at the end of March, June, September and December of each Plan Year under Oxford's compensation practices for Directors, or otherwise as in effect from time to time). Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of the Plan Year for which it is made (except as provided in Section 2.5).
- (2) Meeting Fees. A Director shall have the right during the enrollment period established by the Committee to defer up to the Maximum Deferral Percentage (in whole percentage increments) of his or her meeting fees for services in the following Plan Year. Such deferral elections shall relate to a Director's meeting fees that would, absent such deferral, have been paid during such Plan Year (e.g., payments of meeting fees that are regularly made in connection with each Board meeting in accordance with Oxford's compensation practices for

Directors, or otherwise as in effect from time to time). Any such election that is not revoked by December 31 following (or coinciding with) the end of the enrollment period shall become irrevocable (at the end of the day) on such December 31 and shall remain irrevocable through the end of the Plan Year for which it is made (except as provided in Section 2.5).

- Section 2.3. <u>Minimum Deferral Amount</u>. An Eligible Employee's base salary deferral elections must provide for a deferral at least equal to the Minimum Deferral Amount for the Eligible Employee (prorated for a start-up election pursuant to Section 2.1 or upon Separation from Service during a Plan Year). The Committee may, in its discretion, also impose minimum deferral amounts with respect to deferrals of Eligible Compensation other than base salary.
- Section 2.4. Ongoing Election. A deferral election made in accordance with Sections 2.1 or 2.2 (other than a deferral election with respect to a Restricted Share Units Award) shall remain in effect for a subsequent Plan Year (or subsequent performance period in the case of a "performance-based compensation" deferral with respect to a bonus under Section 2.2(b)(1)) unless revised or revoked during the enrollment period for such Plan Year or performance period, unless the Committee requires a new election.
- Section 2.5. Effect of Hardship Withdrawal. An Eligible Participant who has taken a hardship withdrawal pursuant to Section 7.2(d), or has taken a hardship withdrawal pursuant to the 401(k) Plan, shall have his or her deferral election under this Plan automatically cancelled effective immediately upon such withdrawal and for the remainder of the Plan Year and performance period, or for the remainder of the Plan Year and any subsequent Plan Year and performance period in which deferrals under the 401(k) Plan are suspended. Such Eligible Participant may recommence participation in the Plan only during an annual enrollment period or a Performance-based Election Period and his or her election shall not become effective until the beginning of the following Plan Year or, with respect to the deferral of "performance-based compensation," the applicable performance period.
- Section 2.6. <u>Form of Elections</u>. Any deferral election shall be made in the form and manner provided by the Committee for this purpose and in accordance with such other rules and procedures as may be established from time to time by the Committee.

#### ARTICLE III

#### MATCHING CONTRIBUTIONS

Unless otherwise determined by the Committee, Oxford shall credit the Account of each Eligible Employee who elects to defer the Minimum Deferral Amount for a Plan Year with a Matching Contribution equal to 4% of his or her Excess Compensation for such Plan Year.

#### ARTICLE IV

## **DISCRETIONARY CONTRIBUTIONS**

The Committee may credit each Eligible Employee's Account with a Discretionary Contribution, if any, at such times and in such amounts as recommended by the Committee and approved by the Nominating, Compensation & Governance Committee of the Board, or the Board, in its sole discretion. (Discretionary Contributions do not apply with respect to Directors).

## ARTICLE V

#### ACCOUNT ADJUSTMENTS

- Section 5.1. <u>General</u>. An Eligible Participant's benefit under this Plan shall be based entirely on the amounts credited to his or her Account at any time, which will depend upon the amount deferred under Article II, any Matching Contributions credited under Article III, any Discretionary Contributions credited under Article IV, any dividend equivalent amounts credited with respect to Deferred Stock Units under Section 5.7(b), and the phantom investment and other adjustments made in accordance with this Article V.
- Section 5.2. <u>Deferrals.</u> Amounts deferred by an Eligible Participant shall be credited to his or to her Account as soon as practicable after the date that such Eligible Compensation otherwise would have been payable to the Eligible Participant if no election had been made under Article II, except as otherwise provided in Section 5.7 with respect to deferred Restricted Share Units Awards.
- Section 5.3. <u>Matching and Discretionary Contributions</u>. The Matching Contribution and Discretionary Contribution, if any, shall be credited to an Eligible Employee's Account as of the end of the calendar year, or at such time as otherwise may be determined by the Committee in its absolute discretion.
- Section 5.4. <u>Phantom Investments</u>. The Committee from time to time shall select one or more investment funds that will serve as hypothetical investment options for the deferrals (other than Restricted Share Units Award deferrals, as described in Section 5.7), Matching Contributions, Discretionary Contributions and dividend equivalents with respect to Deferred Stock Units credited to an Account ("phantom investment funds"). The Committee may establish limits on the portion of an Account that may be invested hypothetically in any phantom investment fund or in any combination of phantom investment funds.
- Section 5.5. <u>Phantom Investment Election.</u> Each Eligible Participant shall elect pursuant to procedures established by the Committee to treat amounts credited to his or her Account as if they were invested in one or more phantom investment funds (a "phantom investment election"). An Eligible Participant may change his or her phantom investment elections in accordance with the Committee's procedures. Any phantom investment election shall be effective only if made in accordance with the Committee's procedures.
- Section 5.6. <u>Phantom Investment Adjustments</u>. The Committee shall cause the Eligible Participant's Account to be adjusted from time to time for any earnings and losses as if it were invested in accordance with the Eligible Participant's phantom investment elections (except as otherwise provided in Section 5.7 with respect to Deferred Stock Units). Such adjustments shall be made until his or her Account is distributed in full under Article VII.

## Section 5.7. <u>Restricted Share Units Award Deferrals.</u>

- (a) Restricted Share Units. Restricted Share Units subject to deferral elections described in Section 2.2(c) shall be credited to the Eligible Employee's Account, as Deferred Stock Units, at the time such Restricted Share Units would have become vested and payable to such Eligible Employee under the terms of the LTIP absent his or her deferral election under this Plan, into a Deferred Stock Units subaccount. Prior to such crediting of Deferred Stock Units to the Account, Restricted Share Units subject to such deferral elections shall remain subject to all terms and conditions of the LTIP. Without limiting the generality of the foregoing, such Restricted Share Units shall remain subject to terms of the LTIP regarding vesting (including any applicable time-based and/or performance-based vesting conditions), voting rights, entitlement to dividend equivalents and adjustment of shares of Company Stock.
- (b) <u>Deferred Stock Units.</u> After Restricted Share Units have been credited to an Eligible Employee's Account as Deferred Stock Units, the Eligible Employee's Account shall be credited

with amounts equivalent to any dividends Oxford may pay on an equivalent number of shares of Company Stock. Such amounts shall be credited to the Account, as a cash amount, as of the date on which dividends are paid to shareholders by Oxford on Company Stock.

- (c) Company Stock Adjustments. The Committee shall make or provide for such adjustments in an Eligible Employee's Deferred Stock Units as the Committee shall in good faith determine to be equitably required in order to prevent dilution or enlargement of the rights of Eligible Employees that otherwise would result from (i) any stock dividend, stock split, recapitalization or other change in the capital structure of Oxford; (ii) any merger, consolidation, spin-off, spin-out, split-off, split-up, reorganization or partial or complete liquidation or other distribution of assets (other than a normal cash dividend); or (iii) any other event that would constitute an equity restructuring under the Code. Without limiting the foregoing, the Committee may make or provide for such adjustments in the (x) number of outstanding Deferred Stock Units hereunder, and (y) kind of shares of Company Stock covered by Deferred Stock Units hereunder, as the Committee may in good faith determine to be equitably required in order to prevent dilution or enlargement of the rights of Eligible Employees that otherwise would result from (A) any combination or exchange of shares of Company Stock, (B) any issuance of rights or warrants to purchase securities or (C) any other corporate transaction or event having an effect similar to any of the foregoing. Moreover, in the event of any such transaction or event, the Committee may provide in substitution for any or all outstanding Deferred Stock Units under this Plan such alternative consideration as it may in good faith determine to be equitable under the circumstances and may require in connection therewith the surrender of all Deferred Stock Units so replaced. The amount of such adjustment of Deferred Stock Units in such event will be equal to the adjustment that would have been made if each Deferred Stock Unit credited to an Eligible Employee's Account immediately prior to the event constituted one share of Company Stock.
- (d) <u>Committee Discretion</u>. Notwithstanding the foregoing provisions of this Section 5.7, with respect to any elections by Eligible Employees to defer Restricted Share Units Awards, the Committee may prescribe alternative terms to those described in this Section by specifying any such alternative terms in the election form(s) governing such deferrals.

#### ARTICLE VI

## **VESTING**

- Section 6.1. <u>Amounts Deferred.</u> An Eligible Participant shall be 100% vested at all times in the Eligible Participant's deferrals and the earnings thereon; except that, with respect to deferrals of Restricted Share Units Awards, all vesting conditions in the LTIP shall continue to apply as described in Section 5.7(a).
- Section 6.2. <u>Matching Contributions</u>. An Eligible Employee's Matching Contributions, and earnings thereon, shall be 100% vested at all times, unless otherwise determined by the Committee prior to crediting to the Eligible Employee's Account.
- Section 6.3. <u>Discretionary Contributions</u>. An Eligible Employee's Discretionary Contributions, and earnings thereon, shall become vested as determined by the Committee and as approved by the Nominating, Compensation & Governance Committee of the Board, or the Board.

## ARTICLE VII

#### **DISTRIBUTIONS**

## Section 7.1. Distribution Elections.

- (a) <u>General</u>. At the same time as an Eligible Participant makes a deferral election under Article II, he or she shall elect, pursuant to Section 7.2, the time as of which amounts credited to his or her Account for such Plan Year (adjusted as provided under Article V) will be distributed and, pursuant to Section 7.3, the form in which such distribution will be made.
- (b) Ongoing Election. In the absence of any contrary rule established by the Committee before the applicable Election Period, a Separation from Service distribution election (other than such an election with respect to a Restricted Share Unit Award) shall remain in effect for contributions credited to an Account for a subsequent Plan Year (or subsequent performance period in the case of a "performance-based compensation" deferral with respect to a bonus under Section 2.2(b)(1)), unless revised or revoked during the Election Period for such Plan Year or the Performance-based Election Period. An in-service distribution election will apply only to the Plan Year or performance period with respect to which the election was made and will not apply to a subsequent Plan Year or performance period.
- (c) <u>Default</u>. If an Eligible Participant fails to make an election as to the time or form of distribution of his or her Account (or subaccount, as applicable), his or her distribution will be made in a lump sum in the first calendar month that begins 6 months after the date of his or her Separation from Service.
- Section 7.2. <u>Time of Distribution</u>. Distribution of an Eligible Participant's Account (or subaccount, as applicable) may be made as a result of the Eligible Participant's Separation from Service, death, the occurrence of a hardship due to an unforeseeable emergency, or at a specified time while the Eligible Participant is still an Employee or a Director.
- (a) <u>Separation from Service</u>. If distribution is made as a result of the Eligible Participant's Separation from Service, it will be made or commence in the first calendar month that begins (1) 6 months or 12 months (as selected by the Eligible Participant) after the date the Eligible Participant Separates from Service, if the Separation from Service is after Retirement Age, or (2) 6 months after the date the Eligible Employee Separates from Service, if the Separation from Service is before Retirement Age. If distribution is to be made in annual installments, any subsequent annual installments shall be made in February of the applicable year.
- (b) <u>Death.</u> If an Eligible Participant dies before distributions commence, distribution will be made in the first month of the calendar quarter immediately following the quarter in which his or her death occurred. If an Eligible Participant dies after distributions have commenced pursuant to his or her Separation from Service or paragraph (c) below, the balance, if any, of his or her Account will be distributed in the first month of the calendar quarter immediately following the quarter in which his or her death occurred.
- (c) <u>In-Service</u>. An Eligible Participant may elect that his or her subaccount for a Plan Year or performance period be distributed or commence to be distributed in February of any calendar year that begins at least 2 years after the end of the Plan Year for which the deferrals were credited to such subaccount; provided he or she is an Employee or a Director on the date of the distribution. An Eligible Participant may revise such in-service distribution election to change the time of distribution; provided, however, that (1) the revision will not take effect until 12 months after the date it is made, (2) the revision must be made at least 12 months before the in-service distribution otherwise would commence, and (3) the in-service distribution will be deferred for at least 5 years from the date the in-service distribution would have commenced in the absence of the revision.
- (d) <u>Hardship Withdrawal due to Unforeseeable Emergency</u>. An Eligible Participant shall have the right to request that the Committee distribute all, or a part of, his or her Account to him or to her in a lump sum if he or she experiences severe financial hardship resulting from an illness or

accident of the Eligible Participant, the spouse of the Eligible Participant or a dependent (as defined in Section 152 of the Code, without regard to subsections (b)(1), (b)(2) and (d)(1)(B) thereof) of the Eligible Participant, loss of the Eligible Participant's property due to casualty, or other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Eligible Participant (an "unforeseeable emergency"). The Committee shall have the sole discretion to determine whether to grant an Eligible Participant's withdrawal request under this Section 7.1(d) and the amount to distribute to the Eligible Participant; provided, however, that no distribution shall be made to an Eligible Participant under this Section 7.1(d) to the extent that such hardship is or may be relieved (1) through reimbursement or compensation by insurance or otherwise, (2) by liquidation of the Eligible Participant's assets, to the extent the liquidation of the Eligible Participant's assets would not itself cause severe financial hardship, or (3) by a permissible cessation of deferral elections under this Plan. The amount of any distributions from an Eligible Participant's Account pursuant to this Section 7.1(d) shall be limited to the amount necessary to meet the unforeseeable emergency, plus amounts necessary to pay taxes reasonably anticipated as a result of the distribution. An Eligible Participant who takes a hardship withdrawal under this Section 7.1(d) will be ineligible to make deferrals under the Plan for the remainder of the Plan Year and performance period. Distribution shall be made in the calendar month following the determination by the Committee that a hardship withdrawal will be permitted.

- (e) <u>Delay of Payments Under Certain Circumstances</u>. Notwithstanding the provisions of paragraph (a) through (d) above, to the extent permitted by Section 409A of the Code, Oxford, in its discretion, may delay payment to a date after the payment date designated in such paragraphs under any of the following circumstances:
  - (1) Payments that Would Jeopardize Oxford and its Subsidiaries as a Going Concern. Payment will be delayed where the Committee determines that the making of the payment at the time specified under the Plan would jeopardize the ability of Oxford and its subsidiaries to continue as a going concern; provided that such delayed payment will be made during the first taxable year of Oxford in which the making of the payment will not have such effect.
  - (2) Payments that Would Violate Federal Securities Laws or Other Applicable Law. Payment will be delayed where the Committee reasonably anticipates that the making of the payment will violate federal securities laws or other applicable law; provided that the delayed payment is made at the earliest date at which the Committee reasonably anticipates that the making of the payment will not cause such violation.
  - (3) Payments Subject to Section 162(m). Payment to an Eligible Participant may be delayed to the extent that Oxford reasonably anticipates that if the payment were made as scheduled, Oxford's deduction with respect to such payment would not be permitted due to the application of Section 162(m) of the Code; provided that the payment is made either during the Eligible Participant's first taxable year in which Oxford reasonably anticipates, or should reasonably anticipate, that if the payment is made during such year, the deduction of such payment will not be barred by application of Section 162(m) of the Code or during the period beginning with the date of the Eligible Participant's Separation from Service and ending on the later of the last day of the taxable year of Oxford in which the Eligible Participant Separates from Service or the 15th day of the third month following the Eligible Participant's Separation from Service; provided further that where any scheduled payment to a specific Eligible Participant in a taxable year of Oxford is

delayed in accordance with this Section 7.2(e)(3), all scheduled payments to such Eligible Participant that could be delayed in accordance with this Section 7.2(e)(3) also will be delayed.

## Section 7.3. <u>Distribution Forms</u>.

- (a) <u>Separation from Service After Retirement Age</u>. An Eligible Participant may elect that if he or she Separates from Service after Retirement Age, his or her subaccount for a Plan Year shall be distributed in a lump sum or annual installments over 2 to 15 years. Notwithstanding anything in this paragraph (a) to the contrary, if the Eligible Participant's Account balance following Separation from Service is less than \$25,000, then the Account will be distributed in a lump sum, rather than installments.
- (b) <u>Separation from Service Before Retirement Age or Death.</u> If an Eligible Employee Separates from Service before Retirement Age or if any Eligible Participant dies before his or her entire Account is distributed, his or her entire Account will be distributed in a lump sum, regardless of whether in-service distributions have commenced pursuant to Section 7.2(c).
- (c) In-Service. An Eligible Participant may elect that an in-service distribution be paid in a lump sum or in annual installments over 2 to 5 years. An Eligible Participant may revise such in-service distribution election to change the form of distribution; provided, however, that (1) the revision will not take effect until 12 months after the date it is made, (2) the revision must be made at least 12 months before the in-service distribution otherwise would commence, and (3) the in-service distribution will be deferred for at least 5 years from the date the in-service distribution would have commenced in the absence of the revision. If the Eligible Participant Separates from Service before Retirement Age or dies, his or her Account will be distributed in accordance with Section 7.3(b) and not this Section 7.3(c), even if distributions had commenced under this Section 7.3(c). However, if the Eligible Participant Separates from Service after Retirement Age, then distribution of any subaccount that had commenced under this Section 7.3(c) shall continue to be paid as scheduled, but payment of any subaccounts that had not commenced under this Section 7.3(c) shall be made in accordance with the form elected in Section 7.3(a).
- (d) Installments. The amount of any installment distributable under this Section 7.3 shall be computed by multiplying the portion of the Eligible Participant's Account (or subaccount, as applicable) to be distributed in installments by a fraction, the numerator of which shall be one and the denominator of which shall be the number of installments remaining after such installment has been paid plus one. In the case of installment payments of amounts distributable in the form of shares of Company Stock, such fraction shall be applied to the number of Deferred Stock Units in the Eligible Participant's Account immediately before such payment is made and, if a fractional share would result, it shall instead be rounded up to the next whole share.
- (e) <u>Deferred Stock Units.</u> Notwithstanding anything to the contrary in this Plan, Deferred Stock Units will be distributed in the form of shares of Company Stock.
- Section 7.4. <u>Beneficiary</u>. An Eligible Participant shall designate (on a form provided for this purpose) a person, or more than one person, as his or her Beneficiary to receive the balance credited to his or her Account in the event of his or her death. An Eligible Participant may change his or her Beneficiary designation at any time. If no Beneficiary designation is in effect on the date an Eligible Participant dies or if no designated Beneficiary survives the Eligible Participant, the Eligible Participant's estate automatically shall be treated as his or her Beneficiary under this Plan.

#### ARTICLE VIII

#### NO FUNDING OBLIGATION

The obligation of the Company to make any distributions under this Plan shall be unfunded and unsecured; all distributions to, or on behalf of, an Eligible Participant under this Plan shall be made from the general assets of the Company, and any claim by an Eligible Participant or Beneficiary against the Company for any distribution under this Plan shall be treated, with respect to a Director, the same as a claim of any general and unsecured creditor of Oxford, or with respect to an Eligible Employee, the same as a claim of any general and unsecured creditor of Oxford or any other Company by whom the Eligible Employee was employed. Notwithstanding the foregoing, Oxford may, in its discretion, establish one or more irrevocable grantor trusts for the purpose of funding all or part of the obligations under this Plan; provided, however, that the terms of any such trusts require that the assets thereof remain subject to the claims of Oxford's and the other Company's judgment creditors and are non-assignable and non-alienable by any Eligible Participant or Beneficiary prior to distribution thereof.

#### ARTICLE IX

## **COMPLIANCE WITH CODE SECTION 409A**

Oxford intends that this Plan meet the requirements of Section 409A(a)(2), (3) and (4) of the Code (and any successor provisions of the Code) and the regulations and other guidance issued thereunder (the "Requirements") and be operated in accordance with such Requirements so that compensation deferred under this Plan (and applicable investment earnings) shall not be included in income under Section 409A of the Code. Any ambiguities in this Plan shall be construed to effect the intent as described in this Article IX. If any provision of this Plan is found to be in violation of the Requirements, then such provision shall be deemed to be modified or restricted to the extent and in the manner necessary to render such provision in conformity with the Requirements, or shall be deemed excised from this Plan, and this Plan shall be construed and enforced to the maximum extent permitted by the Requirements as if such provision had been originally incorporated in this Plan as so modified or restricted, or as if such provision had not originally been incorporated in this Plan, as the case may be.

#### ARTICLE X

## **MISCELLANEOUS**

- Section 10.1. <u>Medium of Payment</u>. Distributions under this Plan shall be made in cash; except that, distributions of Deferred Stock Units shall instead be distributed in the form of shares of Company Stock.
- Section 10.2. <u>Making and Revoking Elections and Designations</u>. Any election or designation or revised election or designation under this Plan shall be effective only when the properly completed election or designation form is received by the Committee or its delegate before the Eligible Participant's death, subject to the rules set forth in this Plan.
  - Section 10.3. Statements. Oxford or its agent shall provide periodic statements to the Eligible Participant to show his or her Account balance.
- Section 10.4. <u>Claims Procedure</u>. Any claim for a benefit under this Plan shall be filed and resolved in accordance with the claims procedure provided under the 401(k) Plan, which procedure hereby is incorporated in this Plan by reference, except that (a) the Committee of this Plan shall be the entity with whom a claim for review should be filed under this Plan and (b) the Committee has absolute discretion to resolve any claims under this Plan.

- Section 10.5. <u>Withholding</u>. The Company may take whatever action that the Company deems appropriate to satisfy applicable federal, state and local income tax withholding requirements that the Company determines applicable under this Plan.
- Section 10.6. <u>No Liability</u>. No Eligible Participant and no Beneficiary of an Eligible Participant shall have the right to look to, or have any claim whatsoever against, any officer, director, employee or agent of the Company in his or her individual capacity for the distribution of any Account.
- Section 10.7. <u>Nonalienation of Benefits</u>. No benefit or payment under this Plan shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, levy or charge, and any attempt so to anticipate, alienate, sell, transfer, assign, pledge, encumber, levy upon or charge the same shall be void.
- Section 10.8. Plan Administration. The Committee shall be the administrator of this Plan, and the Committee has the exclusive responsibility and complete discretionary authority to control the operation, management and administration of this Plan, with all powers necessary to enable it properly to carry out those responsibilities, including (but not limited to) the power to construe this Plan, to determine eligibility for benefits, to settle disputed claims and to resolve all administrative, interpretive, operational, equitable and other questions that arise under this Plan. The decisions of the Committee on all matters within the scope of its authority shall be final and binding. To the extent a discretionary power or responsibility under this Plan is expressly assigned to a person by the Committee, that person will have complete discretionary authority to carry out that power or responsibility and that person's decisions on all matters within the scope of that person's authority will be final and binding. Notwithstanding the foregoing, no Eligible Participant shall participate in any determination that relates solely or primarily to his or her own Account.
- Section 10.9. <u>Construction</u>. This Plan shall be construed in accordance with the laws of the State of Georgia. Headings and subheadings have been added only for convenience of reference and shall have no substantive effect whatsoever. All references to the singular shall include the plural and all references to the plural shall include the singular.
- Section 10.10. No Contract of Employment. Nothing contained in this Plan shall be construed as a contract of employment between the Company and an Eligible Participant, as a right of any Eligible Employee to be continued in the employment of the Company, as a right of any Director to remain on the Board, or as a limitation of the otherwise applicable rights of the Company to discharge an Eligible Employee with or without cause, or to cause a Director's service on the Board to end.
- Section 10.11. <u>ERISA</u>. Oxford intends that this Plan come within the various exceptions and exemptions to ERISA for a plan maintained for a "select group of management or highly compensated employees" as described in Sections 201(2), 301(a)(3), and 401(a)(1) of ERISA. Any ambiguities in this Plan shall be construed to affect the intent as described in this Section 10.11.
- Section 10.12. Amendment and Termination. The Nominating, Compensation & Governance Committee of the Board, or the Board, shall have the right to amend this Plan from time to time and to terminate this Plan at any time; provided, however, that (a) the balance credited to each Account immediately after any such amendment or termination shall be no less than the balance credited to such Account immediately before such amendment or termination (as adjusted for phantom investment fund performance), (b) the Nominating, Compensation & Governance Committee, or the Board, may accelerate the distribution of Account balances under this Plan upon termination to the extent permissible under Section 409A of the Code, and (c) except to conform to the requirements of Section 409A of the Code, no amendment or termination shall adversely affect an Eligible Participant's right to the distribution of his or her Account or his or her Beneficiary's right to the distribution of such Account.

## Section 10.13. Pre-2005 Oxford Plan.

- (a) Pre-2005 Deferrals. The Pre-2005 Oxford Plan and any liabilities thereunder were made a part of this Plan effective as of January 1, 2006. Any amounts deferred before January 1, 2005 under the Pre-2005 Oxford Plan (as determined in accordance with Section 409A of the Code) shall be governed by the terms of the Pre-2005 Oxford Plan, which is attached to this Plan as Exhibit A. Nothing herein is intended to give any additional benefits to or enhance the benefits of a participant in the Pre-2005 Oxford Plan and it is intended that amounts deferred under that plan (and any earnings on such amounts) are not subject to Section 409A of the Code. There shall be no further deferrals under the terms of the Pre-2005 Oxford Plan after December 31, 2004.
- (b) <u>Post-2004 and Pre-2006 Deferrals.</u> The Pre-2005 Ox ford Plan was amended to comply with Section 409A of the Code in the form of the addendum attached to this Plan as Exhibit B with respect to amounts deferred in taxable years beginning after December 31, 2004 and before January 1, 2006.

## Section 10.14. Tommy Bahama Plan.

- (a) Pre-2005 Deferrals. The Tommy Bahama Plan and any liabilities thereunder were made a part of this Plan effective as of January 1, 2006. Any amounts deferred before January 1, 2005 under the Tommy Bahama Plan (as determined in accordance with Section 409A of the Code) shall be governed by the terms of the Tommy Bahama Plan, which is attached to this Plan as Exhibit C. Nothing herein is intended to give any additional benefits to or enhance the benefits of a participant in the Tommy Bahama Plan and it is intended that amounts deferred under that plan (and any earnings on such amounts) are not subject to Section 409A of the Code. There shall be no further deferrals under the terms of the Tommy Bahama Plan after December 31, 2004.
- (b) <u>Post-2004 and Pre-2006 Deferrals.</u> The Tommy Bahama Plan was amended to comply with Section 409A of the Code in the form of the addendum attached to this Plan as Exhibit D with respect to amounts deferred in taxable years beginning after December 31, 2004 and before January 1, 2006. Each Deferred Compensation Account maintained under the Tommy Bahama Plan for a person who was an active Employee on January 1, 2006 became fully vested as of January 1, 2006.
- Section 10.15. Special Transition Bonus Election. Notwithstanding any contrary provision in the Plan, the Committee was authorized in its discretion to allow an Eligible Employee to elect during 2007 (in accordance with procedures established by the Committee and in compliance with transition guidance provided under IRS Notice 2006-79) to defer up to the Maximum Deferral Percentage of any bonus attributable to a performance period beginning in 2007 that otherwise would be a short-term deferral (within the meaning of Section 409A of the Code) payable in 2007 or 2008.

ATTEST:	OXFORD INDUSTRIES, INC.
/s/ Suraj A. Palakshappa	/s/ Thomas E. Campbell
	18

IN WITNESS WHEREOF, Oxford Industries, Inc. has caused this Plan document to be executed as of this 13th day of June, 2012.

# EXHIBIT A

# OXFORD INDUSTRIES, INC.

# NON-QUALIFIED DEFERRED COMPENSATION PLAN

## OXFORD INDUSTRIES, INC.

#### NON-QUALIFIED DEFERRED COMPENSATION PLAN

## ARTICLE I - PURPOSE; EFFECTIVE DATE

- 1.1. Purpose. The purpose of this Oxford Industries, Inc. Non-Qualified Deferred Compensation Plan (the "Plan") is to permit a select group of management and highly compensated employees of Oxford Industries, Inc. and its subsidiaries (the "Company") to defer the receipt of income which would otherwise become payable to them. It is intended that this Plan, by providing this deferral opportunity, will assist the Company in attracting and retaining individuals of exceptional ability.
- 1.2. Effective Date. The Plan shall be effective as of January 1, 2001.

## **ARTICLE II - DEFINITIONS**

For the purpose of this Plan, the following terms shall have the meanings indicated unless the context clearly indicates otherwise:

- 2.1. Account(s). "Account(s)" means the account or accounts maintained on the books of the Company used solely to calculate the amount payable to each Participant under this Plan and shall not constitute a separate fund or assets. The Accounts available for each Participant shall be identified as:
  - a) Retirement Account and/or,
  - b) Up to two In-Service Accounts.
- 2.2. **Beneficiary**. "Beneficiary" means the person, persons or entity, as designated by the Participant, entitled under Article VI to receive any Plan benefits payable after the Participant's death.
- 2.3. **Board**. "Board" means the Board of Directors of the Company.
- 2.4. Change in Control. A "Change in Control" shall occur if:
  - a) Any "person" or "group" (within the meaning of Sections 13(d) and 14(d)(2) of the Securities Exchange Act of 1934, as amended) becomes the "beneficial owner" (as defined in Rule 13-d under such Act) of more than fifty (50%) of the then outstanding voting stock of the Company, other than through a transaction arranged by, or consummated with the prior approval of, the Board; or
  - b) During any period of two (2) consecutive years, individuals who at the beginning of such period constitute the Board (and any new Director whose election by the Board or whose nomination for election by the stockholders of the Company was approved by a vote of at least two-thirds (2/3) of the Directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority thereof; or

- c) The shareholders of Company approve a merger or consolidation of Company with any other corporation, other than a merger or consolidation which would result in the voting securities of a Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than eighty percent (80%) of the combined voting power of the voting securities of Company or such surviving entity outstanding immediately after such merger or consolidation; or
- d) The shareholders of Company approve a plan of complete liquidation of Company or an agreement for the sale or disposition by Company of all or substantially all of the Company's assets.
- 2.5. **Committee.** "Committee" means the Committee appointed by the Board to administer the Plan pursuant to Article VII.
- 2.6. <u>Company</u>. "Company" means Oxford Industries, Inc., a Georgia corporation, and any directly or indirectly affiliated subsidiary corporations, any other affiliate which is designated by the Board, or any successor to the business thereof.
- 2.7. Compensation. "Compensation" means the base salary, commissions and/or bonus compensation payable to a Participant with respect to employment services performed for the Company by the Participant and Company matching contributions that would otherwise be included in "wages" for purposes of federal income tax withholding. For purposes of this Plan, Compensation shall be calculated before reduction for any amounts deferred by the Participant pursuant to the Company's tax qualified plans which may be maintained under Section 401(k) or Section 125 of the Internal Revenue Code of 1986, as amended, (the "Code"), or pursuant to this Plan or any other non-qualified plan which permits the voluntary deferral of compensation. Inclusion of any other forms of compensation is subject to Committee Approval.
- 2.8. <u>Deferral Commitment</u>. "Deferral Commitment" means a commitment made by a Participant and accepted by the Committee to defer a portion of Compensation paid to or earned such Participant during a specified Deferral Period. The Deferral Commitment shall apply to each payment of salary and/or bonus, as applicable, earned by or payable to a Participant for a given Deferral Period, and shall specify the Account or Accounts to which such deferrals shall be credited. Such designation shall be made in whole percentages and shall be made in a form acceptable to the Committee. Once made, a Deferral Commitment shall, except as otherwise provided herein, be irrevocable by the Participant for the Deferral Period to which it applies.
- 2.9. <u>Deferral Period</u>. "Deferral Period" means a calendar year to which a Deferral Commitment applies.
- 2.10. <u>Determination Date</u>. "Determination Date" means the last business day of each calendar month.
- 2.11. <u>Disability</u>. "Disability" means a physical or mental condition that prevents the Participant from satisfactorily performing the Participant's duties for Company. The Committee shall, in its sole discretion, determine the existence of Disability and may rely on such evidence of disability as it deems appropriate, including a determination of disability under the Company's long-term disability plan or advice from a medical examiner satisfactory to the Committee.
- 2.12. <u>Discretionary Contribution</u>. "Discretionary Contribution" means the Company contribution credited to a Participant's Account(s) under Section 4.5, below.

- 2.13. <u>Distribution Election</u>. "Distribution Election" means the form prescribed by the Committee and completed by the Participant, indicating the chosen form of payment for benefits payable from each Account under this Plan, as elected by the Participant.
- 2.14. Financial Hardship. "Financial Hardship" means a severe, unexpected and unforeseeable financial hardship of the Participant resulting from a Disability of the Participant, a sudden and unexpected illness or accident of the Participant or of a dependent of the Participant, uninsured loss of the Participant's property due to casualty, or other similar extraordinary and unforeseeable circumstance arising as a result of events beyond the control of the Participant. Financial Hardship shall be determined based upon such standards as are, from time to time, established by the Committee, and such determination shall be in the sole discretion of the Committee.
- 2.15. 401(k) Plan. "401(k) Plan" means the Oxford Industries, Inc. Retirement Savings Plan, or any other successor defined contribution plan maintained by the Company that qualifies under Section 401(a) of the Code and satisfies the requirements of Section 401(k) of the Code.
- 2.16. Investment Option. "Investment Option" means one or more of the independently established funds or indices that are identified and listed by the Committee. These Investment Options are used solely to calculate the investment gains or losses that are credited to each Participant's Account(s) in accordance with Article IV. The determination of the investment gains or losses attributable to the performance of each Investment Option shall be made by the Committee in its reasonable discretion. The Committee shall select and provide a list of the various Investment Options available to the Participants with respect to this Plan; provided, that the Committee may amend such list from time to time in its sole discretion.
- 2.17. <u>Matching Contribution</u>. "Matching Contribution" means the Company contribution credited to a Participant's Account(s) under Section 4.4, below.
- 2.18. **Participant**. "Participant" means any employee who is eligible pursuant to Section 3.1 to participate in this Plan and who has elected to defer Compensation under this Plan in accordance with Article III. Such employee shall remain a Participant in this Plan for the period of deferral and until such time as all benefits payable under this Plan have been paid in accordance with the provisions hereof.
- 2.19. Plan. "Plan" means this Oxford Industries, Inc. Non-Qualified Deferred Compensation Plan, as amended from time to time.
- 2.20. **Retirement.** "Retirement" means the termination of employment with the Company of the Participant on or after attaining age 65 or on or after attaining age 55 with at least 7 Years of Service, or a termination of employment that has received the approval by the Committee as qualifying as a Retirement under this Plan.
- 2.21. Years of Service. "Years of Service" shall have the meaning provided for such term for purposes of vesting under the 401(k) Plan, whether or not the Participant is a participant in such plan.

## ARTICLE III - ELIGIBILITY AND PARTICIPATION

#### 3.1. Eligibility and Participation.

- a) <u>Eligibility</u>. Eligibility to participate in the Plan for a Deferral Period shall be limited to a select group of management or highly compensated employees of the Company designated by management, from time to time, and approved by the Committee.
- b) <u>Participation</u>. An employee's participation in the Plan for a Deferral Period shall be effective upon notification to the employee by the Committee of eligibility to participate, completion and submission of a Deferral Commitment, Distribution Election Form and Investment Allocation Form to the Committee no later than the deadline established by the Committee, and the acceptance by the Committee of such forms
- 3.2. Form of Deferral. A Deferral Commitment shall be made with respect to each payment of salary, commissions and/or bonus earned by or payable to a Participant during the Deferral Period, and shall designate the portion of each deferral that shall be allocated among the various Accounts. The Participant shall set forth the amount to be deferred as a full percentage of salary, commission and/or bonus. In addition, the Participant shall specify in a separate form (known as the "Investment Allocation Form") filed with the Committee, the Participant's initial allocation of the amounts deferred into each Account among the various available Investment Options.
- 3.3. <u>Limitations on Deferral Commitments</u>. The maximum percentage of each payment of base salary and commissions that may be deferred during a Deferral Period shall be fifty percent (50%), and the maximum percentage of bonus compensation that may be deferred during the Deferral Period shall be one hundred percent (100%). The Committee may set such additional limitations for a Deferral Period, as it determines in its sole discretion, once it has reviewed the participation level for such Deferral Period.
- 3.4. <u>Commitment Limited by Termination</u>. If a Participant terminates employment with Company prior to the end of a Deferral Period, the Deferral Commitment in effect for such Deferral Period shall be revoked as of the date of such termination.
- 3.5. <u>Modification of Deferral Commitment</u>. Except as provided in Sections 3.4 and 5.5, a Deferral Commitment for a Deferral Period shall be irrevocable by the Participant during such Deferral Period.
- 3.6. Change in Employment Status. If the Committee, in its sole discretion, determines that the Participant no longer qualifies as a member of a select group of management or highly compensated employees, as determined in accordance with the Employee Retirement Income Security Act of 1974, as amended, the Committee may, in its sole discretion, terminate any Deferral Commitment currently in effect, prohibit the Participant from making any future Deferral Commitments and/or distribute the Participant's Account Balances in accordance with Article V of this Plan as if the Participant had terminated employment with the Company as of that time.

## ARTICLE IV - DEFERRED COMPENSATION ACCOUNT

4.1. Accounts. The Compensation deferred by a Participant under the Plan, any Matching Contributions deferred under the Plan, Discretionary Contributions and Earnings shall be credited to the Participant's Account(s). The Participant shall designate the portion of each deferral that will be

credited to each Account as set forth in Section 3.2(a). These Accounts shall be used solely to calculate the amount payable to each Participant under this Plan and shall not constitute a separate fund of assets.

- 4.2. Timing of Credits; Withholding. A Participant's deferred Compensation shall be credited to each Account designated by the Participant on the last business day of the month during which the compensation deferred would have otherwise been payable to the Participant. Any Matching Contributions shall be credited to each Account on the last business day of the month during which the deferred Compensation to which the Matching Contributions relates was credited to each Account. Any Discretionary Contributions shall be credited to the appropriate Account(s) as provided by the Committee. Any withholding of taxes or other amounts with respect to deferred Compensation that is required by local, state or federal law shall be withheld from the Participant's corresponding non-deferred portion of the Compensation to the maximum extent possible, and any remaining amount shall reduce the amount credited to the Participant's Account in a manner specified by the Committee.
- 4.3. Investment Options. A Participant shall designate, at a time and in a manner acceptable to the Committee, one or more Investment Options for each Account to be used for the sole purpose of determining the amount of Earnings to be credited or debited to such Account. Such election shall designate the portion of each deferral of Compensation made into each Account that shall be allocated among the available Investment Option(s), and such election shall apply to each succeeding deferral of Compensation until such time as the Participant shall file a new election with the Committee. Upon notice to the Committee, the Participant may also reallocate the balance in each Investment Option among the other available Investment Options as of the next succeeding Determination Date, but in no event shall such re-allocation occur more frequently than monthly.
- 4.4. <u>Matching Contributions</u>. The Company shall credit the portion elected by the Participant of the Company's total Matching Contribution on behalf of the Participant to the Account designated by the Participant.
- 4.5. <u>Discretionary Contributions</u>. The Company may make Discretionary Contributions to a Participant's Account. Discretionary Contributions shall be credited and shall become vested at such times and in such amounts as recommended by the Committee and approved by the Compensation Committee of the Board, or the Board, in its sole discretion. Unless the Committee specifies otherwise, such Discretionary Contribution shall be allocated among the various Accounts in the same proportion as set forth in section 4.1.
- 4.6. <u>Determination of Accounts</u>. Each Participant's Account as of each Determination Date shall consist of the balance of the Account as of the immediately preceding Determination Date, adjusted as follows:
  - a) New Deferrals. Each Account shall be increased by any deferrals credited since the prior Determination Date.
  - b) <u>Company Contributions</u>. Each Account shall be increased by any Matching and/or Discretionary Contributions credited since the prior Determination Date.
  - c) <u>Distributions</u>. Each Account shall be reduced by the amount of each benefit payment made from that Account since the prior Determination Date. Distributions shall be deemed to have been made proportionally from each of the Investment Options maintained within such Account based on the proportion that such Investment Option bears to the sum of all

- Investment Options maintained within such Account for that Participant as of the Determination Date immediately preceding the date of payment.
- d) Earnings. Each Account shall be increased or decreased by the Earnings credited to such Account since the prior Determination Date as though the balance of that Account as of the beginning of the current month had been invested in the applicable Investment Options chosen by the Participant.
- 4.7. <u>Vesting of Accounts</u>. Each Participant shall be vested in the amounts credited to such Participant's Account and Earnings thereon as follows:
  - a) Amounts Deferred. A Participant shall be one hundred percent (100%) vested at all times in the Participant's deferrals of salary, commission and/or bonus and the Earnings thereon.
  - b) <u>Matching Contributions</u>. A Participant shall be one hundred percent (100%) vested at all times in the Matching Contributions made under the Plan and the Earnings thereon.
  - c) <u>Discretionary Contributions</u>. A Participant's Discretionary Contributions and Earnings thereon shall become vested as determined by the Committee and as approved by the Committee of the Board, or the Board.
- 4.8. Statement of Accounts. Each Participant shall receive a statement showing the balances in the Participant's Account on a quarterly basis.

#### ARTICLE V - PLAN BENEFITS

- 5.1. Retirement Account. The vested portion of a Participant's Retirement Account shall be distributed to the Participant upon the Participant's termination of employment with the Company. Benefits under this section shall be payable the January following termination of employment, but no sooner than thirty (30) days following termination. The form of benefit payment shall be that form selected by the Participant pursuant to Section 5.6 unless the Participant terminates employment prior to Retirement, in which event, the Retirement Account shall be paid in the form of a lump sum payment unless the Committee determines, upon written request, to allow the payment to be made in the form designation on the Distribution Election Form.
- 5.2. In-Service Account. The vested portion of a Participant's In-Service Account shall be distributed to the Participant upon the date chosen by the Participant in the Distribution Election Form, but in no event shall the date specified for commencement of payment be earlier than five (5) years from the beginning of the first Deferral Period during which the Participant elected compensation to be deferred into that Account. The form of benefit payment shall be that form selected by the Participant pursuant to Section 5.7. However, if the Participant terminates employment with the Company prior to the date so chosen by the Participant, the vested portion of the In-Service Account shall be added to the Retirement Account as of the date of termination of service and shall be paid in accordance with the provisions of Section 5.1.
- 5.3. <u>Death Benefit.</u> Upon the death of a Participant, Company shall pay to the Participant's Beneficiary an amount equal to the remaining unpaid and vested Account balance in each Account in the form of a lump sum payment.

- 5.4. Hardship Distributions. Upon a finding that a Participant has suffered a Financial Hardship, the Committee may, in its sole discretion, amend the existing Deferral Commitment, or make distributions from any or all of the Participant's Accounts. The amount of such distribution shall be limited to the amount reasonably necessary to meet the Participant's needs resulting from the Financial Hardship plus applicable taxes, and shall not exceed the Participant's vested Account balances. If payment is made from any or all of the Participant's accounts due to Financial Hardship, the Participant's deferrals under this Plan shall cease for the remainder of the current Deferral Period and the next subsequent Deferral Period.
- 5.5. Withdrawal with Penalty. The Participant may elect, in the sole discretion of the Participant, to withdraw from participation in this Plan, and to cause the total vested portion of the Participant's Account balances to be distributed in accordance with this Article V as if the Participant had terminated service with the Company as of the time of such election, except that such Account balances shall be reduced by a penalty of ten percent (10%) of such Account Balances. The Participant's account balances, less the 10% penalty, shall be paid to the Participant or the Participant's Beneficiary as soon as administratively practical in the form of a lump sum payment. The Participant, or the Participant's Beneficiary, may file such an election at any time prior to the complete payment of benefits due under this Plan. Upon the filing of this election, any Deferral Commitment for the current Deferral Period shall be terminated and the Participant shall be prohibited from participating in this Plan for the next subsequent Deferral Period
- 5.6. **Form of Payment**. Unless otherwise specified in paragraphs 5.1, 5.2, 5.3, or 5.5, the benefits payable from any Account under this Plan shall be paid in the form of benefit as provided below, and as specified by the Participant in the Distribution Election, which election shall be irrevocable once made. The permitted forms of benefit payments are:
  - a) A lump sum amount which is equal to the vested Account balance;
  - b) In the event of distributions from the Retirement Account, annual installments for a period of five (5), ten (10) or fifteen (15) years where the annual payment shall be equal to the balance of the Account immediately prior to the payment, multiplied by a fraction, the numerator of which is one (1) and the denominator of which commences at the number of annual payment initially chosen and is reduced by one (1) in each succeeding year. Earnings on the unpaid balance shall be based on the most recent allocation among the available Investment Options chosen by the Participant, made in accordance with Section 4.3;
  - c) In the event of distributions from the In-Service Account, annual installments for a period up to five (5) where the annual payment shall be equal to the balance of the Account immediately prior to the payment, multiplied by a fraction, the numerator of which is one (1) and the denominator of which commences at the number of annual payment initially chosen and is reduced by one (1) in each succeeding year. Earnings on the unpaid balance shall be based on the most recent allocation among the available Investment Options chosen by the Participant, made in accordance with Section 4.3; and,
  - d) Any other form of payment requested by the Participant and approved by the Committee.
- 5.7. Small Account. Except as otherwise determined by the Committee, if the total of a Participant's vested, unpaid Account balances as of the Participant's Retirement is less than \$25,000, the remaining unpaid, vested Account(s) shall be paid in a lump sum, notwithstanding any election by the Participant to the contrary.

- 5.8. Withholding; Payroll Taxes. The Company shall withhold from any payment made pursuant to this Plan any taxes required to be withheld from such payments under local, state or federal law.
- 5.9. Payment to Guardian. If a Plan benefit is payable to a minor or a person declared incompetent or to a person incapable of handling the disposition of the property, the Committee may direct payment to the guardian, legal representative or person having the care and custody of such minor, incompetent or person. The Committee may require proof of incompetency, minority, incapacity or guardianship as it may deem appropriate prior to distribution. Such distribution shall completely discharge the Committee and Company from all liability with respect to such benefit.
- 5.10. Effect of Payment. The full payment of the applicable benefit under this Article V shall completely discharge all obligations on the part of the Company to the Participant (and the Participant's Beneficiary) with respect to the operation of this Plan, and the Participant's (and Participant's Beneficiary's) rights under this Plan shall terminate.

#### **ARTICLE VI - BENEFICIARY DESIGNATION**

- 6.1. Beneficiary Designation. Each Eligible Participant shall have the right, at any time, to designate one (1) or more persons or entities as Beneficiary (both primary as well as secondary) to whom benefits under this Plan shall be paid in the event of Participant's death prior to complete distribution of the Participant's vested Account balance. Each Beneficiary designation shall be in a written form prescribed by the Committee and shall be effective only when filed with the Committee during the Participant's lifetime.
- 6.2. <u>Changing Beneficiary</u>. Any Beneficiary designation may be changed by the filing of a new Beneficiary designation with the Committee.
- 6.3. No Beneficiary Designation. If any Participant fails to designate a Beneficiary in the manner provided above, if the designation is void, or if the Beneficiary designated by a deceased Participant dies before the Participant or before complete distribution of the Participant's benefits, the Participant's Beneficiary shall be the Participant's estate.
- 6.4. Effect of Payment. Payment to the Beneficiary shall completely discharge the Company's obligations under this Plan.

## ARTICLE VII - ADMINISTRATION

- 7.1. <u>Committee</u>; <u>Duties</u>. This Plan shall be administered by the Committee, which shall consist of not less than three (3) persons appointed by the Board, except after a Change in Control as provided in Section 7.5. The Committee shall have the authority to make, amend, interpret and enforce all appropriate rules and regulations for the administration of the Plan and decide or resolve any and all questions, including interpretations of the Plan, as may arise in such administration. A majority vote of the Committee members shall control any decision. Members of the Committee may be Participants under this Plan.
- 7.2. <u>Agents</u>. The Committee may, from time to time, employ agents and delegate to them such administrative duties as it sees fit, and may from time to time consult with counsel who may be counsel to the Company.

- 7.3. **Binding Effect of Decisions**. The decision or action of the Committee with respect to any question arising out of or in connection with the administration, interpretation and application of the Plan and the rules and regulations promulgated hereunder and with respect to determining eligibility to participate in the Plan, whether, when and in what amount benefits are payable under the Plan, and any factual determinations shall made in the Committee's sole discretion and shall be final, conclusive and binding upon all persons.
- 7.4. <u>Indemnity of Committee</u>. The Company shall indemnify and hold harmless the members of the Committee against any and all claims, loss, damage, expense or liability arising from any action or failure to act with respect to this Plan on account of such member's service on the Committee, except in the case of gross negligence or willful misconduct.
- 7.5. Election of Committee After Change in Control. After a Change in Control, vacancies on the Committee shall be filled by majority vote of the remaining Committee members and Committee members may be removed only by such a vote. If no Committee members remain, a new Committee shall be elected by majority vote of the Participants in the Plan immediately preceding such Change in control. No amendment shall be made to Article VII or other Plan provisions regarding Committee authority with respect to the Plan without prior approval by the Committee.

## ARTICLE VIII - CLAIMS PROCEDURE

- 8.1. <u>Claim.</u> Any person or entity claiming a benefit, requesting an interpretation or ruling under the Plan (hereinafter referred to as "Claimant"), or requesting information under the Plan shall present the request in writing to the Committee, which shall respond in writing as soon as practicable.
- 8.2. **Denial of Claim.** If the claim or request is denied, the written notice of denial shall state:
  - a) The reasons for denial, with specific reference to the Plan provisions on which the denial is based;
  - b) A description of any additional material or information required and an explanation of why it is necessary; and
  - c) An explanation of the Plan's claim review procedure.
- 8.3. **Review of Claim.** Any Claimant whose claim or request is denied or who has not received a response within sixty (60) days may request a review by notice given in writing to the Committee within sixty (60) days following such denial or lack of response. The claim or request shall be reviewed by the Committee.
- 8.4. Final Decision. The decision on review shall normally be made within sixty (60) days after the Committee's receipt of claimant's claim or request. If an extension of time is required for a hearing or other special circumstances, the Claimant shall be notified and the time limit shall be one hundred twenty (120) days. The decision shall be in writing and shall state the reasons and the relevant Plan provisions. All decisions on review shall be made in the Committee's sole discretion and shall be final and binding on all parties.

## ARTICLE IX - AMENDMENT AND TERMINATION OF PLAN

- 9.1. Amendment. The Board may at any time amend the Plan by written instrument, notice of which is given to all Participants and to Beneficiaries receiving installment payments, subject to the following; provided, that no amendment shall reduce the amount accrued in any Account as of the date such notice of the amendment is given.
- 9.2. <u>Company's Right to Terminate</u>. The Board may at any time partially or completely terminate the Plan, as it determines in its sole discretion.
  - a) Partial Termination. The Board may partially terminate the Plan by instructing the Committee not to accept Deferral Commitments for future Deferral Periods. If such a partial termination occurs, the Plan shall continue to operate and be effective with regard to Deferral Commitments entered into prior to the effective date of such partial termination.
  - b) <u>Complete Termination</u>. The Board may completely terminate the Plan by instructing the Committee not to accept Deferral Commitments for future Deferral Periods, and by terminating all current Deferral Commitments. In the event of complete termination, the Plan shall cease to operate and Company shall distribute each Account to the appropriate Participant. Payment shall be made as a lump sum.

## ARTICLE X - MISCELLANEOUS

- 10.1. Unfunded Plan. This plan is an unfunded plan maintained primarily to provide deferred compensation benefits for a select group of "management or highly-compensated employees" within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and therefore is exempt from the provisions of Parts 2, 3 and 4 of Title I of ERISA. Accordingly, the Board may take such actions as it, in its sole discretion, deems appropriate if it is determined by the United States Department of Labor, a court of competent jurisdiction, or an opinion of counsel that the Plan constitutes an employee pension benefit plan within the meaning of Section 3 (2) of ERISA (as currently in effect or hereafter amended) which is not so exempt.
- 10.2. <u>Unsecured General Creditor</u>. Notwithstanding any other provision of this Plan, Participants and Participants' Beneficiary shall be unsecured general creditors, with no secured or preferential rights to any assets of Company or any other party for payment of benefits under this Plan. Any property held by Company for the purpose of generating the cash flow for benefit payments shall remain its general, unpledged and unrestricted assets. Company's obligation under the Plan shall be an unfunded and unsecured promise to pay money in the future.
- 10.3. **Trust Fund.** Company shall be responsible for the payment of all benefits provided under the Plan. At its discretion, Company may establish one (1) or more trusts, with such trustees as the Board may approve, for the purpose of assisting in the payment of such benefits. Although such a trust shall be irrevocable, its assets shall be held for payment of all Company's general creditors in the event of insolvency. To the extent any benefits provided under the Plan are paid from any such trust, Company shall have no further obligation to pay them. If not paid from the trust, such benefits shall remain the obligation of Company.
- 10.4. Nonassignability. Neither a Participant nor any other person shall have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate or convey in advance of actual receipt the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are, expressly declared to be unassignable and non-transferable. No part of

the amounts payable shall, prior to actual payment, be subject to seizure or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, nor be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency.

- 10.5. Not a Contract of Employment. This Plan shall not constitute a contract of employment between Company and the Participant. Nothing in this Plan shall give a Participant the right to be retained in the service of Company or to interfere with the right of the Company to discipline or discharge a Participant at any time.
- 10.6. **Protective Provisions.** A Participant shall cooperate with Company by furnishing any and all information requested by Company in order to facilitate the payment of benefits hereunder and by taking such action as may be requested by Company.
- 10.7. Governing Law. The provisions of this Plan shall be construed and interpreted according to the laws of the State of Georgia, except as preempted by federal law.
- 10.8. **Validity.** If any provision of this Plan shall be held illegal or invalid for any reason, said illegality or invalidity shall not affect the remaining parts hereof, but this Plan shall be construed and enforced as if such illegal and invalid provision had never been inserted herein.
- 10.9. Notice. Any notice required or permitted under the Plan shall be sufficient if in writing and hand delivered or sent by registered or certified mail. Such notice shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification. Mailed notice to the Committee shall be directed to the company's primary business address. Mailed notice to a Participant or Beneficiary shall be directed to the individual's last known address in company's records
- 10.10. <u>Successors</u>. The provisions of this Plan shall bind and inure to the benefit of Company and its successors and assigns. The term successors as used herein shall include any corporate or other business entity which shall, whether by merger, consolidation, purchase or otherwise acquire all or substantially all of the business and assets of Company, and successors of any such corporation or other business entity.

#### OXFORD INDUSTRIES, INC.

By:

Name: Thomas E. Campbell
Title: Vice President

A-12

## EXHIBIT B

#### PRE-2005 OXFORD PLAN

## SPECIAL RULES APPLICABLE TO 2005 COMPENSATION

Notwithstanding any other provision of the Pre-2005 Oxford Plan to the contrary, the provisions of this Exhibit B shall supersede all inconsistent provisions of the Pre-2005 Oxford Plan with respect to amounts deferred in taxable years beginning after December 31, 2004 and before January 1, 2006 (and earnings on such amounts). All other provisions of the Pre-2005 Oxford Plan shall apply with respect to such deferrals to the extent not inconsistent with the provisions of this Exhibit B or Section 409A of the Code, as determined by the Committee in its sole and absolute discretion. This Exhibit B is intended to (a) satisfy the requirements of Section 409A(a)(2), (3) and (4) of the Code for deferrals made after December 31, 2004 and before January 1, 2006 and (b) not constitute a material modification of the Pre-2005 Oxford Plan with respect to amounts deferred before January 1, 2005.

- 1. <u>Account(s)</u>. A separate bookkeeping account shall be established to account for deferrals made in taxable years beginning after December 31, 2004 and before January 1, 2006 and any earnings on such deferrals. The portion of any Account that was not fully vested on December 31, 2004 shall be treated as a deferral made in taxable years beginning after December 31, 2004.
- 2. <u>Participation</u>. A Deferral Commitment shall only apply to defer a portion of Compensation consisting of base salary, commissions and/or bonus compensation earned by a Participant during the Deferral Period. The deadline for completion and submission of a Deferral Commitment and Distribution Election Form is December 31, 2004.
  - 3. <u>Change in Employment Status.</u> The provisions of Section 3.6 of the Pre-2005 Oxford Plan shall not apply.
- 4. <u>Hardship Distributions.</u> The provisions of Sections 2.14 and 5.4 of the Pre-2005 Oxford Plan shall not apply, and Section 7.2(d) of the Plan shall apply as if incorporated in the Pre-2005 Oxford Plan.
- 5. <u>Distribution of Retirement Account.</u> In order for a termination of employment with the Company to trigger a distribution, the termination of employment must qualify as a "separation from service" within the meaning of Section 409A of the Code and the regulations thereunder. Distribution upon termination of employment will be made in the form selected by the Participant, unless the Participant terminates employment prior to Retirement, in which case the Retirement Account shall be paid in the form of a lump sum payment, with no Committee discretion to pay in another form. A distribution made as a result of the Participant's separation from service (whether prior to or upon Retirement) will commence in the first calendar month that is 6 months from the date the Participant terminates employment. "Retirement" means the separation from service with the Company of the Participant on or after attaining age 55 with at least 7 Years of Service.
- 6. <u>In-Service Account.</u> A Participant may revise an in-service distribution election to change the time of distribution; provided, however, that (1) the revision will not take effect until 12 months after the date it is made, (2) the revision must be made at least 12 months before the in-service

distribution otherwise would commence, and (3) the in-service distribution will be deferred for at least 5 years from the date the in-service distribution would have commenced in the absence of the revision.

- 7. <u>Death.</u> If distribution is made as a result of the Participant's death under Section 5.3 of the Pre-2005 Oxford Plan, distribution will commence in the first month of the calendar quarter immediately following the quarter in which his or her death occurred.
  - 8. Withdrawal with Penalty. The provisions of Section 5.5 of the Pre-2005 Oxford Plan shall not apply.
  - 9. <u>Delay of Payments Under Certain Circumstances</u>. Section 7.2(e) of the Plan shall apply as if incorporated in the Pre-2005 Oxford Plan.
- 10. <u>Amendment and Complete Termination</u>. The provisions of Sections 9.1 and 9.2(b) of the Pre-2005 Oxford Plan shall not apply, and Section 10.12 of the Plan shall apply as if incorporated in the Pre-2005 Oxford Plan.

# EXHIBIT C

# VIEWPOINT INTERNATIONAL, INC.

# NONQUALIFIED DEFERRED COMPENSATION PLAN

#### NONQUALIFIED DEFERRED COMPENSATION PLAN

#### **SECTION 1**

#### Definitions

- 1.1. Affiliate. "Affiliate" means any corporation, partnership, joint venture, association or similar organization or entity that is required to be aggregated with the Company pursuant to Code Sections 414(b), (c), or (m).
- 1.2. <u>Code</u>. "Code" means the Internal Revenue Code of 1986, as amended from time to time. Any reference to a section of the Code includes any comparable section or sections of any future legislation that amends, supplements or supersedes that section.
- 1.3. Company. "Company" means Viewpoint International, Inc. located at 1071 Avenue of the Americas, NY, NY 10018, employer tax identification number 13-3676108, which Company has established the Plan, as set forth herein.
- 1.4. Compensation. "Compensation" means (select one option):
- Option 1. 🗵 Total taxable salary, bonuses and commissions paid to a Participant by the Employer (determined without regard to any amounts in the Participant's Deferred Compensation Account).
- Option 2. 

  Total taxable salary and commissions of the Participant paid or accrued by the Employer, but not including the value of any bonuses, stock options, stock appreciation rights (determined without regard to any amounts in the Participant's Deferred Compensation Account).

Option 3. 

Other

- 1.5. <u>Deferred Compensation Account</u>. "Deferred Compensation Account" means the book-keeping account maintained under the Plan in the Participant's name to reflect amounts deferred under the Plan pursuant to Section 3 (as adjusted under Section 4) and (if elected by the Company) any Employer Discretionary Contributions made on behalf of the Participant (as adjusted under Section 4).
- 1.6. <u>Deferral Election</u>. "Deferral Election" means a written notice filed by the Participant with the Employer specifying the Compensation or bonus to be deferred by the Participant.
- 1.7. <u>Distribution Date</u>. "Distribution Date" means the date a Participant terminates employment or association with the Employers for whatever reason, unless such termination of employment is for Good Cause.
- 1.8. Early Retirement Date. "Early Retirement Date" means (select one option):

☐ The date the Participant attains years of age.

🗵 The date the Participant attains 55 years of age and has been employed by the Company or its Affiliates for at least 10 years.

1.9. Effective Date. "Effective Date" means July 20, 2001.

1.10. Employee. "Employee" means an employee of an Employer who meets the eligibility criteria set forth in Subsection 3.1 of the Plan and who is
member of a select group of management or highly compensated employees as defined under ERISA or the regulations thereunder.

- 1.11. Employer. "Employer" means, individually, the Company and each Affiliate of the Company that adopts the Plan in accordance with Subsection 7.1. The Company and any Affiliates that adopt the Plan are sometimes collectively referred to herein as the "Employers."
- 1.12. <u>ERISA</u>. "ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time. Any reference to a section of ERISA includes any comparable section or sections of any future legislation that amends, supplements or supersedes that section.
- 1.13. Excess Contributions. "Excess Contributions" means contributions determined to be excess contributions or excess deferrals (as such terms are defined in the regulations under Section 401(k) of the Code) for the Plan Year under a plan maintained by an Employer that is qualified under Sections 401(a) and 401(k) of the Code.
- 1.14. Independent Contractor. "Independent Contractor" means an individual who is not a common-law employee of an Employer but who receives payments from the Employer for services rendered.

payments from the Employer for services rendered.			
1.15. Normal Retirement Date. "Normal Retirement Date" means (select one option):			
☐ The date the Participant attains years of age.			
☑ The date the Participant attains 65 years of age and has been employed by the Company or its Affiliates for at least 10 years.			
1.16. Participant. "Participant" means an Employee or Independent Contractor who meets the eligibility criteria set forth in Subsection 3.1 and who has made a Deferral Election in accordance with the terms of the Plan.			
1.17. Plan. "Plan" means the provisions of the Plan, as set forth herein, including the variable provisions selected and agreed to by the Company.			
1.18. Plan Administrator. The "Plan Administrator" means (select one option):			
□ The Company.			
☐ A committee of at least members appointed by the Company			
☑ The C.F.O. (insert title) of the Company.			
□ Other			

- 1.19. <u>Plan Year</u>. "Plan Year' means the calendar year. However, if the Effective Date of the Plan is other than January 1 of a year, the initial Plan Year shall be a short Plan Year, beginning on the Effective Date and ending on the following December 31.
- 1.20. <u>Unforeseeable Financial Emergency</u>. "Unforeseeable Financial Emergency" means a severe financial hardship of the Participant resulting from:

(b) Loss	of the Participant's principal residence due to casualty; or
(c) Such	other similar extraordinary and unforeseeable circumstances resulting from events beyond the control of the Participant.
Whether a Par	rticipant has an Unforeseeable Financial Emergency shall be determined in the sole discretion of the Plan Administrator.
1.21. Valuatio	on Date. "Valuation Date" means (select one option):
⊠ Any busi	iness day.
☐ The last o	day of any calendar month.
☐ The last o	day of any calendar quarter.
☐ The last o	day of the Plan Year.
□ Ot	her

(a) A sudden and unexpected illness or accident of the Participant or of a dependent of the Participant;

1.22. Other Definitions. In addition to the terms defined in this Section 1, other terms are defined when first used in later Sections of this Plan.

#### **SECTION 2**

#### Purpose and Administration

- 2.1. <u>Purpose</u>. The Company has established the Plan primarily for the purpose of providing deferred compensation to a select group of management or highly compensated employees of the Employers. The Plan is intended to be a top-hat plan described in Section 201(2) of ERISA. If elected by the Company under Subsection 3.1 of the Plan, Independent Contractors also may participate in the Plan. The Company intends that the Plan (and each Trust under the Plan (as described in Subsection 6.1)) shall be treated as unfunded for tax purposes and for purposes of Title I of ERISA. An Employer's obligations hereunder, if any, to a Participant (or to a Participant's beneficiary) shall be unsecured and shall be a mere promise by the Employer to make payments hereunder in the future. A Participant (or the Participant's beneficiary) shall be treated as a general unsecured creditor of the Employer.
- 2.2. <u>Administration</u>. The Plan shall be administered by the Plan Administrator. The Plan Administrator shall serve at the pleasure of the Company's Board of Directors and may be removed by such Board, with or without cause. The Plan Administrator may resign upon prior written notice to the Company's Board of Directors.

The Plan Administrator shall have the powers, rights, and duties set forth in the Plan and shall have the power, in the Plan Administrator's sole and absolute discretion, to determine all questions arising under the Plan, including the determination of the rights of all persons with respect to the Plan and to interpret the provisions of the Plan and remedy any ambiguities, inconsistencies, or omissions Any decisions of the Plan Administrator shall be final and binding on ail persons with respect to the Plan and the benefits provided under' the Plan. The Plan Administrator may delegate the Plan Administrator's authority under

the Plan to one or more officers or directors of the Company; provided, however, that (a) such delegation must be in writing, and (b) the officers or directors of the Company to whom the Plan Administrator is delegating authority must accept such delegation in writing.

If a Participant is serving as the Plan Administrator (either individually or as a member of a committee), the Participant may not decide or determine any matter or question concerning such Participant's benefits under the Plan that the Participant would not have the right to decide or determine if the Participant were not serving as the Plan Administrator.

### **SECTION 3**

## Eligibility, Participation, Deferral Elections, and Employer Contributions

	Eligibility and Participation, Subject to the conditions and limitations of the Plan, the following persons are eligible to participate in the Plan (select and applete option(s)):
X	All Employees with a rank of Manager (insert title) or above and with total earnings of at least \$85,000 per Plan Year
	The following Employees of the Employers:
	(Attach a separate sheet if necessary)
	The following Independent Contractors:
	(Attach a separate sheet if necessary)
Part Cor	rindividuals specified above by an Employer may be changed by action of the Employer An Employee or Independent Contractor shall become a icipant in the Plan upon the execution and filing with the Plan Administrator of a written election to defer a portion of the Employee's or Independent atractor's Compensation. A Participant shall remain a Participant until the entire balance of the Participant's Deferred Compensation Account has been ributed.

- 3.2. <u>Rules for Deferral Elections</u>. Any person identified in Subsection 3.1 may make a Deferral Election to defer receipt of Compensation he or she otherwise would be entitled to receive for a Plan Year in accordance with the rules set forth below:
  - (a) All Deferral Elections must be made in writing on the form prescribed by the Plan Administrator and will be effective only when filed with the Plan Administrator no later than the date specified by the Plan Administrator. In no event may a Deferral Election be made later than the last day of the

Plan Year preceding the Plan Year in which the amount being deferred would otherwise be made available to the Participant. However, in the case of a Participant's initial year of employment or association with an Employer, the Participant may make a Deferral Election with respect to compensation for services to be performed subsequent to such Deferral Election, provided such election is made no later than 30 days after the date the Participant first becomes eligible for the Plan. Furthermore, in the case of a short initial Plan Year, each Participant may make a Deferral Election with respect to compensation for services to be performed subsequent to such Deferral Election, provided such election is made no later than 30 days after the Effective Date.

- (b) With respect to Plan Years following the Participant's initial Plan Year of participation in the Plan, failure to complete a subsequent Deferral Election shall constitute a waiver of the Participant's right to elect a different amount of Compensation to be deferred for each such Plan Year and shall be considered an affirmation and ratification to continue the Participant's existing Deferral Election. However, a Participant may, prior to the beginning of any Plan Year, elect to increase or decrease the amount of Compensation to be deferred for the next following Plan Year by filing another Deferral Election with the Plan Administrator in accordance with paragraph (a) above.
- (c) A Deferral Election in effect for a Plan Year may not be modified during the Plan Year, except that a Participant may terminate the Participant's Deferral Election during a Plan Year in the event of an Unforeseeable Financial Emergency.

#### 3.3. Amounts Deferred. (select one option):

Option 1. 

Deferral of a Percentage of Compensation plus Bonus.

Commencing on the Effective Date, a Participant may elect to defer (a) up to 100% of the Participant's Compensation for a Plan Year and (b) up to 100% of the Participant's bonus for a Plan Year. The amount of Compensation and bonus deferred by a Participant shall be credited to the Participant's Deferred Compensation Account as of the Valuation Date coincident with or immediately following the date such Compensation and bonus would, but for the Participant's Deferral Election, be payable to the Participant.

Option 2. 

Deferral of Bonus Only.

Commencing on the Effective Date, a Participant may elect to defer up to % of any bonus awarded to the Participant during a Plan Year. The amount of bonus deferred by a Participant shall be credited to the Participant's Deferred Compensation Account as of the Valuation Date coincident with or immediately following such the date such bonus would, but for the Participant's Deferral Election, be payable to the Participant.

Option 3. 

Deferral of Excess Contributions

Commencing on the Effective Date, a Participant may elect to defer an amount equal to the Excess Contributions payable to the Participant during a Plan Year. Such amount shall be credited to the Participant's Deferred Compensation Account as of the Valuation Date coincident with or immediately following the date such amount would, but for the Participant's Deferral Election, be payable to the Participant.

3.4 Employer Discretionary Contributions. If selected by the Company below, an Employer may, in its sole discretion, credit to the Deferred Compensation Account of any Participant employed by that Employer an amount determined by the Employer in its sole discretion (an "Employer Discretionary"

Contribution") for a Plan Year. Any Employer Discretionary Contribution for a Plan Year will be credited to a Participant's Deferred Compensation Account as of the Valuation Date specified by the Employer.
(select one of the following options)
☐ No Employer Discretionary Contributions will be made under the Plan
☑ Employer Discretionary Contributions may be made under the Plan for a Plan Year as determined by each Employer in its sole discretion.
SECTION 4
Deferred Compensation Accounts
4.1. <u>Deferred Compensation Accounts</u> . All amounts deferred pursuant to one or more Deferral Elections under the Plan and any Employer Discretionary Contributions shall be credited to a Participant's Deferred Compensation Account and shall be adjusted under Subsection 4.2
4.2. <u>Deferral Account Adjustments and Investment Options</u> . As of each Valuation Date, the Plan Administrator shall adjust amounts in a Participant's Deferred Compensation Account to reflect earnings (or losses) in the Investment Options (as defined in Subsection 4.4) attributable to the Participant's Deferred Compensation Account Earnings (or losses) on amounts in a Participant's Deferred Compensation Account first has a positive balance and shall continue to accrue until the entire balance in the Participant's Deferred Compensation Account has been distributed. Earnings (or losses) shall be credited to a Participant's Deferred Compensation Account based on the realized rate of return (net of any expenses and taxes paid from the Trust) on the Investment Options attributable to the Participant's Deferred Compensation Account.
4.3. <u>Vesting</u> . A Participant shall be fully vested in the amounts in the Participant's Deferred Compensation Account attributable to the Participant's Deferral Elections. If Employer Discretionary Contributions are made under the Plan, a Participant shall be vested in the amount in the Participant's Deferred Compensation Account attributable to Employer Discretionary Contributions in accordance with the following (select Options 1, 2, or 3 and, if desired, Option 4. and/or Option 5):
Option 1. ⊠ Five Year Vesting Schedule
Vesting for Participants will be determined by (select one):
⊠ Years of Service with the Employer.
☐ Years of Participation in this Plan.
Nonforfeitable Percentage
Less than 5 years 0%
5 or more years 100%
Option 2 □ Seven Year Graded Vesting Schedule
Vesting for Participants will be determined by (select one):
☐ Years of Service with the Employer.
☐ Years of Participation in this Plan.
C-7

#### Nonforfeitable Percentage

Less than 3 years	0%
3 years	20%
4 years	40%
5 years	60%
6 years	80%
7 years	100%

- Option 3. 

  Other vesting schedule as described below:
- Option 4. 
  Notwithstanding the foregoing vesting schedule, the balance in a Participant's Deferred Compensation Account attributable to Employer Discretionary Contributions will be forfeited if the Participant's employment or association with the Employer is terminated for Good Cause.
- Option 5 Notwithstanding the foregoing vesting schedule, the entire balance in a Participant's Deferred Compensation Account attributable to Employer Discretionary Contributions will be fully vested upon the Participant's Early Retirement Date.

For the purpose of determining a Participant's vested benefit with respect to Employer Discretionary Contributions, a "Year of Service" means each twelve-month period of employment or association with the Company and the Affiliates, and a "Year of Participation" means each twelve-month period of active participation in the Plan. Notwithstanding the foregoing, a Participant shall be fully vested in the entire balance in the Participant's Deferred Compensation Account upon the Participant's Normal Retirement Date, death or becoming disabled (as provided in Subsection 5.2 below), provided the date on which the Participant dies or becomes disabled occurs while the Participant is actively employed by or associated with the Employers. The portion of a Participant's Deferred Compensation Account in which the Participant is not fully vested shall be forfeited to the Employer by the Participant.

If elected by the Company under Option 4. above, notwithstanding the vesting schedule selected in Option 1., 2., or 3. above, the balance in a Participant's Deferred Compensation Account attributable to Employer Discretionary Contributions will be forfeited (and neither the Participant nor the Participant's beneficiaries will have any rights thereto) if the Participant's employment with the Employer is terminated for Good Cause. "Good Cause" means the Participant's gross negligence, fraud, dishonesty, or willful violation of any law or significant policy of the Employer that is committed in connection with the Participant's employment by or association with the Employer Whether a Participant has been terminated for Good Cause shall be determined by the Plan Administrator

4.4 <u>Investment Options</u>. The Company shall, from time to time and in its sole discretion, select one or more investment vehicles ("Investment Options") to be made available as the measuring standards for crediting earnings or losses to each participant's Deferred Compensation Account A Participant may select from such Investment Options in a manner established by the Company, the investment vehicle or vehicles to apply to his or her accounts and may change such selections, all in accordance with such rules as the Company may establish. Notwithstanding the foregoing, the Committee may change the method for crediting earnings or losses to each participant's accounts as described above by written notice to each Participant (including former Participants who then have a Deferred Compensation Account which would

be affected by such change), which notice shall specify the new method for crediting earnings or losses to be used under this section, the effective date of such change and the Deferred Compensation Accounts to which such new method shall apply.

#### **SECTION 5**

#### Payment of Benefits

5.1. <u>Time and Method of Payment.</u> Payment of the vested portion of a Participant's Deferred Compensation Account shall be made as soon as practicable
following the Valuation Date coincident with or next following the Participant's Distribution Date; provided, however, that if the Company has elected a
daily Valuation Date, such payment will be made as soon as practicable following the last business day of the month in which the Participant's Distribution
Date occurs. Payment of the vested portion of a Participant's Deferred Compensation Account shall be made as follows (select one option):

Option 1.  $\square$  A single, lump sum payment.

- Option 2. 

  Substantially equal monthly installment payments for months.
- Option 3. Substantially equal monthly installment payments for 60 months with a one-time option to receive a lump sum payment. The Participant may elect to receive a single, lump sum payment in lieu of installment payments. Such election must be made by filing a written election with the Plan Administrator at least 30 days prior to the time installment payments would otherwise begin, and such election is subject to approval by the Employer of the Participant.
- 5.2 <u>Payment Upon Disability</u>. In the event a Participant becomes disabled (as defined below) while the Participant is employed by or associated with an Employer, payment of the Participant's Deferred Compensation Account shall be made (or shall commence) as soon as practicable after the Valuation Date coincident with or next following the date on which the Plan Administrator determines that the Participant is disabled. For purposes of this Subsection 5.2, a Participant shall be considered disabled if the Participant is unable to engage in any substantially gainful activity by reason of any medically determined physical or mental impairment that can be expected to result in death or that has lasted or can be expected to last for a continuous period of not less than twelve months. Whether a Participant is disabled for purposes of the Plan shall be determined by the Plan Administrator, and in making such determination, the Plan Administrator may rely on the opinion of a physician (or physicians) selected by the Plan Administrator for such purpose.
- 5.3. <u>Payment Upon Death of a Participant</u>. A Participant's Deferred Compensation Account shall be paid to the Participant's beneficiary (designated in accordance with Subsection 5.4) in a single lump sum as soon as practicable following the Valuation Date coincident with or next following the Participant's death.
- 5.4. Beneficiary. If a Participant is married on the date of the Participant's death, the Participant's beneficiary shall be the Participant's spouse, unless the Participant names a beneficiary or beneficiaries (other than the Participant's spouse) to receive the balance of the Participant's Deferred Compensation Account in the event of the Participant's death prior to the payment of the Participant's entire Deferred Compensation Account. To be effective, any beneficiary designation must be filed in writing with the Plan Administrator in accordance with rules and procedures adopted by the Plan Administrator for that purpose. A Participant may revoke an existing beneficiary designation by filing another written beneficiary designation with the Plan Administrator. The latest beneficiary designation received by the

Plan Administrator shall be controlling. If no beneficiary is named by a Participant, or if the Participant survives all of the Participant's named beneficiaries and does not designate another beneficiary, the Participant's Deferred Compensation Account shall be paid in the following order of precedence:

- (a) The Participant's spouse;
- (b) The Participant's children (including adopted children) per stripes; or
- (c) The Participant's estate.
- 5.5. <u>Unforeseeable Financial Emergency</u>. If the Plan Administrator determines that a Participant has incurred an Unforeseeable Financial Emergency, the Participant may receive in cash the portion of the balance of the Participant's Deferred Compensation Account needed to satisfy the Unforeseeable Financial Emergency, but only if the Unforeseeable Financial Emergency may not be relieved (a) through reimbursement or compensation by insurance or otherwise or (b) by liquidation of the Participant's assets to the extent the liquidation of such assets would not itself cause severe financial hardship. A payment on account of an Unforeseeable Financial Emergency shall not be in excess of the amount needed to relieve such Unforeseeable Financial Emergency and shall be made as soon as practicable following the date on which the Plan Administrator approves such payment.
- 5.6. Withholding of Taxes. In connection with the Plan, the Employers shall withhold any applicable Federal, state or local income tax and any employment taxes, including Social Security taxes, at such time and in such amounts as is necessary to comply with applicable laws and regulations.

#### **SECTION 6**

#### Miscellaneous

- 6.1. <u>Funding</u>. Each Employer under the Plan shall establish and maintain one or more trusts (individually, a "Trust") to hold assets to be used for payment of benefits under the Plan. The assets of the Trust with respect to benefits payable to the Participants employed by or associated with an Employer shall remain the assets of such Employer subject to the claims of its general creditors. Any payments by a Trust of benefits provided to a Participant under the Plan shall be considered payment by the applicable Employer and shall discharge such Employer from any further liability under the Plan for such payments.
- 6.2. <u>Rights</u>. Establishment of the Plan shall not be construed to give any Employee or Independent Contractor the right to be retained by the Employers or to any benefits not specifically provided by the Plan.
- 6.3. Interests Not Transferable. Except as to withholding of any tax under the laws of the United States or any state or locality and the provisions of Subsection 5.4, no benefit payable at any time under the Plan shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, or any other encumbrance of any kind or to any attachment, garnishment, or other legal process of any kind. Any attempt by a person (including a Participant or a Participant's beneficiary) to anticipate, alienate, sell, transfer, assign, pledge, or otherwise encumber any benefits under the Plan, whether currently or thereafter payable, shall be void. If any person shall attempt to, or shall alienate, sell, transfer, assign, pledge or otherwise encumber such person's benefits under the Plan, or if by any reason of such person's bankruptcy or other event happening at any time, such benefits would devolve upon any other person or would not be enjoyed by the person entitled thereto under the Plan, then the Plan Administrator, in the Plan Administrator's sole discretion, may terminate the interest in any such benefits of the person otherwise entitled thereto under the Plan and may hold or apply such benefits in such manner as the Plan Administrator may deem proper.

- 6.4. <u>Forfeitures and Unclaimed Amounts</u>. Unclaimed amounts shall consist of the amounts in the Deferred Compensation Account of a Participant that cannot be distributed because of the Plan Administrator's inability, after a reasonable search, to locate a Participant or the Participant's beneficiary, as applicable, within a period of two years after the Distribution Date upon which the payment of benefits became due. Unclaimed amounts shall be forfeited at the end of such two-year period. These forfeitures will reduce the obligations of the Employers, if any, under the Plan. After an unclaimed amount has been forfeited, the Participant or beneficiary, as applicable, shall have no further right to amounts in the Participant's Deferred Compensation Account.
- 6.5. Controlling Law. The law of the state New Hampshire shall be controlling in all matters relating to the Plan to the extent not preempted by Federal law.
- 6.6. Number. Words in the plural shall include the singular, and the singular shall include the plural.
- 6.7. Action by the Employers. Except as otherwise specifically provided herein, any action required of or permitted to be taken by an Employer under the Plan shall be by resolution of its Board of Directors or by resolution of a duly authorized committee of its Board of Directors or by action of a person or persons authorized by resolution of such Board of Directors or such committee.
- 6.8. Offset for Obligations to Employer. If, at such time as a Participant or a Participant's beneficiary becomes entitled to benefit payments hereunder, the Participant has any debt, obligation or other liability representing an amount owing to an Employer or an Affiliate of the Employer, and if such debt, obligation, or other liability is due and owing at the time benefit payments are payable hereunder, the Employer may offset the amount owing it or an Affiliate against the amount of benefits otherwise distributable hereunder.
- 6.9. No Fiduciary Relationship. Nothing contained in this Plan, and no action taken pursuant to its provisions by either the Employers or the Participants shall create, or be construed to create a fiduciary relationship between the Employer and the Participant, a designated beneficiary, other beneficiaries of the Participant, or any other person.
- 6.10. Claims Procedures. Any person (hereinafter referred to as a "Claimant") who believes that he or she is being denied a benefit to which he or she may be entitled under the Plan may file a written request for such benefit with the Plan Administrator. Such written request must set forth the Claimant's claim and must be addressed to the Plan Administrator, at the Company's principal place of business. Upon receipt of a claim, the Plan Administrator shall advise the Claimant that a reply will be forthcoming within ninety days and shall deliver a reply within ninety days. The Plan Administrator may, however, extend the reply period for an additional ninety days for reasonable cause. If the claim is denied in whole or in part, the Plan Administrator shall issue a written determination, using language calculated to be understood by the Claimant, setting forth:
  - (a) The specific reason or reasons for such denial;
  - (b) The specific reference to pertinent provisions of the Plan upon which such denial is based;
  - (c) A description of any additional material or information necessary for the Claimant to perfect the Claimant's claim and an explanation why such material or such information is necessary; and
  - (d) Appropriate information as to the steps to be taken if the Claimant wishes to submit the claim for review, and the time limits for requesting such a review.

Within sixty days after' the receipt by the Claimant of the written determination described above, the Claimant may request in writing, that the Plan Administrator review the Plan Administrator's determination. The request must be addressed to the Plan Administrator, at the Company's principal place of business. The Claimant or the Claimant's duly authorized representative may, but need not, review the pertinent documents and submit issues and comments in writing for consideration by the Plan Administrator. If the Claimant does not request a review of the Plan Administrator's determination within such sixty day-period, the Claimant shall be barred and estopped from challenging the Plan Administrator's determination. "Within sixty days after the Plan Administrator's receipt of a request for review, the Plan Administrator will review the determination. After considering all materials presented by the Claimant, the Plan Administrator will render a written determination, written in a manner calculated to be understood by the Claimant setting forth the specific reasons for the decision and containing specific references to the pertinent provisions of the Plan on which the decision is based. If special circumstances require that the sixty day time period be extended, the Plan Administrator will so notify the Claimant and will render the decision as soon as practicable, but no later than one hundred twenty days after receipt of the request for review.

6.11. Notice. Any notice required or permitted to be given under the provisions of the Plan shall be in writing, and shall be signed by the party giving or making the same. If such notice, consent or demand is mailed to a party hereto, it shall be sent by United States certified mail, postage prepaid, addressed to such party's last known address as shown on the records of the Employers. Notices to the Plan Administrator should be sent in care of the Company at the Company's principal place of business. The date of such mailing shall be deemed the date of notice. Either party may change the address to which notice is to be sent by giving notice of the change of address in the manner set forth above.

#### **SECTION 7**

#### **Employer Participation**

- 7.1. Adoption of Plan. Any Affiliate of the Company may, with the approval of the Company, adopt the Plan by filing with the Company a resolution of its Board of Directors to that effect.
- 7.2. Withdrawal from the Plan by Employer. Any Employer shall have the right, at any time, upon the approval of, and under such conditions as may be provided by the Plan Administrator, to withdraw from the Plan by delivering to the Plan Administrator written notice of its election so to withdraw. Upon receipt of such notice by the Plan Administrator, the portion of the Deferred Compensation Account of Participants and beneficiaries attributable to amounts deferred while the Participants were employed by or associated with such withdrawing Employer shall be distributed from the Trust at the direction of the Plan Administrator in cash at such time or times as the Plan Administrator in the Plan Administrator's sole discretion, may deem to be in the best interest of such Participants and their beneficiaries. To the extent the amounts held in the Trust for the benefit of such Participants and beneficiaries are not sufficient to satisfy the Employer's obligation to such Participants and their beneficiaries accrued on account of their employment with the Employer, the remaining amount necessary to satisfy such obligation shall be an obligation of the Employer, and the other Employers shall have no further obligation to such Participants and beneficiaries with respect to such amounts.

#### **SECTION 8**

### Amendment and Termination

The Company intends the Plan to be permanent, but reserves the right at any time to modify, amend or terminate the Plan; provided however, that except as provided below, any amendment or termination of the Plan shall not reduce or eliminate any balance in a Participant's Deferred Compensation Account

accrued through the date of such amendment or termination. Upon termination of the Plan, the Company may provide that notwithstanding the Participant's Distribution Date, all Deferred Compensation Account balances will be distributed on a date selected by the Company.

#### **SECTION 9**

#### Change of Control

- 9.1. Overriding Provisions Applicable During a Restricted Period. The following provisions of this Section 9 will become effective on a Restricted Date as the result of a Change of Control and will remain in effect during the Restricted Period beginning on that date until the following related Unrestricted Date, and during the Restricted Period, will supersede any other provisions of the Plan to the extent necessary to eliminate any inconsistencies between the provisions of this Section 9 and any other provisions of the Plan, including any supplements thereto.
- 9.2. <u>Suspension of Part or All of the Overriding Provisions</u>. If a majority of the members of the Entire Board are Continuing Directors (provided such majority is equal to the same number as constituted a majority of the Entire Board immediately prior to the Change of Control), by the affirmative vote of a majority of the Entire Board and a majority of those members of the Entire Board who are Continuing Directors, all or a designated portion or portions of the following provisions of this Section 9 may be declared not applicable as to the specified transaction or event. No portion of the provisions of this Section 9 will apply to any transaction or event to the extent such portion is inconsistent with the requirements of applicable law.
- 9.3. <u>Definitions</u>. For purposes of this Section 9, the definitions set forth in Paragraphs (a) through (k) below will apply. Definitions set forth elsewhere in the Plan also will apply to the provisions set forth in this Section 9, except that where a definition set forth elsewhere in the Plan and a definition set forth in this Subsection conflict, the definition set forth in this Subsection will govern.
  - (a) "Acquiring Person" will mean any Person, who or which, together with all Affiliates and Associates of such Person, is the Beneficial Owner of shares of common stock of the Company constituting more than 20 percent of the common stock then outstanding.
  - (b) "Affiliate" and "Associate" will have the meaning ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Securities Exchange Act of 1934 (the "Act").
  - (c) "Beneficial Owner" will have the meaning ascribed to such term in Rule 13d-3 of the Act.
  - (d) "Board of Directors" will mean the Board of Directors of the Company.
  - (e) A "Change of Control" will be deemed to occur (i) upon any Person becoming an Acquiring Person if the Board of Directors has not recommended that stockholders of the Company tender or otherwise sell their common stock to such Acquiring Person; (ii) upon the approval by the stockholders of the Company of a reorganization, merger or consolidation, in each case, with respect to which persons who were stockholders of the Company immediately prior to such reorganization, merger or consolidation, do not, immediately thereafter, own more than 50 percent of the combined voting power entitled to vote generally in the election of directors of the reorganized, consolidated or merged Company's then outstanding securities; or (iii) upon a liquidation or dissolution of the Company or the sale of all or substantially all of the Company's assets.
  - (f) "Continuing Director" will mean:
    - (i) any member of the Board of Directors immediately prior to a Change of Control, or

(ii) any successor of a Continuing Director who is recommended or elected to succeed such Continuing Director by a majority of the Continuing Directors then in office and is neither an Acquiring Person, an Affiliate of an Acquiring Person, nor a representative or nominee of an Acquiring Person or of any such Affiliate while such person is a member of the Board of Directors.

Notwithstanding the foregoing, a successor will not be deemed to be a Continuing Director unless, immediately prior to his or her appointment or election, a majority of the members of the Entire Board were Continuing Directors (and unless such majority is equal to the same number as constituted a majority of the Entire Board immediately prior to the Change of Control).

- (g) "Person" will mean any individual, firm, corporation or other entity, and will include any "group" as that term is used in Rule 13d-5(b) of the Act.
- (h) "Restricted Date" will mean the date on which a Change of Control occurs.
- (i) "Restricted Period" will mean the period beginning on a Restricted Date and ending on the fifth anniversary of such Restricted Date.
- (j) "Unrestricted Date" will mean the last day of a Restricted Period.
- (k) "Entire Board" will mean the total number of members of the Board of Directors that there would be if there were no vacancies on such Board.
- 9.4. <u>Benefits Vested on Restricted Date</u>. Effective on a Restricted Date, the balances in the Deferred Compensation Accounts (including any contributions and investment earnings after that date) of each Participant who is a Participant in the Plan on that date will become fully vested and nonforfeitable.
- 9.5. <u>Prohibition Against Amendment</u>. During the Restricted Period, the provisions of this Section 9 may not be amended or deleted and may not be superseded by any other provision of the Plan (including the provisions of any exhibit or supplement thereto).

IN WITNESS WHEREOF, the Company has caused this Plan to be executed by its duly authorized officers on this 23 day of July, 2001.

	Viewpoint International, Inc. (Name of Company)	
	Ву:	
	Its:	
ATTEST: ts:		
	C-14	

# AMENDMENT TO THE VIEWPOINT INTERNATIONAL, INC. NONQUALIFIED DEFERRED COMPENSATION PLAN

THIS AMENDMENT to the Viewpoint International, Inc. Nonqualified Deferred Compensation Plan is adopted by Viewpoint International, Inc. (the "Company"), effective as of the date set forth herein.

### WITNESSETH:

WHEREAS, the Company maintains the Viewpoint International, Inc. Nonqualified Deferred Compensation Plan (the "Plan"), and such Plan is currently in effect; and

WHEREAS, the Company wishes to amend the Plan as permitted by Section 8 of the Plan.

**NOW, THEREFORE**, the Company hereby amends the Plan as follows:

- 1. Appendix A shall be added to the Plan in the form attached hereto.
- 2. This amendment shall be effective immediately upon execution.

IN WITNESS WHEREOF, the undersigned has adopted this Amendment effective as of the dates indicated above.

	VIEWPOINT INTERNATIONAL, INC.	
Date:	Ву	
	Name	
	Title	
	C-15	

## APPENDIX A SPECIAL RULES APPLICABLE TO 2003 SPECIAL CLOSING BONUSES

- A1. Exclusion of Certain Bonuses. Notwithstanding any other provision of the plan to the contrary, Compensation as defined in Section 1.4 of the Plan shall not include any bonus (a "Closing Bonus") payable to a Participant contingent on the consummation of the sale of the Company pursuant to that certain Stock Purchase Agreement dated as of April 26, 2003 by and among the Oxford Industries, Inc., the Company, and the stockholders of the Company (the "sale") and so no deferral will be effective with respect to any such bonus except as otherwise expressly provided in this Appendix A.
- A2. Special Deferral Election. Participants who are notified that they may become entitled to receive a Closing Bonus equal to or exceeding \$250,000 (an "Eligible Bonus") may make a special Deferral Election (a "Special Election") with respect to any such Eligible Bonus in accordance with the rules set forth below:
  - (a) The Special Election must be made in writing on the form prescribed by the Plan Administrator for the purpose of such Special Election and must be delivered to the Plan Administrator prior to the consummation of the Sale.
  - (b) The Special Election is subject to the consummation of the Sale and the payment of an Eligible Bonus.
  - (c) The Special Election shall not be valid if the actual Closing Bonus paid to the participant is than \$250,000.
  - (d) The Special Election is applicable solely to an Eligible Bonus and does not revoke or modify any Deferral Election otherwise in effect under the Plan with respect to a participant's Compensation (including any other bonuses).
  - (e) The Special Election is irrevocable.
- A3. Special Deferral Amount. A Participant may elect to defer all or any portion (in a whole percentage or dollar amount) of an Eligible Bonus. The amount deferred by a Participant shall be credited to the Participant's Deferred Compensation Account at the same time and shall be adjusted under Section 4.2 in the same manner as any other bonus under the Plan but shall be accounted for separately from all other amounts credited to such Participant's Account.
- A4. <u>Vesting in Special Deferral Amount</u>. A Participant shall be fully vested in the Participant's Deferred Compensation Account attributable to the Participant's Special Election Pursuant to this Appendix A.
- A5. <u>Time and Method of Payment</u>. Payment of a Participant's Deferred Compensation Account attributable to the Participant's Special Election shall be made in accordance with one of the following options elected by the Participant on the special election form provided to the Participant pursuant to Section A2 above:
  - (a) A single lump sum payment made no sooner than January 1, 2005 and no later than May 31, 2007.
  - (b) Substantially equal annual installment payments commencing on any date elected by the Participant and ceasing no later than May 31, 2007.

Such election is irrevocable and may not be modified at any time for any reason.

A6. Other Plan Provisions Apply. The provisions of this Appendix A shall supercede all inconsistent provisions of the Plan, provided that all other provisions of the Plan shall apply with respect to a Participant and the Deferred Compensation Account attributable to the Participant's Special Election made in accordance with this Appendix A to the extent not inconsistent with the provisions of this Appendix A as determined by the Plan Administrator in its sole and absolute discretion.

# AMENDMENT TO THE VIEWPOINT INTERNATIONAL, INC. NONQUALIFIED DEFERRED COMPENSATION PLAN

Pursuant to § 8 of the Viewpoint International, Inc. Executive Deferred Compensation plan (the "PLAN"), Viewpoint International, Inc. (the "Company") hereby amends the Plan as follows:

1.

Effective as of January 1, 2005, Section 1.18 of the Plan shall be amended to read as follows:

"The Plan Administrator means a committee of at least three (3) persons appointed by the Company."

2

Effective as of January 1, 2005, Section 3.1 of the Plan shall be amended to read as follows:

"3.1 Eligibility and Participation. Subject to the conditions and limitations of the Plan, the following persons are eligible to participate in the Plan: Any Employee who is Employed by the Employer and who is determined by the Employer, in its sole discretion, to be both (i) a member of a select group of management or highly compensated employees and (ii) eligible to participate in the Plan. Any individuals specified by the Employer may be changed by action of the Employer. An Employee shall become a Participant in the Plan upon the execution and filing with the plan Administrator of a written election to defer a portion of the Employee's Compensation. A participant shall remain a Participant until the entire balance of the Participant's Deferred Compensation Account has been distributed."

3.

Except as specifically set forth herein, the terms of the plan shall remain in full force and effect.

IN WITNESS WHEREOF, the Company has caused this Amendment to be executed on the date set forth below.

VIEWPOINT INTERNATIONAL, INC.

By: Title: Date:

C-18

#### EXHIBIT D

#### TOMMY BAHAMA PLAN

#### SPECIAL RULES APPLICABLE TO 2005 COMPENSATION

Notwithstanding any other provision of the Tommy Bahama Plan to the contrary, the provisions of this Exhibit D shall supersede all inconsistent provisions of the Tommy Bahama Plan with respect to amounts deferred in taxable years beginning after December 31, 2004 and before January 1, 2006 (and earnings on such amounts) and earnings in 2005 on deferrals made in taxable years before January 1, 2005. All other provisions of the Tommy Bahama Plan shall apply with respect to such deferrals to the extent not inconsistent with the provisions of this Exhibit D or Section 409A of the Code, as determined by the Committee in its sole and absolute discretion. This Exhibit D is intended to (a) satisfy the requirements of Section 409A(a)(2), (3) and (4) of the Code for deferrals made after December 31, 2004 and before January 1, 2006 and (b) not constitute a material modification of the Tommy Bahama Plan with respect to amounts deferred before January 1, 2005.

- 1. Account(s). A separate bookkeeping account shall be established to account for deferrals made in taxable years beginning after December 31, 2004 and before January 1, 2006 (and any earnings on such deferrals) and earnings in 2005 on deferrals made in taxable years before January 1, 2005. The portion of any Deferred Compensation Account that was not fully vested on December 31, 2004 shall be treated as a deferral made in taxable years beginning after December 31, 2004.
- 2. <u>Deferral Elections.</u> In no event may a Deferral Election be made later than the last day of the Plan Year preceding the Plan Year in which the amount being deferred is earned by the Participant, except that a Deferral Election with respect to Excess Contributions payable to the Participant in 2005 may be made on or before December 31, 2004 in accordance with Q&A 21 of IRS Notice 2005-1.
- 3. <u>Time and Method of Payment.</u> In order for a termination of employment or association with the employers to qualify as a Distribution Event, the termination of employment or association must qualify as a "separation from service" within the meaning of Section 409A of the Code and the regulations thereunder. Section 5.1 of the Tommy Bahama Plan is amended to provide that distributions shall be made in a single, lump sum payment and will commence in the first calendar month that is 6 months from the Participant's Distribution Date.
- 4. <u>Disability or Death.</u> If distribution is made as a result of the Participant's disability or death under Sections 5.2 or 5.3 of the Tommy Bahama Plan, distribution will commence in the first month of the calendar quarter immediately following the quarter in which his or her disability or death occurred. A Participant shall be considered disabled if the Participant is unable to engage in any substantially gainful activity by reason of any medically determined physical or mental impairment that can be expected to result in death or that can be expected to last from a continuous period of not less than twelve months.
- 5. <u>Unforeseeable Financial Emergency.</u> The provisions of Sections 1.20, 3.2(c) and 5.5 of the Tommy Bahama Plan shall not apply, and Section 7.2(d) of the Plan shall apply as if incorporated in the Tommy Bahama Plan.

- 6. <u>Delay of Payments Under Certain Circumstances</u>. Section 7.2(e) of the Plan shall apply as if incorporated in the Tommy Bahama Plan.
- 7. <u>Amendment and Termination</u>. The provisions of Section 8 of the Tommy Bahama Plan shall not apply, and Section 10.12 of the Plan shall apply as if incorporated in the Tommy Bahama Plan.

#### CERTIFICATION

#### I, J. Hicks Lanier, certify that:

- 1. I have reviewed this report on Form 10-Q of Oxford Industries, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: December 5, 2012

/s/ J. Hicks Lanier

J. Hicks Lanier

Chairman and Chief Executive Officer

#### CERTIFICATION

#### I, K. Scott Grassmyer, certify that:

- 1. I have reviewed this report on Form 10-Q of Oxford Industries, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: December 5, 2012 /s/ K. Scott Grassmyer

K. Scott Grassmyer

Senior Vice President - Finance, Chief Financial Officer and Controller

### CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Oxford Industries, Inc. (the "Company") on Form 10-Q ("Form 10-Q") for the quarter ended October 27, 2012 as filed with the Securities and Exchange Commission on the date hereof, I, J. Hicks Lanier, Chairman and Chief Executive Officer of the Company, and I, K. Scott Grassmyer, Senior Vice President — Finance, Chief Financial Officer and Controller of the Company, each certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

(1) The Form 10-Q fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Form 10-O fairly presents, in all material respects, the financial co	ondition and results of	operations of the Company
--	-------------------------	---------------------------

/s/ J. Hicks Lanier

J. Hicks Lanier
Chairman and Chief Executive Officer
December 5, 2012

/s/ K. Scott Grassmyer

K. Scott Grassmyer

K. Scott Grassmyer Senior Vice President - Finance, Chief Financial Officer and Controller December 5, 2012